### **DEVELOPER'S AGREEMENT**

This Agreement (hereinafter the "Agreement" or "Contract") is made this <u>day</u> of <u>Jeffember</u>, 2012, between **THE CITY OF FORT LAUDERDALE**, a Florida municipal corporation, whose address is 100 North Andrews Avenue, Fort Lauderdale, Florida 33301 (hereinafter "Owner") and **RECREATIONAL DESIGN & CONSTRUCTION**, **INC.**, a Florida corporation, whose address is 3990 North Powerline Road, Oakland Park, Florida 33309 (hereinafter "Developer").

WHEREAS, the Owner owns a certain parcel of land (hereinafter referred to as the "Project Site") having a street address of 501 Seabreeze Boulevard, Fort Lauderdale, Florida 33316 and legally described as the INTERNATIONAL SWIMMING HALL OF FAME COMPLEX, according to the Plat thereof, as recorded in Plat Book 138, Page 19 of the Public Records of Broward County, Florida; said lands lying, situate and being in the City of Fort Lauderdale, Broward County, State of Florida;

WHEREAS, the Owner hereby retains the Developer to perform all Work and Design Services in connection with the design and construction of the renovation of the Fort Lauderdale Aquatic Complex (FLAC), including the International Swimming Hall of Fame building (the "Project") at the Site;

WHEREAS, the Plans and Specifications (as hereinafter defined) will be prepared by the Developer for the Project; and

WHEREAS, the following exhibits are attached to this Agreement and incorporated herein by reference:

	Exhibit A	Truth-In-Negotiation Certificate
	Exhibit B	Guaranteed Maximum Price (GMP)
	Exhibit C	<b>Owner Direct Material Equipment Purchase Program</b>
	Exhibit D	Scope of Design Services
	Exhibit E	Scope of Construction Services
	Exhibit F	Form of Payment Bond
	Exhibit G	Form of Performance Bond
	Exhibit H	Release and Affidavit
	Exhibit I	List of Design Professionals
Schedule 3.10.3 – Basis for GMP		

NOW, THEREFORE, in consideration of the mutual covenants and promises contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

### ARTICLE 1: GENERAL AGREEMENT PROVISIONS

1.1 <u>Recitals</u>. The recitals set forth in the Whereas clauses are incorporated by reference and made a part of this Agreement.

1

1.1A <u>Defined Terms</u>. The following terms, as used and referred to herein, shall have the meaning(s) set forth below, unless the context indicates otherwise:

Application for Payment means the written form accepted by the Owner which is to be used by the Developer in requesting progress or final payments and which is to include supporting documentation as is required by the Contract Documents.

Building Code means The Florida Building Code, together with Broward County Amendments thereto, existing and applicable at the time of the execution of this Agreement, which is the code which governs building and construction standards, review of plans for construction, issuance of Building Permits, inspections for compliance with construction standards, issuance of Certificates of Occupancy and Certificates of Completion by the Building Official.

Building Official means that term as defined in the Building Code.

Building Permit or Permit means any building permit issued by the appropriate department, office or official of the City (or other governmental authority having jurisdiction over the Project and Project Site) charged with reviewing the Plans, Specifications, Drawing, details and other Construction Documents for compliance with the Building Code, U.L.D.R. or similar codes applicable to the construction of the Project and having the authority to issue building permits for construction of buildings, structures or other improvements in accordance with Construction Documents therefor reviewed and approved by such department, office or official.

*Business Day* means those days that are Monday through Friday, not including Holidays observed by the City.

*Certificate of Completion* means that certificate issued by the Building Official as proof that a structure or system is complete and for certain types of permits is released for use and may be connected to a utility system. A Certificate of Completion does not grant authority to occupy or connect a building, such as a shell building, prior to the Certificate of Occupancy.

*Certificate of Occupancy* means a certificate of occupancy or temporary certificate of occupancy issued by the Building Official or other appropriate governmental authority under the Building Code for any part of the Project.

*Certificate of Substantial Completion* means the Certificate provided by the City Manager on behalf of the Owner certifying that all Work, excluding the Punch List items, has been completed, inspected and accepted by Owner.

*Change Order* is a written document prepared by the Developer and executed by the Owner's City Manager and Developer, setting out in detail and authorizing an addition, deletion or revision to the Work, which may involve a reduction in the GMP or

an adjustment of the Contract Time issued on or after the Effective Date of the Agreement. A Change Order must comply with the Contract Documents.

*City* means the City of Fort Lauderdale, a Florida municipal corporation and a political subdivision of the State of Florida. The term includes the elected officials, officers, employees, agents and volunteers of the City. In the event that the City exercises its regulatory authority as a government body, the exercise of such regulatory authority and the enforcement of any rules, regulations, codes, laws and/or ordinances shall be deemed to have occurred pursuant to City's authority as a governmental body and shall not be attributable in any manner to the City as a party to this Agreement.

*City Code* or *Code* means the ordinances and code of the City that regulate the development and construction of projects and buildings, including the Building Codes and zoning regulations, known as the ULDR.

*City Commission* means the governing body of the Owner.

*City Manager* means the Owner's Chief Executive Officer.

*City of Fort Lauderdale* means the City of Fort Lauderdale, a Florida municipal corporation and Owner herein. The term includes the elected officials, officers, employees, agents and volunteers of the City.

*Claim* means a demand by either the Developer or Owner seeking an adjustment or interpretation of an Agreement term, payment of money, extension of time, or any other relief with regard to the terms of this Agreement or the Construction Documents.

*Construction Change Directive* means a written document prepared and executed by the Owner which serves to change the Work and provides for any proposed adjustment in the GMP and/or Contract Time. City Manager has the authority to issue Construction Change Directives in the cumulative amount not to exceed five (5) percent of the GMP and increase the Contract Time by up to 90 days without City Commission approval. City Manager shall receive prior City Commission approval for any single Construction Change Directive in excess of \$25,000.

Construction Documents means and includes, without limitation, the Agreement, the Design Criteria Package, the Exhibits and any duly executed addenda, Plans, Drawings and Specifications, Project Documents for the construction of the Project prepared by the Developer, but only after the Construction Documents have been approved in writing by the Owner and any duly executed addenda issued thereto and modifications, Change Orders, Construction Change Directives, Amendments and/or clarifications issued after execution of the Agreement, all of which shall have been agreed to and executed by both Owner and Developer.

*Construction Services* means, after the Construction Documents have been sufficiently completed by Developer (as defined in § 3.10.1 below), those services

provided by the Developer, including management, supervision, labor, materials, tools, fuel, supplies, utilities, equipment and services of every kind and type necessary to diligently, timely and fully perform and complete in a good and workmanlike manner the construction of the Work (or designated portions thereof) in accordance with all of the terms and conditions of the Contract Documents.

Contract Documents means and shall consist of this Agreement, the Design Criteria Package, the Exhibits identified herein, and any duly executed Addenda, Drawings and Specifications for construction of the Project, being prepared by the Developer, but only after the Construction Documents have been completed by Developer and approved in writing by Owner and any duly executed addenda issued thereto and modifications, Change Orders, Construction Change Directives, Amendments and or clarifications issued after execution of the Agreement, all of which shall have been agreed to and executed by both Owner and Developer.

*Contract Time* means the number of days stated in the Agreement for the completion of the Work. The dates on which the Work shall be started and the dates on which the Work shall be completed as stated in the Notice to Proceed.

*Cost of the Work* means all costs necessarily and reasonably incurred by Developer in the proper performance of the designated Work for the Construction Services in accordance with the terms of this Agreement. Such costs shall be at rates not higher than the standard paid at the place of the Project except with the prior written consent of Project Manager only after Developer has provided sufficient support in writing that exceptional circumstances exist which justify the payment of rates higher than the standard.

*Day(s).* In computing any period of time expressed in day(s) in this Developer's Agreement, the day of the act, event, or default from which the designated period of time begins to run shall not be included. The last day of the period so computed shall be included unless it is a Saturday, Sunday, or Holiday, in which event the period shall run until the end of the next day which is neither a Saturday, Sunday or Holiday. When the period of time prescribed or allowed is less than seven (7) days, intermediate Saturdays, Sundays, and Holidays shall be excluded in the computation.

Designer is the entity or entities whose services have been retained by the Developer to provide the main professional architectural and engineering design services. The Developer has appointed Zyscovich Inc., as Architect of Record and shall be deemed to be the Designer. Developer has appointed C3TS - Corzo Castella Carballo Thompson Salman, P.A. as the Engineer of Record. The term "Designer" includes the Designer's authorized representatives, successors, assignors and/or subconsultants. Developer has retained, or will retain, additional design professionals, as noted on the list of design professionals attached hereto as **Exhibit I** (collectively, "Additional Designers").

4

*Design Criteria Package* means the Phase II-Design Development Documents as described in **Exhibit D** to this Agreement.

Design Fee means the total lump sum compensation for all Design Services.

Design Services means those services involved in the architectural and engineering designs necessary and/or incidental to preparing the Construction Documents, in providing value engineering services, reviewing Construction Documents for constructability, assisting and meeting with the Owner during various design phases, and preparing cost estimates and schedules as provided herein.

Developer's Fee shall be Developer's total compensation for all overhead not reimbursable as Cost of the Work under § 3.5, as well as Developer's total profit for Construction Services.

*Direct Purchase Program* means that purchasing program set forth in **Exhibit C** to this Agreement.

Drawings means the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, and generally include Plans, elevations, sections, details, schedules and diagrams and the written requirements for the materials, equipment, systems and workmanship for completion of the Work and performance of related services.

*Environmental Assessments* means an assessment of the environmental condition of the Project Site which such assessment must be conducted by engineers licensed to perform such tasks by the State of Florida.

*Field Order* is a written approval for the Developer to proceed with the Work requested by the Owner, which is so minor in nature that it does not involve additional costs or additional construction time.

*GMP* means the Guaranteed Maximum Price that Owner is obligated to pay for ' completion of the Project and all Work associated therewith.

Hazardous Materials (HAZMAT) means any solid, liquid, or gaseous material that is toxic, flammable, radioactive, corrosive, chemically reactive, or unstable upon prolonged storage in quantities that could pose a threat to life, property, or the environment defined in Section 101(14) of Comprehensive Environmental Response, Compensation and Liability Act of 1980 and in 40 CFR 300.6). Also defined by 49 CFR 171.8 as a substance or material designated by the Secretary of Transportation to be capable of posing an unreasonable risk to health, safety, and property when transported in commerce and which has been so designated.

*Hazardous Substance* means substances as defined by Section 101(14) of the Comprehensive Environmental Response, Compensation and Liability Act; any substance

designated pursuant to Section 311(b) (2) (A) of the Clean Water Act; any element, compound, mixture, solution or substance designated pursuant to Section 102 identified under or listed pursuant to Section 3001 of the Solid Waste Disposal Act (but not including any waste listed under Section 307[a] of the Clean Water Act); any hazardous air pollutant listed under Section 112 of the Clean Air Act; and any imminently hazardous chemical substance or mixture pursuant to Section 7 of the Toxic Substances Control Act. The term includes any material, substance of waste in concentrations that violate any other federal, state or local environmental statute, ordinance or regulation. The term does not include petroleum, including crude oil or any fraction thereof, which is not otherwise specifically listed or designated as a hazardous substance in the first sentence of this paragraph, and the term does not include natural gas, natural gas liquids, liquefied natural gas, or synthetic gas usable for fuel (or mixtures of natural gas and such synthetic gas).

*Hazardous Waste* means those solid wastes designated by OSHA in accordance with 40 CFR 261 due to the properties of ignitability, corrosivity, reactivity, or toxicity. Any material that is subject to the Hazardous Waste Manifest requirements of the EPA specified in 40 CFR Part 262.

*Holidays* means those designated non-work days as established by the City Commission of the City of Fort Lauderdale.

Instruments of Service are the representations in any medium of expression now known or later developed of the tangible and intangible creative work performed by Developer's Designer or Additional Designers under the professional services agreement between the Designer or Additional Designers and Developer for the Project. Instruments of Service may include, but are not limited to, studies, surveys, models, sketches, drawings, Specifications, and other similar materials.

Notice to Proceed means that written notice by the Project Manager on behalf of the Owner fixing the date on which the Contract Time will commence to run and on which the Contract Time will end. Notwithstanding anything contained herein to the contrary, in no event will Developer be required to commence construction until Developer has received the Building Permit and all other necessary Permits to commence construction of the Project in accordance with the agreed upon Construction Documents.

Owner means the City of Fort Lauderdale, a Florida municipal corporation ("City") whose governing body is the City Commission. Unless otherwise indicated, actions taken by the Owner shall be authorized by the City Commission. The term includes all Authorities, Boards, Committees, Bureaus, Divisions, Departments and offices thereof together with individual members, elected officials, officers, employees agents and volunteers thereof when acting in their official capacity or within the scope and authority of their position on behalf of the Owner.

6

*Person* means any natural person, firm, partnership (general or limited), corporation, company, professional association, unincorporated association, joint venture, joint stock association, estate, trust, business trust, cooperative, limited liability company, limited liability partnership or body politic, including any heir, executor, personal representative, executor, administrator, guardian, trustee, receiver, successor or assignee, or other person acting in a similar representative capacity.

#### Permit - see Building Permit.

*Plans* means those drawings which show the character and scope of the Work to be performed and which have been prepared by the Developer and approved by Owner and are referred to in the Contract Documents.

*Project* means the completion of all Work necessary or incidental to the construction and renovation of FLAC, including the International Swimming Hall of Fame Museum, within the Project Site. The term *Project* also includes the non-exclusive use, occupation and operation of the Project Site for the parking of motor vehicles, ingress to and egress from the Project site, landscaping and associated irrigation and draining in accordance with the Construction. The term *Project* shall not include the possession, use or occupancy of the Project Site for any other purpose, except as expressly authorized in this Agreement. The term *Project* includes any portion thereof.

*Project Documents* means copies or originals of all records, documents, drawings, notes, tracings, plans, Auto CADD files, Specifications, maps, evaluations, reports and other technical data, other than working papers, prepared or developed by Developer under this Agreement.

Project Manager means the person designated by the City Manager with the authority to supervise the Work and approve matters contemplated in this Agreement which are presented in writing, provided, however, that the Project Manager is not authorized to issue any orders or instructions to the Developer that would have the effect of modifying or changing in any manner whatsoever the (1) Scope of Services to be provided and performed by Developer hereunder; (2) the Contract Time; or (3) the amount of compensation Owner is obligated or committed to pay Developer.

Project Schedule means that Exhibit 15 attached to GMP Exhibits B.

*Project Site* means the real property upon which the Project is to be constructed, said real property having a street address of 501 Seabreeze Boulevard, Fort Lauderdale, FL 33316, legally described as SWIMMING HALL OF FAME COMPLEX, according to the Plat thereof, as recorded in Plat Book 138, Page 19 of the Public Records of Broward County, Florida; said lands lying, situate and being in the City of Fort Lauderdale, County of Broward, State of Florida.

*Project Program* means the design concept, cost estimates and Project Schedule submitted to the Owner by the Developer on March 28, 2012 and presented to the City Commission.

*Provide* means to furnish and install materials and equipment, together with all incidentals for a complete and ready to use item and system.

*Punch List* means the Owner's list of Work yet to be done or corrected by the Developer before the Final Completion date can be determined by Owner.

*Specifications* means that portion of the Contract Documents which are the written requirements for the materials, equipment, systems, standards and workmanship for completion of the Work and performance of related services.

Substantial Completion or Substantially Complete means the date when the Work is sufficiently complete in accordance with the Contract Documents so that the Owner may occupy or utilize the Project for its intended use, and only minor Punch List items which can be corrected or completed without any material interference with the Owner's use of the Project remain to be corrected or completed. Issuance of all Certificates of Occupancy and Certificates of Completion for Work on the Project by the Building Official shall conclusively establish Substantial Completion.

Substantially Completed Date means the earlier of the date of issuance of the Certificate of Occupancy or Certificate of Completion by the Building Official or the date when the Developer has requested in writing, stating that the Work is substantially completed and is ready for an inspection and issuance of a final punch list for the Project.

Staging of Materials or Equipment means, during construction or reconstruction of the Project, the placement of materials or equipment or parking of vehicles within the public rights-of-way other than those appearing on the face of the plat for the Project Site or vehicular travel lanes adjacent thereto in any manner other than (a) temporarily and (b) for the purpose of and while actually engaged in the act of loading or offloading materials or equipment from a vehicle. Staging of Materials or Equipment shall include equipment or materials off-loaded from a vehicle and placed within the Project Site when not being removed from the public rights of way described above to the Project Site as soon as practicable.

*U.L.D.R.* or *ULDR* means the Unified Land Development Regulations for the City found in the City's Code of Ordinances, Chapter 47.

*Work* means and shall include all construction services required by the Contract Documents, including all Design Services, labor, materials, equipment and services, and shall include the entire completed delivered Project or the various separately identifiable parts thereof required to be furnished under the Contract Documents, including, but not limited to, (i) performing all tests, inspections, examinations deemed necessary for an evaluation of the condition of the Project Site and its suitability for performing the Work needed to complete the Project, and (ii) applying for and securing of all governmental approvals needed to complete the Project, including all Certificates of Completion and Certificates of Occupancy required under applicable Building Codes. *Work* is also the result of performing services, furnishing labor and furnishing and incorporating material and equipment into the Project as required by the Contract Documents.

### 1.2 <u>Relationship of Parties</u>.

- 1.2.1 The Developer accepts the relationship of trust and confidence established with the Owner by this Agreement, and covenants with the Owner to furnish the Developer's best efforts, skill and judgment, consistent with industry standards for respective Work performed (if dispute arises as to the Work, then limited to industry standard requirements), in performing the Work under this Agreement for the Owner. The Developer shall furnish efficient business administration and supervision, and use the Developer's best efforts, consistent with industry standards, to perform the Work and completion of the Project in an expeditious manner consistent with the terms of the Contract Documents.
- 1.2.2 Wherever the terms of this Agreement refer to some action, consent, or approval (excluding approvals of Change Orders, Construction Change Directives, except as provided for in Section 1.3 below, or amendments to the Agreement) to be provided by Owner or some notice, report or document is to be provided to Owner, such reference to "Owner" shall mean Owner's City Manager or his designee (to the extent such designee has been authorized under this Agreement), unless otherwise stated herein and any such consent or approval shall not be unreasonably withheld or delayed.

### 1.2.3 Intentionally deleted.

- 1.2.4 Nothing contained in the Contract Documents shall be construed to create a contractual relationship between any other person or entity other than the Owner and Developer. However, it is agreed that Owner is an intended third party beneficiary of all contracts for design and engineering services, all subcontracts, purchase orders and other agreements between Developer and third parties pertaining to the Work or this Project. Developer shall incorporate the obligations of this Contract into its respective consultant agreements, subcontracts, supply agreements and purchase orders.
- 1.3 <u>The Contract Documents</u>. Developer shall prepare the Contract Documents for approval by the Owner. Any modification to this Agreement shall only be effective if it is reduced to writing, and duly executed by both parties, except a

Construction Change Directive shall be effective after review by Developer and approval and execution by the City Commission or Owner's City Manager in instances where the Construction Change Directive does not exceed the cumulative amount of five (5) percent of the GMP or the increase in Contract Time does not exceed 90 days.

It is the intent of the Contract Documents to describe a functionally complete 1.4 Project (or portion thereof) to be designed and constructed in accordance with the Contract Documents. Any work, services, materials or equipment that may reasonably be inferred from the Contract Documents (in accordance with industry standards, if a dispute arises as to this matter) as being required to produce the intended result shall be supplied whether or not specifically called for in the Contract Documents. If the Contract Documents include words or terms that have a generally accepted technical or industry meaning, then such words or terms shall be interpreted to have such standard meaning unless. otherwise expressly noted in the Contract Documents. Reference to standard specifications, manuals or codes of any technical society, organization or association or to the laws or regulations of any governmental authority having jurisdiction over the Project, whether such reference be specific or by implication, shall mean the latest standard specification, manual, code, law or regulation in affect at the time the Work is performed, except as may be otherwise specifically stated herein.

1.5 If during the performance of the Work, Developer discovers a conflict, error or discrepancy in the Contract Documents, Developer shall, within five (5) days of discovery by Developer, report same to Project Manager in writing, and before proceeding with the Work affected thereby, shall provide a written interpretation or clarification to Project Manager for Project Manager's review and approval. Prior to commencing each portion of the Work affected by the conflict, error or discrepancy, Developer shall first take all necessary field measurements and verify the applicable field conditions. After taking such measurements and verifying such conditions, Developer shall carefully compare such measurements and conditions with the requirements of the Contract Documents, taking into consideration all other relevant information known to Developer, for the purpose of identifying and bringing to Owner's attention all conflicts or discrepancies with the Contract Documents.

1.6 Construction Documents approved by Owner are intended to show general arrangements, design and extent of Work and are not intended to serve as shop drawings. Specifications are separated into divisions for convenience of reference only and shall not be interpreted as establishing divisions for the Work, trades, subcontracts or extent of any part of the Work. In the event of a discrepancy between or among the Drawings, Specifications or other Contract Document provisions, Developer shall follow the order of priority of documents listed in Section 3.43.1 hereof. Unless otherwise specifically mentioned, all anchors, bolts, screws, fittings, fillers, hardware, accessories, trim and other parts

required in connection with any portion of the Work to make a complete, serviceable, finished and first quality installation shall be furnished and installed as part of the Work, whether or not called for by the Contract Documents; provided, same complies with industry standards.

- 1.7 All indications or notations which apply to one of a number of similar situations, materials or processes shall be deemed to apply to all such situations, materials or processes wherever they appear in the Work, except where a contrary result is clearly indicated by the Contract Documents or industry standards dictate otherwise.
- 1.8 Where no explicit quality or standards for materials or workmanship are established for the Work, such Work is to be consistent with the standard practices of the trades and free of defects and as reasonably inferable from the Drawings and consistent with the quality of the surrounding Work and of the construction of the Project generally and consistent with industry standards.
- 1.9 All manufactured articles, materials, and equipment shall be applied, installed, connected, erected, used, cleaned, and conditioned in accordance with the manufacturer's written or printed directions and instructions unless otherwise indicated in the Contract Documents.
- 1.10 The mechanical, electrical and fire protection Drawings are diagrammatic only, and are not intended to show the alignment, physical locations or configurations of such Work. Such Work shall be installed without additional cost to the Owner to clear all obstructions, permit proper clearances for the work of other trades, and present an orderly appearance where exposed. Prior to beginning such Work, the Developer shall prepare coordination drawings showing the exact alignment, physical location and configuration of the mechanical, electrical and fire protection installations and demonstrating to the Developer's satisfaction that the installations will comply with the preceding sentence.
- 1.11 The Developer must coordinate with the Owner's Information Technology Services (ITS) Department during Design Development Phase, for the installation of technology infrastructure cabling throughout the Project. Fiber optic or copper cabling will also be installed or coordinated with a service provider(s) via underground conduit provided by the Developer to the public right-of-way from a designated communication room. The Developer must include ITS during site planning to determine the number of underground conduits required for technology purposes. Owner agrees to provide Developer, when necessary, with all of its requirements relating to technology infrastructure and other information, depending on discipline involved, during the development of the Design Development Documents.
- 1.12 Exact locations for fixtures and outlets shall be identified and located as to quantity and approximate locations. Project Manager may approve minor

location variations (not affecting cost) during rough in phase of construction for Owner convenience.

1.13 Where the Work is to fit with existing conditions or Work to be performed by others, the Developer shall fully and completely join the Work with such conditions or Work, unless otherwise specified.

### 1.14 Ownership and Use of Documents.

- 1.14.1 Developer shall furnish Owner with one (1) signed and sealed copy and one (1) reproducible set of the Construction Documents. All copies of the Construction Documents, required by Developer for execution of the Work, shall be made by Developer from its reproducible set at Developer's sole cost and expense. Upon the completion or termination of this Agreement, if directed by Owner, Developer shall deliver to Owner copies or originals of all records, documents, drawings, notes, tracings, plans, Auto CADD files, specifications, maps, evaluations, reports and other technical data, other than working papers, prepared or developed by Developer under this Agreement ("Project Documents"). Project Manager shall specify whether the originals or copies of such Project Documents are to be delivered by Developer. Developer shall be solely responsible for all costs associated with delivering to Project Manager the Project Documents. Developer, at its own expense, may retain copies of the Project Documents for its files and internal use.
- 1.14.2 Notwithstanding anything in this Agreement to the contrary and without requiring Owner to pay any additional fees, provided that Owner is not in default of any of its obligations hereunder, Developer hereby grants Owner a nonexclusive, irrevocable license in all of the Project Documents for Owner's use on this Project. Developer warrants to Owner that it has full right and authority to grant this license to Owner. Further, Developer consents to Owner's use of the Project Documents to complete the Project following Developer's termination for cause or to perform additions to or remodeling or renovation of the Project; provided, however, Owner assumes all responsibility and liability for any reuse of such Project Documents without written authorization by Developer, Designer and any applicable Additional Designers, and Owner hereby agrees to indemnify, defend and hold Developer, Designer and such Additional Designers harmless from any liability, claim, cost or damage, including attorneys' fees and costs, resulting from such reuse of the Project Documents.

# ARTICLE 2: OWNER'S RESPONSIBILITIES

### 2.1 Information and Services.

- 2.1.1 Owner has approved the Project Program as submitted by the Developer on March 28, 2012.
- 2.1.2 The Owner has established and updated an overall budget for the Project, based on consultation with the Developer, which shall include a contingency line item for changes in the Work, unforeseen conditions and other costs which are the responsibility of the Owner, as more particularly described in Section 3.10 hereof.
- 2.1.3 The parties agree that before execution of this Agreement, Developer has not had the opportunity to conduct investigations and/or testing of the Project Site with respect to its suitability for construction of the Project and any impact on construction costs, including, without limitation, testing of such items as soil conditions and/or environmental hazards/ contaminants and obtain reports with respect thereto (collectively, the "Test Reports"). Consequently, Developer shall have ninety (90) days from the execution of this Agreement by both parties ("Inspection Period"), to conduct such investigations and/or testing as it deems necessary and obtain the Test Reports.
  - Within five (5) days following the expiration of the Inspection Period, Developer shall provide written notice to Owner, together with copies of such Test Reports, indicating whether the Test Reports reveal any conditions which will increase the Contract Time and/or increase the GMP ("Inspection Notice") and requesting the Owner's approval of a Change Order reflecting any such increase in Contract Time and/or increase in the GMP. Owner shall have twenty (20) days from receipt of the Inspection Notice to either (i) provide Developer with a signed Change Order reflecting the requested increase in Contract Time and/or increase in GMP; whereupon this Agreement shall continue, or (ii) provide written notice to Developer ("Owner's Response Notice"), indicating that it will either (a) not change the Contract Time and/or GMP, as requested by Developer, or (b) agree to some extension of Contract Time and/or increase GMP. Developer shall have five (5) days following receipt of Owner's Response Notice to accept the terms of Owner's Response Notice or provide written notice to Owner that it is terminating this Agreement; whereupon, neither party shall have any further obligations under this Agreement.
- 2.1.4 The Project Manager is hereby designated by the City Manager to be Albert Carbon Project Manager, whose address is 100 N. Andrews Ave Fort Lauderdale, FL 33301. Any change to the Project Manager shall be made in writing to the Developer.

- 2.2 The Project Manager shall review and make appropriate recommendations within five (5) days on all requests submitted by Developer for payment for services and work provided and performed in accordance with this Agreement.
- 2.3 Information or services required of the Owner by this Agreement shall be furnished by the Owner within five (5) Business Days after receipt from the Developer of a written request for such information or services, unless otherwise specified in the Contract Documents. Owner's failure to provide information or services within the time period provided shall entitle Developer to an extension of the Contract Time with respect thereto on a day for day basis after expiration of such five (5) Business Day period.
- 2.4 In the event the Developer fails and refuses to correct any Work in accordance with the Contract Documents, within a five (5) Business Day time period after receipt of written notice from the Project Manager with regard to such failure, which is not in accordance with the requirements of the Contract Documents, (provided, however, if such correction cannot be completed within five (5) Business Days, then if Developer commences to cure same within five (5) Business Days and diligently pursues same to completion, then Developer shall have satisfied this Agreement), the City Manager at the City Manager's sole discretion may direct the Developer to stop the Work or any portion thereof, until such time as the non-conforming Work has been corrected.
- 2.5 In the event the Developer fails to address and agree upon corrective action with the Project Manager in accordance with the Contract Documents, within a five (5) Business Day time period after receipt of written notice from the Project Manager with regard to such failure (provided, however, if such correction cannot be completed within five (5) Business Days, then if Developer commences to cure same within five (5) Business Days and diligently pursues same to completion. then Developer shall have satisfied this Agreement), the Owner may, without waiving its rights to pursue any and all other remedies, move forward to correct such deficiencies itself. In that event, the Owner, except as provided for in 1.3, will issue a Construction Change Directive deducting from any and all payments due to the Developer the cost of correcting such non-conforming Work, including any compensation to third parties for additional services and expenses incurred as a result of such failure by the Developer. The method for determining the Construction Change Directive value shall be determined mutually by the City Manager and the Developer. In the event the payments due the Developer are insufficient to pay for the Construction Change Directive referenced above, then the Developer shall immediately pay the difference to the Owner.

14

### ARTICLE 3: DEVELOPER'S RESPONSIBILITIES

### A. <u>Design Services</u>.

- 3.1 Within ten (10) days after Owner's issuance of the Notice to Proceed, the Developer shall perform the Design Services, and shall commence and diligently complete preparation of the Construction Documents. Issuance of the Notice to Proceed shall fix the date on which the Contract Time will commence to run and on which the Contract Time will end. The scope of Work for Design Services shall include the work and services set forth in **Exhibit D**.
- 3.2 In conjunction with Developer's rendition of Design, Developer shall receive the fixed amount of \$2,553,180.00 ("Design Fee") as the total lump sum compensation for all Design Services. Said lump sum amount shall be paid on a percent completion basis, in accordance with a schedule submitted by Developer, and reviewed and approved by Project Manager at the time of the City's issuance of the Notice to Proceed. The Design Fee is comprised of \$977,500.00 for the parking garage component of the Project and \$1,575,680.00 for the swimming pool component of the Project.

### B. <u>Construction Services</u>.

- 3.3 The scope of Work for Construction Services shall include the work and services set forth in **Exhibit E**, which is attached hereto.
- 3.4 With respect to the Construction Services to be provided by Developer hereunder, Owner shall reimburse Developer for the Cost of the Work and pay Developer a fixed Developer's Fee ("Developer's Fee") as set forth in the GMP for the Construction Services. Developer agrees to provide Owner with a Guaranteed Maximum Price Proposal for the total sum of the Developer's Fee plus the Cost of the Work broken down into the sub-trade proposals and other construction categories and level of detail reasonably required by Owner. Developer agrees that all of its books, records and files, with respect to its presentation of the Guaranteed Maximum Price Proposal, shall be open to Owner for review and copying during normal business hours and subject to the terms of Section 3.14 hereof.

In the event of any termination by Owner of Developer, the Developer shall promptly provide all materials and information relating to the Project to Owner in appropriate formats and expressly grants the Owner the right to use the Design documents to contract for the construction phase of the work and the Developer will be considered the Architect/Engineer of Record for such purposes. Developer's agreement for Design Services from the Architect of Record and Engineer of Record shall provide their joinder and consent to the Owner's use of the Design documents. The Developer will not be entitled to any additional compensation in the event of such termination and compensation paid as the Design Fee (through the date of any such termination) shall be considered full compensation for the Owner's use of the Design Document for construction.

- 3.5 <u>Costs to be Reimbursed</u>. Owner shall pay Developer the Cost of the Work for the Construction Services. Except as otherwise provided herein, the Cost of the Work for Construction Services which Owner shall pay Developer shall include only those items set forth below in this section 3.5:
  - 3.5.1 Labor Costs.

3.5.1.1

3.5.1.2

3.5.1.3

- Wages of construction workers directly employed by Developer to perform the construction of the Work at the Project site or, with Project Manager written agreement, at off-site workshops. Costs to be reimbursed will be the actual wages paid inclusive of labor burden to the individuals performing the work.
  - Wages or salaries of Developer's supervisory and administrative personnel who are stationed at the Project site which includes Schedule 3.10.3 – Basis for GMP, and all such personnel listed on **Exhibit B-4**. Costs to be reimbursed will be per Division 1, General Conditions Schedule and will be the actual wages paid inclusive of labor burden to the individuals performing the work.

The parties hereby establish the fixed markup rate of twentynine percent (29%) for all labor burden, including all taxes, insurance, contributions, assessments and benefits required by law and collective bargaining agreements and, for personnel not covered by such agreements, customary benefits such as sick leave, medical and health benefits, holidays, vacations and pensions, provided such markup is to apply only upon those wages and salaries included in the Cost of the Work under subsections 3.5.1.1 and 3.5.1.2, above, but in any event excluding bonuses and other similar extraordinary benefits.

- 3.5.2 <u>Subcontract Costs</u>. Payments made or due by Developer to subcontractors in accordance with the requirements of the applicable written subcontracts.
- 3.5.3 <u>Cost of Materials and Equipment Incorporated into the Completed</u> <u>Construction</u>.
  - 3.5.3.1 Costs, including transportation and storage of materials and equipment incorporated or to be incorporated in the completed construction.

3.5.3.2

Costs of materials described in subsection 3.5.3.1, above, in excess of those actually installed but required to provide reasonable allowance for waste and for spoilage, such reasonable allowance to be mutually agreed upon by the parties. Unused excess materials, if any, shall be handed over to Owner at the completion of the Work or, at Owner's option, shall be sold by Developer; amounts realized, if any, from such sales, shall be credited to Owner as a deduction from the Costs of the Work.

# 3.5.4 <u>Costs of other materials and equipment, temporary facilities and related</u> items.

3.5.4.1

Costs, including transportation, storage, installation, maintenance, dismantling and removal, of materials, supplies, temporary facilities (including project field offices, furniture and fixtures), temporary utilities, machinery, equipment, and hand tools not customarily owned by the construction workers, which are provided by Developer at the Project site and fully consumed in the performance of the Work; and costs less salvage value on such items if not fully consumed, whether sold to others or retained by Developer.

3.5.4.2

Rental charges, at standard industry rates for the area, for temporary facilities, machinery, equipment, and hand tools not customarily owned by the construction workers, which are provided by Developer at the Project site, whether rented from Developer or others, and costs of transportation, installation, minor repairs and replacements, dismantling and removal thereof. Rates and quantities of all equipment rented, whether from Developer or others, shall be detailed in the Final GMP Contract provided for in **Exhibit B**.

3.5.4.3 Cost of removal and proper legal disposal of debris from the Project site.

3.5.4.4

Costs of document reproduction (limited to \$.15/page), parcel delivery charges and telephone service at the Project site and reasonable petty cash expenses of the Project site office.

3.5.4.5

That portion of the reasonable travel and subsistence expenses of Developer's personnel, assigned to the Project site, incurred while traveling outside of the Fort Lauderdale/Broward County metropolitan area in discharge of duties connected with the Work, detailed in the Final GMP Contract provided for in **Exhibit B.** 

### 3.5.5 <u>Miscellaneous Costs</u>.

3.5.5.1 That portion of any premiums for (i) bonds directly attributable to this Contract and (ii) any additional insurance coverages which are purchased by Developer, detailed in the Final GMP Contract provided for in **Exhibit B**.

# 3.5.5.2 Sales, use or similar taxes imposed by a governmental authority which are related to the Work and for which Developer is liable.

3.5.5.3 Fees and assessments for the Building Permit and other permits, licenses and inspections for which Developer is required by the Contract Documents to pay and impact fees, as listed in Schedule of Values as an Allowance.

- 3.5.5.4 Fees of testing laboratories for tests required by the Contract Documents, except those related to defective or nonconforming Work for which reimbursement is excluded pursuant to the terms of this Contract.
- 3.5.5.5 Royalties and license fees paid for the use of a particular design, process or product required by the Contract Documents.
- 3.5.5.6 Deposits lost for causes other than Developer's fault or negligence.
- 3.5.5.7 Legal, mediation and arbitration costs, including reasonable attorneys' fees, other than those arising from disputes between Owner and Developer, reasonably incurred by Developer in performance of the Work and with City Manager's prior written consent, which consent shall not be unreasonably withheld.
- 3.5.5.8

Costs reasonably incurred in repairing or correcting damage or nonconforming Work executed by Developer, or its subcontractors, subconsultants, or suppliers, provided that such damage or nonconforming Work was not caused by (i) the negligence or failure to fulfill a specific responsibility of Developer to Owner set forth in the Contract Documents, or (ii) Developer's foremen, engineers, superintendents or other supervisory, administrative or managerial personnel, (iii) the failure of Developer's personnel to supervise adequately those portions of the Work to be performed by Developer's subcontractors, subconsultants or suppliers, and only to the extent that the cost of repair or correction is not recoverable by Developer from (i) insurance or bonds, or (ii) any of the subcontractors, subconsultants, or suppliers.

- 3.5.6 <u>Other Costs</u>. Other costs incurred in performance of the Work if and to the extent approved in advance in writing by Owner, which approval shall not be unreasonably withheld.
- 3.5.7 Costs incurred in taking action to prevent threatened damage, injury or loss in case of an emergency affecting safety of persons or property, such as, without limitation, properly securing the Project Site in the event of a threatened hurricane or tropical storm and repairing any resulting damage therefrom. Extensions of Contract Time, extended General Conditions to Developer on account of delays to Contract Time and costs incurred due to such threatened harm or emergency shall be mutually and reasonably agreed upon by Owner and Developer.
- 3.6 <u>Costs Not To Be Reimbursed</u>. The Cost of the Work shall not include the following items:
  - 3.6.1 Salaries and other compensation of Developer's personnel stationed at Developer's principal office or offices other than the Project Site office, except as noted in 3.5.1.2.
  - 3.6.2 Expenses of Developer's principal office and offices other than the Project Site office.
  - 3.6.3 Overhead and general expenses, except as may be expressly included in section 3.5 above or otherwise in this Agreement.
  - 3.6.4 Developer's capital expenses, including interest on Developer's capital employed for the Work.
  - 3.6.5 Except as expressly provided in subsection 3.5.5.8 above, costs due to the fault or negligence of Developer, subcontractors, anyone directly or indirectly employed by any of them, or for whose acts any of them may be liable, including, but not limited to, costs for the correction of damaged, defective, or nonconforming Work, disposal and replacement of materials and equipment incorrectly ordered or supplied, and making good damage to property not forming part of the Work.
  - 3.6.6 Any costs not specifically and expressly described in GMP Proposal or Section 3.5 or otherwise in this Agreement.
  - 3.6.7 Costs which would cause the GMP to be exceeded, except those costs included in approved Change Orders and Construction Change Directives (as the GMP may be adjusted pursuant to the terms herein for Change

Orders and Construction Change Directive) or as otherwise provided in this Agreement.

- 3.7 <u>Discounts, Rebates and Refunds</u>. Cash discounts obtained on payments made by the Developer shall be added to the Contingency (as hereinafter defined). Trade discounts, rebated, refunds, and amounts received from sales of surplus materials and equipment shall be added to the Contingency.
- 3.8 <u>General</u>. The construction phase shall commence within ten (10) days after the latest to occur of the issuance of all Building Permits necessary to commence construction and the Owner's written authorization to Developer to proceed with construction. Notwithstanding anything contained herein to the contrary, in no event will Developer be obligated to commence construction of the Project unless and until the costs for the Design Services and Construction Services (together with, the Design Fee and Developer's Fee) under the approved Construction Documents do not exceed the GMP, in Developer's sole discretion, and the Contract Time shall be extended accordingly, if necessary.
- 3.9 The Developer shall cause all Work required by the Contract Documents to be properly completed in accordance with the terms of the Contract Documents and within the Contract Time, subject to the GMP and adjustment of the Contract Time, as provided in this Agreement.

### 3.10 Guaranteed Maximum Price (GMP)

THE GUARANTEED MAXIMUM PRICE (GMP) CANNOT EXCEED (1) **TWENTY-FOUR MILLION EIGHT-HUNDRED SIXTY-FOUR THOUSAND NINE-HUNDRED AND FIFTY DOLLARS** (\$24,864,950.00) FOR THE PUBLIC IMPROVEMENTS (AS DELINEATED ON THE PROJECT PROGRAM SUBMITTED TO THE CITY OF FORT LAUDERDALE ON MARCH 28, 2012) AND (2) **SEVEN MILLION FIVE-HUNDRED SEVENTY-TWO THOUSAND FOUR-HUNDRED AND EIGHTY-FOUR DOLLARS** (\$7,572,484.00) FOR THE PARKING GARAGE, BOTH ELEMENTS INCLUSIVE OF THE COSTS FOR DESIGN SERVICES, DESIGN FEE AND DEVELOPER'S FEE, OR A TOTAL GMP OF THIRTY-TWO MILLION, FOUR HUNDRED THIRTY-SEVEN THOUSAND, FOUR HUNDRED THIRTY-FOUR DOLLARS (\$32,437,434.00).

# ONCE THE CONSTRUCTION PHASE HAS COMMENCED, DEVELOPER AGREES THAT IT WILL BE RESPONSIBLE FOR PAYING ALL COSTS OF COMPLETING THE WORK WHICH EXCEED THE GMP, AS MAY BE ADJUSTED IN ACCORDANCE WITH THE TERMS OF THIS AGREEMENT.

3.10.1 Subject to the provisions of Section 3.8 hereof, when the Construction Documents are 90% complete, Developer will provide the final cost detail to support the GMP, which is the total not-to-exceed estimate of the Cost of Work, cost of Design Services, the Design Fee and the Developer's Fee, and the form of which is attached hereto as **Exhibit B**. The GMP shall include all labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and any other services necessary for the proper execution and completion of the Work.

- 3.10.2 After the City's approval of 90% Construction Documents, the Construction Documents shall not be modified to include such things as changes in scope, systems, kinds and quality of materials, finishes or equipment. If these modifications are required, all of which, if required, shall be incorporated by Change Order and shall increase the Contract Time and/or GMP, as appropriate.
- 3.10.3 <u>Basis of Guaranteed Maximum Price</u>. The Developer shall include as Exhibit B to this Agreement, a written statement of its basis for the GMP, which shall include:
  - 3.10.3.1 A list of the Drawings and Specifications, including all addenda thereto and the conditions of the Contract Documents, which were used in preparation of the GMP.
  - 3.10.3.2 A list of allowances and a statement of their basis.
  - 3.10.3.3 A list of the clarifications and assumptions made by the Developer in the preparation of the GMP to supplement the information contained in the Drawings and Specifications
    - 3.10.3.4 The GMP, including a statement of the estimated cost organized by trade categories, allowances, contingency, and other items and the Developer's Fee that comprise the GMP.
    - 3.10.3.5 The Date of Substantial Completion upon which the GMP is based, and a schedule of the Construction Documents' issuance dates upon which the date of Substantial Completion is based.

Notwithstanding anything contained herein to the contrary, the Owner has acknowledged and agreed to the qualifications, clarifications and conditions set forth in the Basis for the GMP attached hereto and made a part hereof as <u>Schedule 3.10.3</u>.

3.10.4 Included within the GMP is the Developer's Fee. The Developer's Fee is hereby established as \$4,710,577.80 based on a fixed percentage of 17.78% for each of the parking garage component and for the swimming pool component of the Project (the "Developer's Fee"), calculated on all amounts reflected in the GMP Sum, more particularly described in Exhibit "B," excluding permits, payment and performance bond, builder's risk insurance, project specific E & O policy, Developer's Contingency and Owner's Contingency. The sum of the Cost of the Work, cost of Design Services, the Design Fee and the Developer's Fee shall not exceed the GMP also referred to as the Contract Sum. The Developer's Fee shall constitute Developer's total compensation for profit and overhead for Construction Services. The Developer's Fee is to be included in the Developer's monthly Application for Payment, in accordance with Article 9.1.4 of this Agreement, and shall be proportional to the percentage of Work completed, less payments of the Developer's Fee previously made.

- 3.10.5 The GMP shall be subjected to additions and deductions by changes in the Work, as provided in the Contract Documents and the date of Substantial Completion shall be subject to adjustment, as provided in the Contract Documents.
- 3.10.6 The GMP Sum is guaranteed by the Developer not to exceed the amount provided in **Exhibit B** to this Agreement, subject to additions and deductions by changes in the Work, as provided in the Contract Documents.
- 3.10.7 The Developer acknowledges that this Agreement is to be administered on an "open book" arrangement relative to the cost of the Work and approved subcontract agreements; provided, however, that Owner shall not unreasonably withhold or delay approval to such subcontract agreements.
- 3.10.8 Included within the GMP is the Developer's general conditions costs ("General Conditions"), which shall be referenced as a separate line item in the Schedule of Values. The Developer's General Conditions are to be included in Developer's monthly Application for Payment, in accordance with Article 9.1.4 of this Agreement, and shall be proportional to the percentage of Work completed, less payments of General Conditions previously made. Notwithstanding anything contained herein to the contrary, the General Conditions are a fixed amount and shall not be subject to audit or shared "Savings", as hereinafter defined.
- 3.10.9 Included within the GMP will also be a Contingency line item as follows:
  - 3.10.9.1

The parties agree that, given the size, scope, and nature of the Project, unforeseen contingencies may arise during the course of construction. The Schedule of Values will contain line items for a Developer's Contingency (as hereinafter defined) and an Owner's Contingency (as hereinafter defined), such contingencies being referred to collectively as the "Contingency Funds". The Developer's Contingency will consist of (i) one line item labeled Developer's Aquatic Contingency in the amount of \$237,897.78; and (ii) one line item labeled Developer's Parking Contingency in the amount of \$217,871.22 (collectively, the "Developer's Contingency"). The Owner's Contingency will consist of (i) one line item labeled Owner's Aquatic Contingency in the amount of \$150,000.00; and (ii) one line item labeled Owner's Parking Contingency in the amount of \$100,000.00 (collectively, the "Owner's Contingency"). There will also be a line item for Savings (as defined in Section 9.1.12) in which any value engineering savings, material cost savings, subcontractor cost reductions and buy-out achieved during the course of the Project will be deposited. During the course of construction, Developer shall, in accordance with the Pay Application mechanism outlined in Section 9, have the right to draw down on the Contingency Funds to pay for and/or otherwise address the aforesaid unforeseen contingencies, in the order of priority set forth in this Section 3.10.9.1. For the purposes of this Agreement, items for which the Contingency Funds may be used include unforeseen items, including, but not limited to, conflicts in the Design Documents or Construction Documents or Specifications, governmental or building code requirements not known to Developer during the Design Phase, unforeseen field or other conditions, extended General Conditions and additional repair and construction costs based on Force Majeure events, unanticipated escalation of materials, subsurface conditions and other unforeseen conditions not otherwise identified or reasonably ascertained at the time of execution of this Agreement (collectively, "Unforeseen Contingencies"); however, the Contingency Funds shall not be used to pay for Owner's requested changes to the Work which would be addressed in a Change Order increasing the GMP and/or Contract Time, as appropriate. Notwithstanding anything contained herein to the contrary, the Developer's Contingency can be used by the Developer to pay for and/or address not only the Unforeseen Contingencies, but also to pay for and/or address the following (collectively. "Developer's Additional Contingencies"): (a) construction costs (including, labor, equipment and materials) resulting from errors or omissions in the Design Documents, Construction Documents or Specifications; provided, however, no such contingency funds shall be used to pay for additional design costs relating to such error or omission; and (b) costs of acceleration of construction to remain on the agreed upon schedule. For the purposes of this Section 3.10.9.1, the order of priority in which the Contingency Funds may be drawn upon shall be as follows: Developer's Contingency first, then Savings (as defined in Section 9.1.12) second, and then Owner's Contingency.

The parties agree that to the extent either identifies an Unforeseen Contingency, the party identifying such condition and seeking to have that condition satisfied from the contingency line item, shall give written notice to the other party within ten (10) days of identifying the condition and quantifying the amount to be applied to the with contingency line item, reasonable backup documentation supporting such request ("Contingency Notice"). The party receiving the Contingency Notice shall, within five (5) days of receipt of the Contingency Notice, provide written affirmation that the amount sought to be applied to the contingency line item is acceptable, or alternatively, provide written notice of all objections to the Contingency Notice to the other party. If the parties are unable to resolve any disagreement relating to the Contingency Funds within ten (10) days thereafter, the parties shall address this matter as with all other Claims outlined in this Agreement. Notwithstanding anything in the foregoing to the contrary, the use of the Developer's Contingency by the Developer shall not require the Owner's approval; provided, however, that such funds are being used to pay for and/or address Unforeseen Contingencies or Developer's Additional Contingencies, as such terms are defined in Section 3.10.9.1. The use of the Owner's Contingency and the Owner's share of the Savings by Developer shall be subject to the prior approval of the Owner.

3.10.9.3

3.10.9.2

Notwithstanding anything contained herein to the contrary, any amounts remaining in the Contingency Funds at the conclusion of the Project shall be allocated between the parties as follows: (i) all amounts remaining in the Developer's Contingency shall be retained by the Developer; and (ii) all amounts remaining in the Owner's Contingency shall be retained by the Owner. Any amounts remaining in Savings shall be divided between the parties in accordance with the Savings clause in Section 9.1.12.

3.10.10 The parties agree that the Developer shall be the commercial pool contractor for the Project and the pool pricing contained in the GMP shall be subject to verification at 90% completion of the Construction Documents by a cost estimator retained by the Owner. In such capacity as the pool contractor, for the purposes of this Agreement, Developer shall be subject to the terms and provisions of this Agreement relating to subcontractors.

3.11 <u>Changes in Work</u>. Adjustments to the GMP because of changes in the Work shall be determined by any of the methods listed in Article 7 of this Agreement.

All tests, inspections and approvals of portions of the Work required by the 3.12 Contract Documents or any applicable laws, ordinances, rules, regulations or orders of public authorities having jurisdiction, shall be made at an appropriate time. The Developer shall be responsible for making arrangements for such tests, inspections and approvals with an independent testing laboratory acceptable to the Owner's Project Manager, or with the appropriate public authority. In the event the Owner or public authority having jurisdiction determine that portions of the Work require additional testing, inspection or approval, the Owner's Project Manager must approve such additional testing, inspection or approval in writing. The Developer shall make arrangements for such additional testing, inspection or approval, and give timely prior notice to the Owner's Project Manager. In the event the testing, inspection or approval described in this subparagraph reveals the failure of portions of the Work to be in compliance with the requirements of the Contract Documents, the Developer shall bear all costs of such non-conforming Work, including the additional testing, inspection or approval. Retesting costs for additional testing occasioned by the failure of portions of the Work to be in compliance with the Contract Documents are not to be included in the reimbursable Cost of the Work.

3.13 The Developer shall obtain and deliver promptly to the Owner's Project Manager any occupancy permit and/or any certificates of final inspection of any part of the Developer's Work and operating permits for any mechanical apparatus, such as elevators, escalators, boilers, and air compressors, which may be required by law to permit full use and occupancy of the premises by the Owner. Receipt of such permits or certificates by the Owner's Project Manager shall be a condition precedent to Substantial Completion of the Work unless the certificates stated herein are withheld for reasons that are neither the fault nor the negligence of the Developer.

# C. <u>Other Responsibilities</u>.

### 3.14 Accounting Records and Audit.

3.14.1 The Developer shall keep in electronic form full and detailed accounts and exercise such controls as may be necessary for proper financial management under this Contract; the accounting and control systems shall be Timberline accounting/Estimating software. During construction and until final payment by Owner hereunder, the Owner, including Owner's employees, officer, agents and accountants shall be afforded reasonable access to the Developer's records, books, correspondence, instructions, Drawings, receipts, subcontracts, purchase orders, vouchers, memoranda, and other data relating to this Project for review and audit purposes during normal business hours and on five (5) days written notice to Developer. The Developer shall preserve in electronic form these records, documents and data for a period of five fiscal years after final payment, or for such longer period as may be required by Florida law. If the Developer receives notification of a dispute or the commencement of litigation regarding the Project within this five-year period, the Developer shall continue to maintain all Project records until final resolution of the dispute or litigation.

3.14.1.1

The Developer shall furnish the Owner with one electronic copy of all financial records as to the costs to be reimbursed under Article 3.5 regarding the Project, the receipt of which by the Owner is required prior to release of the final payment.

- 3.14.2 Upon five (5) days' written notice, from the date of this Agreement to the latest date described in Section 3.14.1 for record preservation, the Developer shall make its Project records available during normal business hours to the Owner or its authorized representative(s) at the Developer's office. Owner and its authorized representative(s) shall be entitled to inspect, examine, review and copy the Developer's Project records at the Owner's reasonable expense, within adequate work space at the Developer's facilities. Failure by the Developer to supply substantiating Project records shall be reason to exclude the related costs from amounts which might otherwise be payable by the Owner to the Developer pursuant to this Agreement.
- 3.14.3 This Article 3.14, "Accounting Records and Audit," shall survive the termination of this Agreement.
- 3.15 The Developer is solely responsible for reviewing and comparing the Contract Documents with each other and within five (5) days of discovery, report to the Owner any and all errors, inconsistencies or omissions. In the event the Developer performs any Work with the knowledge that it involves an error, inconsistency or omission in the Contract Documents, the Developer shall be responsible for such erroneous, inconsistent, or omitted Work. The Developer is also responsible for taking such field measurements as are necessary in order to verify field conditions and to compare such field measurements and conditions with the Contract Documents. Any and all errors, inconsistencies or omissions shall be reported to the Owner within five (5) days of discovery.
- 3.16 The Developer shall be solely responsible for supervising and directing the Work, and shall have sole responsibility for determining appropriate construction means, methods, techniques, sequences and procedures, and for coordinating

the Work under the Contract Documents. The Developer shall be solely responsible to the Owner for the acts and omissions of all Persons performing or supplying any portion of the Work for which the Developer has contracted.

- 3.17 Warranty. For a period of one (1) year from the date of Final Completion of the Work, the Developer hereby warrants to the Owner that all materials and equipment furnished under the Agreement will be of good quality and new, and that the Work performed will be free from any and all defects (except for those inherent in the quality or nature of the Work or any of the materials incorporated therein) and will be in conformity with the requirements of the Contract Documents. Interpretation of the Contract Documents shall be made upon the basis that any Work to be done and materials to be furnished shall be regarded as meaning that only the best general practice is to prevail and that only material and workmanship of the best quality is to be used. All Work not conforming to these requirements may be declared defective by the Owner. If requested, the Developer shall furnish evidence to the satisfaction of the Owner of the quality of the materials and equipment supplied. The warranty provided in this paragraph shall be in addition to and not in limitation of any other warranty required by the Contract Documents or otherwise prescribed by law. Prior to Final Payment the Developer shall procure and deliver to the Owner all special warranties required by the Contract Documents.
- 3.18 In no case shall a substitution be granted where the Owner requires a sole source item to match existing standards.
- 3.19 In requesting approval of deviations or substitutions, the Developer shall provide evidence leading to a reasonable certainty that the proposed substitution or deviation will provide a quality of result at least equal to that otherwise attainable. If, in the opinion of the Owner, the evidence presented by the Developer does not provide a sufficient basis for such reasonable certainty, the Owner may reject such substitution or deviation without further investigation. The Owner shall not unreasonably reject the request of the Developer.
- 3.20 The Contract Documents are intended to produce a Project of consistent character and quality of design. All components of the Project including visible items of mechanical and electrical equipment must have a coordinated design in relation to the overall appearance of the Project. The Owner shall judge the design and appearance of proposed substitutes on the basis of their suitability in relation to the overall design of the Project, as well as for their intrinsic merits. The Owner will not approve as equal to materials specified proposed substitutes which, in the Owner's opinion, would be out of character, obtrusive, or otherwise inconsistent with the character or quality of design of the Project. In order to permit coordinated design of color and finishes the Developer shall, if required by the Owner, furnish the substituted material in any color, finish, texture, or pattern which would have been available from the manufacturer originally specified, at no additional cost to the Owner.

- 3.21 The Developer shall pay all applicable sales, consumer, use and similar taxes for the Work which are legally required, except on items the Owner chooses to utilize its Direct Purchase Program; provided, however, that same shall be reimbursable Costs of Work per Section 3.5.5.2.
- 3.22 City permits are required on City property. The Developer will pay all applicable permit, utility, transportation concurrency fees, FP&L and impact fees attendant to the Work; provided, however, that same shall be reimbursable, and included as Allowances under the Schedule of Values.
- 3.23 In-progress inspections by the Owner or representatives employed by the Owner will check for compliance with applicable codes and monitor the course of construction on the Owner's behalf. Owner's inspectors shall at all times have access to the Work during normal work hours and Developer shall provide proper facilities for such access and for inspecting, measuring and testing. This inspector may call to the Developer's attention, Work which is considered to be not in accordance with the Drawings and Specifications, and therefore unacceptable. Such inspections shall not be deemed an approval of any work of the Developer or others nor shall such inspections in any manner reduce or limit Owner's rights or remedies in regard to Developer non-compliance with the terms of this Agreement. The Developer will either repair or replace such construction, or appeal to the Owner for a ruling. The City Manager will be the final authority as to acceptability of Work, and the inspector will not by inference. be accepting Work on the Owner's behalf. The inspector may from time to time request changes in the Work which will enhance the job or remove an undesirable condition. In such instances, the request will be in writing to the Developer.
- 3.24 The Developer shall comply with and give notices required by all applicable laws, ordinances, rules, regulations and lawful orders of public authorities with regard to the performance of the Work. In the event the Developer observes that any portions of the Contract Documents are not in compliance with the above, the Developer shall, within five (5) days of discovery, notify the Owner's Project Manager in writing of such variances. In the event the Developer performs any Work with knowledge that it is at variance with applicable laws, ordinances, rules, regulations and lawful orders of public authorities without giving such notice, then the Developer shall be responsible for the attributable costs for such Work.
- 3.25 The Developer shall employ a competent superintendent, reasonably acceptable to the City Manager, and necessary assistants who shall be in attendance at the Project Site full time during the progress of the Work until the date of Substantial Completion, and for such additional time thereafter as the Owner and Developer both agree to be necessary for the expeditious completion of the Work; provided, however, no such superintendent shall have the authority to bind Developer. The Developer's Representative for the purposes of having authority

to act on behalf of Developer and for the purposes of making decisions to bind Developer shall be Joseph Cerrone, subject to change by written notice to Owner. The Developer shall remove the superintendent if requested to do so in writing by the City Manager for reasonable cause, and shall promptly replace her/him with a competent person reasonably acceptable to the City Manager.

- 3.26 The Developer shall retain a competent registered professional engineer or registered land surveyor, who shall establish the exterior lines and required elevations of all buildings and structures to be erected on the Project Site and shall establish sufficient lines and grades for the construction of associated Work such as, but not limited to, roads, utilities and site grading. The engineer or land surveyor shall certify the actual location of the constructed improvements and facilities in relation to property lines, building lines, easements, vertical datum and other restrictive boundaries.
- 3.27 The Developer shall establish in the Project Documents the building grades, lines, levels, column, wall and partition lines required by the various subcontractors in laying out their Work.
- 3.28 The Developer shall arrange for and attend job meetings with the Owner's Project Manager and such other persons as the City Manager may from time to time wish to have present. The Developer shall be responsible for recording and distributing meeting minutes. The Developer shall be represented by a principal, project manager, general superintendent or other authorized main office representative, as well as by the Developer's own superintendent. An authorized representative of any subcontractor or sub-subcontractor shall attend such meetings if the representative's presence is required by the City Manager. Such representatives shall be empowered as appropriate on all matters to be discussed at such meetings, including costs, payments, change order, time schedules, manpower and Construction Change Directive, unless the above listed items require written approval by the Owner and the Developer.
- 3.29 The superintendent as designated by the Developer and accepted by the Owner shall not be reassigned from the job prior to final completion, except with the approval of the Owner's Project Manager. In the event of separation or termination of employment of Developer's superintendent, the Developer may assign a new superintendent subject to the reasonable approval of Owner.
- 3.30 The Developer shall be required to prepare a schedule of submittals for the Owner's approval which is coordinated with the construction schedule, allowing sufficient time for review. The submittals shall reference progress schedule dates for installation and Specification section. The Owner shall be provided with one copy of the accepted schedule of submittals for the Owner's records.
- 3.31 The Developer shall maintain at the office one record copy of the Drawings, Specifications, addenda, Change Orders and any other modifications, which serve

to indicate all current changes, in addition to all approved shop drawings, product data samples, and all other similar submittals. The Developer shall maintain "as-built" drawings at the Project Site office. The "as-built" drawings shall document all changes in the Drawings and Specifications made during the course of the Work. The "as-built" drawings shall be in sufficient form and detail to permit the Owner to maintain and repair the Work after its completion. At the completion of the Work, these items shall be delivered to the Owner prior to release of final payment.

3.32 The Developer shall review, approve and submit to the Owner any and all shop drawings, product data, samples and any other similar submittals which are required by the Contract Documents, and allow a reasonable amount of time for review by the Owner, without impacting the Work schedule, but in no event more than five (5) days after delivery to Owner. The Developer shall not perform any part of the Work relating to the shop drawings, product data, samples or any other similar submittals, until such items have been approved by the Owner within five (5) days. Notwithstanding anything contained herein to the contrary, Contractor shall be entitled to an extension of the Contract Time (on a day for day basis) and any additional General Conditions relating to any delay in Owner's approval which affects the critical path.

3.33 The Owner's approval by its Project Manager of any shop drawing, product data, sample or other similar submittal does not in any way relieve the Developer of responsibility for deviations from the requirements of the Contract Documents. Further, the Developer shall not be relieved of responsibility for any errors or omissions in the shop drawings, product data, samples or any other similar submittals simply by the Owner's approval. In the event a deviation is requested, the Developer shall specifically identify the deviation in writing to the Owner's Project Manager at the time of the submittal and the Owner's Project Manager is required to respond in writing to approve the specified deviation within five (5) days.

3.34 By approving and submitting shop drawings, product data, samples, and similar submittals, the Developer represents that the Developer has determined and verified all dimensions, quantities, field dimensions, relations to existing Work, coordination with Work to be installed later, coordination with information on previously accepted shop drawings, product data, samples, or similar submittals and verification of compliance with all the requirements of the Contract Documents. The accuracy of all such information is the responsibility of the Developer.

3:35 When professional certification of performance criteria of materials, systems or equipment is required by the Contract Documents, the Owner shall be entitled to rely upon such certifications, and the Owner shall not be expected to make any independent examination with respect thereto.

> Exhibit 1 CAM 15-1467 Page 30 of 158

- 3.36 The right of possession of the premises and the improvements made thereon by the Developer shall be retained at all times by the Owner. The Developer's right to enter arises solely from the permission granted by the Owner under the Contract Documents. The Developer shall confine the Developer's equipment, the storage of materials and the operations of the Developer's workmen to the Project Site and according to the directions of the Owner's Project Manager or City Manager, and shall not unreasonably encumber the Project Site with the Developer's materials.
- 3.37 The Developer shall keep the Project Site, the surrounding area, surrounding rights-of-way and properties free from all waste, construction debris, or trash. At the completion of the Work, the Developer shall remove all tools, construction equipment, machinery and surplus materials. In the event the Developer fails to keep the Project Site, surrounding area, surrounding rights-of-way and properties in a clean condition, then upon three (3) days' written notice to Developer and failure of Developer to correct such condition, then the Owner may do so and charge the cost back to the Developer.
- 3.38 <u>Developer Use of Project Site</u>: The Developer shall confine construction equipment, the storage of materials and equipment and the operations of workmen to areas permitted by law, ordinances, permits and the requirements of the Contract Documents, and shall not encumber the premises with construction equipment or other materials or equipment. Notwithstanding anything to the contrary contained herein, Owner shall allow Developer to utilize D.C. Alexander Park as a construction staging area; provided, however, that Developer agrees to restore the property to its pre-construction condition upon Developer's completion of the Project.

The Developer shall not enter upon private or public property (not constituting the Project Site) for any purpose without first securing the written permission of the property owner, furnishing the Project Manager with a copy of said permission. This requirement will be strictly enforced, particularly with regard to such properties as may be utilized for storage or staging by the Developer. The use of such off-site property for storage or staging must be permissible under the ULDR.

The Developer shall conduct all Work in such a manner as to avoid and prevent damage to adjacent private or public property. Any damage to existing structures or property of any kind, including permanent reference markers or property corner markers, or the interruption of a telephone, telecommunications or utility service caused by Developer or its subcontractors, shall be repaired or restored by the Developer as soon as reasonably possible in light of the extent of the damages and/or repairs, at Developer's expense and no expense to the Owner. The Developer will preserve and protect all existing vegetation such as trees, shrubs and grass on or adjacent to the Project Site which do not interfere with the Work, as determined by the Project Manager. The Developer will be responsible for repairing or replacing anything damaged by the Developer, including, but not limited to, any trees, shrubs, lawns, landscaping and structures that may be damaged due to operation of equipment, stockpiling of materials, tracking of grass by equipment or other Work activity. The Developer will be liable for, and will be required to replace or restore at no expense to the Owner all structures and vegetation not protected or preserved as required herein that may be destroyed or damaged.

During the progress of the Work, the Developer shall keep the Project Site free from accumulations of waste materials, rubbish and debris resulting from the Work. Upon the completion of the Work, or more frequently if so directed by the Project Manager, the Developer shall remove all waste materials, rubbish and debris from and about the premises as well as all tools, appliances, construction equipment and machinery, and surplus materials, and shall leave the site clean and ready for occupancy by the Owner. The Developer shall restore to their original condition those portions of the Project Site not designated for alteration by the Contract Documents at no cost to the Owner.

- 3.39 Immediately prior to the Owner's inspection for Substantial Completion, the Developer shall completely clean the Project Site utilizing a licensed cleaning service. Concrete and ceramic surfaces shall be cleaned and washed. Resilient coverings shall be cleaned, waxed and buffed. Woodwork shall be dusted and cleaned. Sash, fixtures and equipment shall be thoroughly cleaned. Stains, spots, dust, marks and smears shall be removed from all surfaces. Hardware and all metal surfaces shall be cleaned and polished. Glass and plastic surfaces shall be thoroughly cleaned by professional window cleaners. All damaged, broken or scratched glass or plastic shall be replaced by the Developer at the Developer's expense.
- 3.40 <u>Responsibility for Work</u>. Notwithstanding any requirements herein for Owner's review, inspection or approval, the parties acknowledge and agree that Developer shall be solely responsible and liable for the performance of the design and construction of the Work in accordance with the Contract Documents.
- 3.41 <u>Notices and Compliance with Law</u>. Developer shall be responsible for giving all notices and for complying with all laws, ordinances, rules, regulations and lawful orders of any public authorities having jurisdiction over the particular portion of the Project with respect to the performance of the subject Work. The Master Project Schedule for the subject Work and the GMP associated therewith shall be based upon the laws, ordinances and regulations which are in effect on the date of this Agreement. Any changes in laws, ordinances or regulations thereafter that require additional work outside Developer's established scope shall be the subject of a Change Order under Article 7 of this Agreement.

- 3.42 <u>Indemnification for Infringement</u>. Developer shall pay all royalty and license fees required for the design and construction of any portion of the Project assigned to it. To the maximum extent permitted by law, Developer shall defend any and all suits or claims for infringement of patent rights and shall indemnify and save Owner harmless from all loss or expense on account thereof (including attorneys' and paralegals' fees); provided, however, Developer shall not be responsible for such defense or loss when a particular design, process or product is required by Owner.
- 3.43 <u>Design Services</u>. Developer shall provide the architectural and engineering design for the Project. The design for the Project shall be set forth in the Construction Documents. All changes, supplements and additions to the Construction Documents shall be subject to Owner's review within five (5) days and written approval, such approval to be obtained prior to the commencement of any portion of the Work relating thereto. Notwithstanding anything contained herein to the contrary, Contractor shall be entitled to an extension of the Contract Time (on a day for day basis) relating to any delay in Owner's approval which affects the critical path.

3.43.1 The precedence of documents is as follows:

List of Clarifications and Assumptions listed in 3.10.3.3. List of Project Plans and accompanying Specifications 3.10.3.1 City of Fort Lauderdale Construction Standards and Specifications The GMP as listed in 3.10.3.4 This Agreement

- 3.44 <u>Review, Recommendations and Warranty</u>. Developer shall familiarize itself thoroughly with the evolving architectural, civil, mechanical, plumbing, electrical and structural Plans and Specifications being prepared by the Designer and shall follow the development of the Project design through all required Design activities. Developer shall make recommendations with respect to the selection of systems and materials, and cost-reducing alternatives including assistance to Owner in evaluating alternative comparisons versus long term cost effects. The evaluation shall address the benefits of the speed of erection and early completion of the Work. Developer shall furnish pertinent information as to the availability of materials and labor that will be required. Developer shall submit to Owner such comments as may be appropriate concerning construction feasibility and practicality. Developer shall call to Owner's attention any defects in the design, drawings and specifications or other documents of which it is aware.
- 3.45 AT THE TIME THE CITY APPROVES 90% CONSTRUCTION DOCUMENTS, THE DEVELOPER SHALL BE DEEMED TO HAVE WARRANTED TO OWNER, THAT THE CONSTRUCTION DOCUMENTS ARE CONSISTENT WITH EACH OTHER, AND CONSTRUCTIBLE FOR THE CONTRACT AMOUNT, SUBJECT TO ANY CHANGES REQUIRED BY BUILDING CODE AND THE PERMITTING PROCESS. FURTHER, THE DEVELOPER SHALL BE DEEMED TO HAVE WARRANTED TO OWNER THAT THE

WORK DESCRIBED IN THE CONSTRUCTION DOCUMENTS IS CONSTRUCTABLE WITHIN THE CONTRACT TIME, AND AT THE GMP.

3.46 Long Lead Procurement. Developer shall review the Project design for the purpose of identifying long lead procurement items (machinery, equipment, materials and supplies) and consult with Owner's Project Manager concerning same. When each item is identified, Developer shall notify the subcontractors and Owner of the required procurement and schedule. Such information shall be included in the bid documents and made a part of all affected subcontracts. Developer shall keep itself informed of the progress of the respective subcontractors or suppliers, manufacturing or fabricating such items, and advise Owner of any problems or possible delays in delivery.

### 3.47 Interfacing.

- 3.47.1 Developer shall take such measures as are appropriate to provide that all construction requirements will be covered in the separate procurement of long lead items, the separate construction subcontractors and the general conditions items without duplication or overlap, and sequenced to maintain completion of all Work on schedule. Particular attention shall be given to provide that each bid package clearly identifies the Work included in that particular separate subcontract, its schedule for start and completion and its relationship to the other separate subcontractors.
- 3.47.2 Developer shall include in the reports required under Paragraph 3.45 above, comments on overlap with any other separate subcontracts, omissions, lack of correlation between drawings, and any other deficiencies noted, in order that Developer may arrange for necessary corrections.

### 3.48 <u>Truth-In-Negotiation and Adjustment</u>.

The Developer shall execute a Truth-in-Negotiation Certificate (in the form of **Exhibit A** to the Agreement) stating that wage rates and other factual unit costs supporting the compensation are accurate, complete, and current at the time of contracting.

### **ARTICLE 4: ADMINISTRATION OF THE AGREEMENT**

- 4.1 The Developer will provide administration of the Work and the Project Site.
- 4.2 Any needed communications by and with subcontractors or material suppliers shall be through the Developer, with the Developer then communicating with the Owner's Project Manager.
- 4.3 The Owner's Project Manager shall review the Developer's applications for payment. The Owner's Project Manager shall reject Work which it determines

does not conform to the Contract Documents. However, this authority does not in any respect serve to release or otherwise discharge the Developer's responsibility with regard to performing the Work in compliance with the Contract Documents.

- 4.4 All Claims are required to be made in writing by the parties asserting the Claim.
  - 4.4.1 Claims arising prior to final payment or the earlier termination of the Agreement shall be referred to the Owner's City Manager for action.
  - 4.4.2 Initial notice of Claims by Developer shall be made in writing to Owner's City Manager within fifteen (15) days after the incident giving rise to the Claim or within fifteen (15) days after the complaining party first knew of the condition giving rise to the Claim, whichever is later, or else Developer shall be deemed to have waived the Claim. Written supporting data shall be submitted to Owner's City Manager within fifteen (15) days after the Claim is made to Owner's City Manager, unless Owner's City Manager grants additional time in writing, or else Developer shall be deemed to have waived the Claim. Pending final resolution of any Claim, unless otherwise mutually agreed between the parties in writing, the Developer shall proceed with the performance of the Work, to the extent reasonably possible, and the Owner shall proceed to make uncontested payments in accordance with the Contract Documents.
  - 4.4.3 The making or acceptance of any progress payment or a final payment shall not in any respect constitute a waiver of any Claim by the Owner or the Developer, respectively.
- 4.5 If the Developer encounters conditions at the site which are sub-surface or otherwise concealed physical conditions which could not have been discovered by Developer in the exercise of reasonable due diligence prior to the expiration of the Inspection Period set forth in Section 2.1.3, then the Developer shall provide notice to the Project Manager of the conditions, no later than fifteen (15) days after the first observance of the condition. After investigation, the Owner's City Manager will determine if conditions meet the foregoing criteria, and what, if any, adjustment is needed to the Contract Sum or Contract Time at the Owner's sole discretion. Owner is under no obligation to adjust the Contract Sum or Contract Time based on the existence of such condition. Developer reserves all rights to object to the determination by the City Manager hereunder.
- 4.6 In the event the Developer wishes to make a Claim for an increase in the GMP, the Developer shall provide written notice prior to proceeding to execute the Work which is the subject of the Claim or in connection with its request for an extension of Contract Time. However, prior notice is not required for Claims involving an immediate emergency endangering health, safety, welfare or property, but shall be provided as soon as possible.

- 4.7 In the event the Developer wishes to make a Claim for an increase in the Contract Time, the Developer is required to provide written notice. The Developer shall describe the effect of the claimed delay on the Contract Time, and shall furnish the Owner with such supporting documents as the Owner may reasonably require. Extensions of Contract Time may only be granted by the City Commission, except as provided for in Section 1.3 of this Agreement. In the event adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data, establishing that the weather condition was abnormal (in accordance with the standard provided in Section 8.6 hereof), could not have been reasonably anticipated, and had an adverse effect on the scheduled Work. Owner shall respond within the time frames provided in Section 4.9 hereof.
- 4.8 In the event either party to this Agreement suffers injury or damage to person or property as a result of an act or omission of the other party, of any of the other party's employees or agents, written notice of such injury or damage shall be provided to the other party immediately, and in no event later than 21 days after the injury or damage. In the event there is to be a Claim for additional cost or time, it shall be provided as described in this Article.
- 4.9 The Owner shall review within five (5) days, Claims and may (1) defer any action with respect to all or any part of a Claim and request additional information from the Developer; (2) decline to render a decision for any reason which the Owner deems appropriate; or (3) render a decision on all or a part of the Claim within fourteen (14) days from the date of the Claim. The Owner shall notify the Developer in writing of the disposition of such Claim. If the Owner decides that the Work relating to such Claim should proceed regardless of the disposition of such Claim, the Owner shall issue to the Developer a written order to proceed. The Developer shall proceed as instructed, and all rights of both parties with respect to such Claim shall be deemed to have been reserved. Owner shall be required to evaluate any such Claims in a reasonable manner, taking into account industry standards in the community.
- 4.10 Either party may pursue any Claim against the other in the State Court having jurisdiction in Broward County, Florida, as provided in Article 15 of this Agreement, provided the party has first complied with the provisions of this Article 4 with respect to such Claim.
- 4.11 Developer shall prepare, maintain and submit to Owner's Project Manager, for its review and approval, the various logs, reports, and schedules. These logs, reports and schedules shall not constitute nor take the place of any notice required to be given by Developer to Owner pursuant to the Contract Documents.
- 4.12 Developer shall maintain in a safe place at the Project site one record copy and one permit set of the Contract Documents, including, but not limited to, all Drawings, Specifications, addenda, amendments, Change Orders, Construction
Change Directive and Field Orders, as well as all written interpretations and clarifications issued by Developer, in good order and annotated to show all changes made during design and construction. The record Contract Documents shall be continuously updated by Developer throughout the prosecution of the Work to accurately reflect all field changes that are made to adapt the Work to field conditions, changes resulting from Change Orders, Construction Change Directive and Field Orders, and all concealed and buried installations of piping, conduit and utility services. Developer shall certify the accuracy of the updated record Contract Documents. As a condition precedent to Owner's obligation to pay Developer, Developer shall provide evidence, reasonably satisfactory to Owner, that Developer is fulfilling its obligation to continuously update the record Contract Documents. All buried and concealed items, both inside and outside the Project site, shall be accurately located on the record Contract Documents as to depth and in relationship to not less than two (2) permanent features (e.g. interior or exterior wall faces). The record Contract Documents shall be clean and all changes, corrections and dimensions shall be given in a neat and legible manner in red. The record Contract Documents, together with all approved samples and a counterpart of all approved shop drawings shall be available to Owner for reference. Upon completion of the Work and as a condition precedent to Developer's entitlement to final payment for the Work, the record Contract Documents, samples and shop drawings shall be delivered to Owner by Developer.

4.13 Developer shall advise Owner's Project Manager of requested or required participation in any meeting or inspection giving at least one week written notice unless such notice is made impossible by conditions beyond Developer's fault and control, in which case at least 48 hours prior written notice must be given.

## ARTICLE 5: SUBCONTRACTORS AND MATERIAL SUPPLIERS

5.1 The Developer shall continue to develop subcontractor interest in the Project and shall furnish to the Owner, through its City Manager, a list of possible subcontractors, including material suppliers who are to furnish materials or equipment fabricated to a special design, from which proposals will be requested for each portion of the Work. The Owner, through its City Manager, will reply in writing to the Developer, within five (5) days of receipt, if the Owner knows of any objection to such subcontractor or material supplier. The receipt of such list shall not require the Owner to investigate the qualifications of proposed subcontractors or material suppliers, nor shall it waive the right of the Owner later to reasonably and timely object to or reject any proposed subcontractor or supplier.

To the extent not prohibited by law, the Developer agrees to employ local subcontractors to the maximum extent feasible and warrants that the use of local subcontractors shall not increase costs to the City. The term local subcontractors is defined as "any business that has established and agrees to maintain a permanent place of business located in a non-residential zone and staffed with full-time employees within the limits of the City of Fort Lauderdale or within 10 miles of the corporate limits of the City of Fort Lauderdale or shall maintain a staffing level for their work of at least fifty percent (50%) of whom are residents of the City of Fort Lauderdale." The Developer agrees to provide the appropriate documentation necessary for the City to confirm that each subcontractor selected meets the criteria as defined herein.

The use of any subcontractor who does not meet the above-referenced definition of local subcontractor for any portion of the work must be requested in writing by Developer and is subject to approval by the City Manager.

- 5.2 <u>Equal Employment Opportunity and Affirmative Action</u>. The Developer shall comply with applicable laws, regulations, and special requirements of the Contract Documents regarding equal employment opportunity and affirmative action programs.
- 5.3 The Developer shall not subcontract with any proposed person or entity to which the Owner, through its City Manager, has made a reasonable and timely objection. If the Owner has a reasonable objection to a person or entity proposed by the Developer, the Developer shall promptly propose another person or entity for that portion of the Work. Further, the Developer shall not change a subcontractor, person or entity previously selected, in the event the Owner makes a reasonable objection to such change.
- 5.4 The Developer is required to enter into written agreements with each subcontractor who will perform any portion of the Work on the Project. The subcontract agreement shall incorporate the terms of the Contract Documents, and the terms of this Agreement. Further, where appropriate, the Developer shall require each subcontractor to enter into similar agreements with sub-subcontractors and material suppliers. Before entering any agreement with a subcontractor, Developer shall confirm that the subcontractor is properly licensed by the state and Broward County, and any applicable municipality, for the portion of the Work to be performed on the Project and shall supply such information or proof of licensing, in writing, to Owner.
- 5.5 Each subcontract agreement for a portion of the Work shall be assignable by the Developer to the Owner in the event of a termination of this Agreement by the Owner with cause or by Developer, with cause, and only for those subcontract agreements which the Owner accepts by notifying the subcontractor in writing.

#### 5.6 <u>Contract Requirements</u>.

5.6.1 All Subcontracts shall provide:

5.6.1.1 Limitation of Remedy – No damages for delay. The subcontractor's exclusive remedy for delay in the performance of the contract caused by events beyond its control, including delays claimed to be caused by the Owner or the Architect or attributable to the Owner or Architect and including claims based on breach of contract or negligence, shall be an extension of Contract Time.

5.6.1.2

In the event of a change in the Work, the subcontractor's claim for adjustments in the contract sum are limited exclusively to its actual costs for such changes plus no more than 10% for overhead, profit and bond costs for additional work and 10% for deducts. A detailed breakdown, with necessary backup, shall be provided.

5.6.1.3

5.6.1.4

The subcontract shall require the subcontractor to expressly agree that the foregoing constitutes its sole and exclusive remedies for delays and changes in the work and thus eliminate any other remedies or claims for increase in the contract price, damages, losses or additional compensation.

Developer shall include Owner as a third-party beneficiary with respect to all subcontracts between the Developer, Designer, Additional Designers, subconsultants and subcontractors. This third-party beneficiary status shall apply, but not be limited to, all insurance policies carried by the Developer, Designer, Additional Designers, subconsultants and subcontractors.

## ARTICLE 6: CONSTRUCTION BY OWNER

- 6.1 The Owner has the right to perform construction work related to the Project that is not covered by the Work, with the Owner's own employees, or by contracting with other individuals or entities. The Owner shall be responsible for coordination of activities of Owner's own employees or of any separate contractors, with the Work performed by the Developer and shall attempt to minimize any interference with Developer's performance of the Work. The Developer shall cooperate or participate with any separate contractor and the Owner in reviewing and coordinating construction schedules.
- 6.2 The Developer shall not unreasonably interfere with either the Owner or any separate contractor's ability to store materials and equipment, or perform construction Work, to the extent possible. In the event the Developer's Work depends upon, or connects to, the construction by the Owner or any separate contractor, the Developer is required, prior to proceeding with that portion of the Work, to report to the Owner any and all discrepancies or defects which would

Exhibit 1 CAM 15-1467 Page 39 of 158 render it unsuitable for continuation of the Work. In the event the Developer fails to report within five (5) Business Days of observation of any defect, such failure shall be deemed an acceptance of the Work performed by the Owner or separate contractor by the Developer.

6.3 Any and all costs caused by delays or by improperly timed or coordinated activities, or defective construction Work, shall be borne by the party responsible. The Developer shall promptly repair any damage caused by the Developer to any Work or to the property of the Owner or any separate contractor and Owner shall repair, or reimburse Developer for any repairs, relating to any damage caused by any of them to any portion of the Work or the property of Developer or any of its subcontractors.

### ARTICLE 7: CHANGES TO THE WORK

- 7.1 Changes to the Work may be accomplished by preparation of and execution of a Change Order or Construction Change Directive. Change orders or Construction Change Directives will be preceded by a change proposal request initiated by the Developer or the Owner. The Developer shall provide the Owner prices and details within ten (10) days of receipt of a change proposal request.
- 7.2 A Construction Change Directive shall be used in the event the parties to this Agreement cannot reach an agreement on the terms of a Change Order.
  - 7.2.1 Upon request of the Owner, the Developer shall, without any increase in the GMP for such estimate, submit to the Owner within ten (10) days, in such form as the Owner may require, an accurate written estimate of the cost of any proposed Work set forth in the Construction Change Directive.
  - 7.2.2 The estimate shall indicate the quantity and unit cost of each item of materials, and the number of hours of Work and hourly rate for each class of labor; as well as the description and amounts of all other cost chargeable under the terms of this subparagraph. Unit labor costs for the installation of each item of materials shall be shown if required by the Owner. The Developer shall promptly revise and resubmit such estimate if the Owner determines that it is not in compliance with the requirement of this Article, or that it contains errors of fact or mathematical errors. If required by the Owner, in order to establish the exact cost of new Work added or of previously required Work omitted, the Developer shall obtain and furnish to the Owner bona fide proposals from recognized suppliers for furnishing any material included in such Work. Such estimates shall be furnished promptly so as to occasion no delay in the Work, and shall be furnished at the Developer's expense. The Developer shall state in the estimate any extension of time required for the completion of the Work if the change or extra Work is ordered. Notwithstanding any other provision

of this Agreement, and without limitation, the Developer shall obtain and maintain supporting documentation for Change Orders that reports detailed and specific amounts for labor, materials and equipment charges, as well as mark-up.

- 7.2.3 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods, as selected by the Owner.
  - (a) By unit prices stated in the Contract Documents or otherwise mutually agreed upon.
  - (b) By cost and percentages estimated by the Developer as defined in this subparagraph and accepted by the Owner. The Developer's estimate shall become a fixed price which shall not be changed by any variation in the actual cost of executing the Work covered by the change.
  - (c) By actual cost determined after the Work covered by the change is completed, plus percentage.

As used in this subparagraph, "cost" shall mean the estimated or actual net increase or decrease in cost to the Developer, subcontractor, or subsubcontractor for performing the Work covered by the change, including actual payments for materials, equipment rentals, expendable items, wages and associated benefits to workmen and to supervisors employed full time at the site, insurance bonds and other provable direct costs, but not including any administrative, accounting, or other indirect or overhead costs, or any wages or benefits of supervisory personnel not assigned full time to the site, or any amount of profit or fee to the Developer, subcontractor or sub-subcontractor.

As used in this subparagraph, "percentage" shall mean an allowance to be added to or subtracted from the cost in lieu of overhead and profit and of any other expense which is not included in the cost of the Work covered by the change, as defined above. Percentage for the Developer, a subcontractor, a sub-subcontractor or any other lower tier subcontractors for any work performed by their own forces shall be a maximum of 14% of any net increase, and 14% of any net decrease when the decrease results in a reduction of overhead requirements. The maximum percentage of 14% as applied to subcontractors shall be a cumulative percentage, inclusive of sub-contractor, sub-subcontractors and other subordinate contracting parties. In the event the Developer does not perform any of the Work with its own forces, the percentage for the Developer shall be 12%.

- 7.3 When in the reasonable judgment of the Owner a series of Construction Change Directives effect a single change, percentage shall be calculated on the cumulative net increase or decrease in cost, if any.
- 7.4 If there are no applicable unit prices in the Contract Documents, it shall be the Owner's option to require the cost of any change to be determined by one of the other methods stated in 7.2.3. If the cost of the change to the Work is by unit prices and the nature of the Work is such that it cannot readily be measured after the completion of such Work or any subsequent Work, the Developer shall keep daily records, available at all times to the Owner for inspection, of the actual quantities of such Work put in place, and delivery receipts or other adequate evidence, indicating the quantities of materials delivered to the Project Site for use in such unit price Work, and distinguishing such from other similar material delivered for use in work included in the GMP.
- 7.5 If Owner elects to determine the Costs of the Work as provided in paragraph 7.2.3 (b) or (c), or if the method of determining the costs has not been established before the Work is begun, the Developer shall keep detailed daily records of labor and materials costs applicable to the Work.
- 7.6 Upon receipt of a Construction Change Directive, the Developer shall promptly proceed with the change in the Work. In the event the Owner and Developer do not agree with the adjustment in the Contract Time and/or adjustment to GMP as a result of the Construction Change Directive, the dispute shall be referred to the Owner for determination. Once the Owner and Developer reach agreement concerning the adjustment in the Contract Time and/or adjustment to GMP, such agreement shall be reduced to writing and executed in an appropriate change order.
- 7.7 Intentionally Omitted.
- 7.8 Notwithstanding any other provision in this Agreement concerning audits of this Project, and not in derogation thereof, Owner shall have the right to conduct an audit of Developer's books and records to verify the accuracy of Developer's Claim with respect to Developer's costs associated with any Change Order or Construction Change Directive, subject to the provisions of Section 3.14.
- 7.9 Owner's Project Manager may direct Developer to make nonmaterial changes to the Work, so long as such changes do not require or result in any adjustment to the Contract Amount/GMP, Contract Time or Project quality and are within the scope of the Work. All such changes must be evidenced by a written order from Owner to Developer. Developer shall comply with all such orders following proper execution of same.
- 7.10 No action, conduct, omission or course of conduct by Owner shall act to waive, alter, or change the requirement that Change Orders must be in writing and signed by Owner's Project Manager. Such written and signed Change Orders are

the sole and exclusive way to change either the amount of compensation to be paid to Developer or the time within which Developer is to perform its obligations hereunder. No changes will be allowed based upon actual, constructive or oral notice.

## ARTICLE 8: TIME

8.1 Time is of the essence in the performance of the Work under this Contract. The Developer shall commence Work, including Design Services, within ten (10) days after issuance of the Notice to Proceed by the Owner. The Developer shall furnish the Owner with a design and construction schedule which shall be considered a binding part of this Agreement. The completion of the design phase (indicated by receipt of the complete building permit) shall occur no later than two hundred ninety five (295) Business Days following the issuance of the Notice to Proceed by the Owner, subject to extension on a day-for-day basis for any delay in the permitting process beyond that shown in the attached Project Schedule. The Substantial Completion of the construction phase shall occur no later than three hundred thirty one (331) Business Days following the issuance of the written authorization by the Owner to commence construction, subject to adjustment pursuant to the terms hereof.

8.2 Substantial Completion of the Work shall be achieved when the Work has been completed to the point where Owner can occupy or utilize the Work for its intended purpose, with only minor Punch List items which can be corrected or completed without material interference with the Owner's use of the Project remaining to be completed or corrected. The entire Work shall be fully completed and ready for final acceptance by Owner within sixty (60) days after the Substantial Completion Date, subject to extension for any Punch List items requiring long lead time materials.

- 8.3 If the Owner has determined that the Developer should be permitted to extend the time for completion, the dates in the Project Schedule shall be adjusted accordingly to retain their same relationship to the adjusted date of Substantial Completion, and the dollar value of Work to be completed as of the first of each month shall be adjusted prorata.
- 8.4 Time is of the essence in the performance of the Work under the Contract Documents, subject to the provisions of the Contract Documents. The Owner and Developer agree that the losses suffered by Owner, if Substantial Completion of the Work is not achieved, are not ascertainable at this time. Developer acknowledges and agrees that, since time is of the essence, the Owner will suffer financial and other losses if substantial completion of the Work is not achieved within the Contract Time, as said Contract Time may be adjusted pursuant to the terms of the Contract Documents. Should the Developer fail to achieve Substantial Completion of the Work within the Contract Time, subject to adjustments as provided in the Contract Documents, Owner shall be entitled to assess, as liquidated damages, but not as a penalty, and as Owner's sole and

exclusive remedy for such failure, the sum of Two Thousand Dollars (\$2,000.00) for each day thereafter until Substantial Completion is achieved. The Developer shall have sixty (60) days to complete the Punch List items identified at the time of Substantial Completion. In the event Final Completion (which is the completion of all Punch List items) is not achieved within sixty (60) days after Substantial Completion (subject to extension in the event any repair/correction of a Punch List item involves long lead time items), the Developer shall be assessed One Thousand Dollars (\$1,000.00) for each day beyond such sixty (60) day period that Final Completion is delayed. Developer hereby expressly waives and relinguishes any right which it may have to seek to characterize the above noted liquidated damages as a penalty, which the parties agree represents a fair and reasonable estimate of the Owner's actual damages at the time of contracting if Developer fails to achieve Substantial Completion of the Work within the Contract Time. Further, the parties acknowledge that it would be extremely difficult, if not impossible, to ascertain Owner's actual damages with any degree of certainty in the event Developer fails to achieve Substantial Completion of the Work within the Contract Time. For the purposes of this Agreement, "Final Completion" shall mean that all Work, including Punch List items, has been completed by Developer.

- 8.5 Developer shall diligently pursue the completion of the Work and coordinate the Work being done on the Project by its subconsultants, subcontractors and materialmen, as well as coordinating its Work with all work of others at the Project site, so that its Work or the work of others shall not be delayed or impaired by any act or omission by Developer or anyone for whom Developer is liable. Developer shall be solely responsible for all construction means, methods, techniques, sequences, and procedures, as well as coordination of all portions of the Work under the Contract Documents, and the coordination of Owner's suppliers and contractors.
- 8.6 If Developer is obstructed or delayed in the prosecution of or completion of the Work as a result of unforeseeable causes beyond the control of Developer, and not due to its fault or neglect, including but not restricted to those force majeure items listed in Section 15.23 hereof, natural disasters, acts of God or of the public enemy, acts of government, fires, floods, epidemics, terrorist activities, material shortages or embargos or blockages, quarantine regulation, labor unrest, war (declared or undeclared), strikes, lockouts, existence of injunctions, moratoriums, or requirements for obtaining licenses, easements, permits or other compliance with applicable laws, rules and regulations, unusually severe weather conditions by comparison with the ten-year Broward County, Florida, average not capable of reasonable anticipation (collectively, "Force Majeure"), Developer shall notify Owner in writing within ten (10) days after the commencement of such delay, stating the cause or causes thereof, or be deemed to have waived any right which Developer may have had to request a time extension and/or extended General Conditions. As used herein, the term "acts of God" shall include a tropical storm or hurricane making landfall within fifty (50) miles from the Project Site or any

weather event that the Emergency Management Response enacts an evacuation of the coastal areas.

8.7 If Developer encounters on the Project site any materials reasonably believed by Developer to be petroleum or petroleum related products, Hazardous Materials, Hazardous Substance or Hazardous Waste which have not been rendered harmless, Developer immediately shall (i) stop Work in the area affected and (ii) report the condition to Owner in writing. If the Work is so stopped and Hazardous Materials, Substances or Waste is found, the Work in the affected area shall not thereafter be resumed except by Change Order. If an adjustment to the Contract Time is warranted a request for a Construction Change Directive shall be initiated. If no Hazardous Material, Substance or Waste is found after the Work is stopped, no Change Order is required to resume the Work in the affected area. Further, if the Hazardous Material, Substance or Waste was generated or caused by Developer or any of its employees, agents, subconsultants, subcontractors, or material suppliers, no Construction Change Directive will be required for an adjustment in the Contract Time and Developer shall indemnify Owner and hold Owner harmless for any costs incurred by Owner with respect to such hazardous material.

8.8 No Extended Damages: For other and additional good and valuable consideration the receipt and sufficiency of which is hereby acknowledged, except as otherwise permitted in the Contract Documents, and subject to availability of funds in the Contingency and Savings, the Developer covenants and agrees that in the event of any delay of construction or for any reason, allegation or claim, and notwithstanding the reason of the delay, reason, claim or allegation or who caused them or the construction delay or whether they were caused by the Owner, that there will be no entitlement to Developer to or for any direct or indirect financial damages or losses for extended corporate overhead impact, extended project overhead impacts, project support services. mobilization or demobilization or by whatever other label or legal concept or theory and types of names or labels or basis such claims may have, or any business damages or losses of whatever type or nature, and Developer hereby walves any right to make any such claim or claims. This provision will have application and effect when construction delays are anticipated and agreed upon by both the Owner and the Developer and in all other circumstances as well.

Delays caused by or resulting from Persons who are not affiliated with the Developer shall not give rise to a claim by Developer for damages for increase in material and/or labor costs, subject to availability of funds in the Contingency and Savings. Such Persons include, but are not limited to, the Owner's contractors and subcontractors, Florida Power and Light Company, AT&T, and any other utility at the project site.

8.9

### ARTICLE 9: PROGRESS PAYMENTS AND COMPLETION OF CONSTRUCTION

- 9.1 <u>Progress Payments</u>. Based upon applications for payment submitted to the Owner by the Developer, the Owner shall make progress payments on account of the Contract Sum to the Developer as provided below and elsewhere in the Contract Documents.
  - 9.1.1 Developer's monthly Applications for Payment shall be in such form and contain such detail and backup as Owner reasonably may require. Prior to submitting its first monthly Application for Payment, Developer shall submit to Owner's Project Manager, for review and approval, a Schedule of Values based upon the lump sum compensation to be paid Developer for Design Services hereunder. After its approval by Owner, that Schedule of Values shall be used as the basis for Developer's monthly Applications for Payment with respect to the Design Services. The first Application for Payment shall be submitted no earlier than thirty (30) days after the Notice to Proceed. The approved Schedule of Values shall be updated to reflect current Change Orders and Construction Change Directives and submitted each month to Owner along with a completed and notarized copy of the Application for Payment form. The Application for Payment shall be signed by the Developer, and notarized, and shall be supported by sufficient data which serves to establish the Developer's right to the payment, such as requisitions from subcontractors or material suppliers, and reflecting retainage of 10% which may be withheld until substantial completion or final completion. The amount of retainage will be reduced to 5% upon 50% completion of the Agreement value. The Design Fee retainage will be reduced to 0% upon acceptance by the City of 100% Construction Documents and permits being issued. The period covered by each Application for Payment shall be one calendar month ending on the last day of the month. The format and number of copies of such applications for payment shall be in a format as requested by the Owner's Project Manager. Retainage withheld may be reduced upon Substantial Completion of the Work to 150% of the estimated cost of all outstanding punch list items, upon approval by the Owner.
  - 9.1.2 The Developer shall submit to Owner's Project Manager, for its review, a Schedule of Values based upon the GMP listing the major elements of the Work and the dollar value for each element. That Schedule of Values shall be used as the basis for Developer's monthly Applications for Payment. The revised Schedule of Values shall be updated for the current month Change Orders and Construction Change Directives and submitted each month to Owner's Project Manager by Developer along with a completed and notarized copy of the Application for Payment form.

- 9.1.3 The Owner agrees to make payments in accordance with the Florida Local Government Prompt Payment Act for Construction Services, Section 218.735, Florida Statutes.
- 9.1.4 The Application for Payment shall reflect the amount of Work completed each month separated by materials stored and labor, inclusive of Developer's Fee, as a percent complete of each line item within the schedule of values for the Project. The Developer shall, upon request from the Owner, provide all required invoices, payrolls, petty cash accounts and any other evidence required by the Owner to verify the values indicated as percent complete in the Application for Payment.
- 9.1.5 Applications for payment may also include requests for payment for changes in the work which have been authorized by Change Order or Construction Change Directives, only when such Construction Change Directives have documented an adjustment to the Contract Sum. Further, applications for payment shall not include any requests for payment of amounts the Developer does not intend to pay to a subcontractor or material supplier, for any reason.
- 9.1.6 If payment is requested on the basis of materials and equipment not incorporated into the Project, but delivered and suitably stored at the Project Site, the Application for Payment shall also be accompanied by a bill of sale, invoice or other documentation warranting that Owner has received the materials and equipment free and clear of all liens, charges, security interests and encumbrances, together with evidence that the materials and equipment are covered by appropriate property insurance and other arrangements to protect Owner's interest therein, all of which shall be subject to Owner's satisfaction and Project Manager's ability to confirm same on inspection. Payments shall not be made for materials which are not suitably stored at the Site, or off-site at a bonded warehouse.
- 9.1.7 By submitting an Application for Payment, the Developer warrants that full and complete title will vest in the Owner with regard to all Work covered by an Application for Payment, no later than the time of such payment. Additionally, all Work represented by a submitted Application for Payment shall be free and clear of any and all liens, claims, security interests or encumbrances in favor of any Person.
- 9.1.8 Upon receipt of the Owner's Certificate for Payment, the Owner shall make payment to the Developer. The Developer shall, within seven (7) days of receipt of payment from Owner, pay each subcontractor out of the amounts paid to the Developer on account of such subcontractor's portion of the Work, minus any percentages retained as retainage from payments made to the Developer on account of such subcontractor's

portion of the Work. The Developer shall, by appropriate written agreement with each subcontractor, require each subcontractor to make payments to sub-subcontractors and material suppliers in a similar manner. However, the Owner shall not have an obligation to pay or to see the payment of money to a subcontractor, sub-subcontractor or material supplier.

- 9.1.9 Any certificate for payment, progress or monthly payment to Developer, or partial or entire use or occupancy of the Project by the Owner, shall in no way imply approval or acceptance of Developer's work.
- 9.1.10 Each Application for Payment shall be accompanied by a Release and Affidavit, in the form attached to the Agreement as **Exhibit H**, showing that all materials, services, labor, equipment and other bills associated with that portion of the Work payment is being requested on have been paid in full through the previous month's Application for Payment. Owner shall not be required to make payment until and unless the Release and Affidavit is furnished by Developer.
- 9.1.11 All Applications for Payment are subject to Owner's review and approval. Owner shall have the right to refuse to approve for payment any amounts, or portions thereof, requested by Developer in an Application for Payment, or rescind any amount previously approved for payment, and Owner may withhold any payments otherwise due Developer under this Contract or any other agreement between Owner and Developer, to the extent it is reasonably necessary, to protect Owner from any expense, cost or loss attributable to: (a) defective or deficient Work not properly remedied in accordance with the terms of the Contract Documents (provided, however, that the amount being withheld is only the amount necessary to address the defective Work); (b) the filing or reasonable evidence indicating the probable filing of third party claims against Owner attributable to the fault or neglect of Developer; (c) Developer's failure to make timely and proper payments to all subconsultants, subcontractors and suppliers; (d) Developer's failure to satisfactorily prosecute the Work in accordance with the requirements of the Contract Documents; or (e) any other material breach of the requirements of the Contract Documents by Developer. Owner shall have the right, but not the obligation, to take any corrective action Owner deems appropriate to cure any of the above noted items, at Developer's expense, if such items are not cured by Developer to Owner's reasonable satisfaction within five (5) Business Days after Developer's receipt of written notice from Owner; (provided, however, if such correction cannot be completed within five (5) Business Days, then if Developer commences to cure same within five (5) Business Days and diligently pursues same to completion, then Developer shall have satisfied this Agreement). The Owner's accountants, auditors or other representatives will review and report in writing on the Developer's

final accounting within 30 days after delivery of the final accounting to the Owner by the Developer. Based upon such Cost of the Work as the Owner's accountants report to be substantiated by the Developer's final accounting, and provided the other conditions of final payment have been met, the Owner will, within seven (7) days after receipt of the written report of the Owner's accountants, either issue to the Owner a final certificate for payment with a copy to the Developer, or notify the Developer and Owner in writing of the Owner's reason for withholding a certificate.

- 9.1.12 Upon Final Completion of the Project, Developer shall prepare and furnish to the Owner for the review and approval of the Owner, a final cost accounting of the Work. All Savings on the Project shall be divided as follows: seventy-five percent (75%) of the Savings to Owner; twenty-five percent (25%) of the savings to Developer. For the purposes of this Agreement, "Savings" shall mean, the GMP, less all Costs of Work to complete the Project (including all costs for Design Services), the Design Fee and the Developer's Fee. The parties stipulate and agree that this Contract is not a line-item GMP; consequently, the total Cost of Work used in the calculation of Savings to be distributed between the parties, shall be the sum total of all amounts paid or payable on all line items together and not by individual losses or savings on each line item. Developer's Final Application for Payment.
- 9.1.13 The Owner shall pay any amount within the given Application for Payment, including the final application that is not disputed by Owner in the time set forth within.
- 9.2 When the Developer deems the Work or a portion thereof to be Substantially Complete, the Developer shall notify the Owner. The Developer and Owner shall schedule a complete project walk-through and review. At the end of the walkthrough and review, the Owner and Developer shall prepare a single Punch List which shall include a comprehensive list of items to be completed or corrected. The Developer shall be responsible for completion and correction of all items on the Punch List. Items on the Punch List shall be corrected by the Developer within sixty (60) days of establishment of the Punch List; provided, however, that additional time is not necessitated by long lead time items. The creation of the Punch List does not in any respect alter the ultimate responsibility of the Developer to complete the project in accordance with the Contract Documents.
- 9.3 When the Owner determines that the Work or designated portion thereof is Substantially Complete, the Owner shall issue a Certification of Substantial Completion which establishes the date of Substantial Completion, provided, however, that where all Certificates of Occupancy and Certificates of Completion for the Work on the Project have been issued by the Building Official, then

Owner must issue a Certificate of Substantial Completion. After issuance of the Certificate of Substantial Completion, when the Owner determines that the Work on the Punch List has been completed in accordance with the Contract Documents, then Owner shall issue and Developer shall counter-sign a Certificate of Completion for the Project.

- 9.4 Intentionally Omitted.
- 9.5 The Owner may, at its discretion, upon issuance of a Certificate of Occupancy or a Certificate of Completion for a portion of the Work, occupy or use such portion of the Work, prior to the final completion of the Project. Prior to such Owner partial occupancy or use, the Owner and Developer shall inspect the area to be occupied in order to document the condition of the Work. However, in no event, shall the Owner's partial occupancy or use of the Work constitute acceptance of any Work not in compliance with the requirements of the Contract Documents.
- 9.6 Final payment of the Contract Sum will be made after the Owner certifies that the Work is complete, and Owner's representatives complete their final acceptance report and issuance of the Certificate of Final Completion, which shall all occur no later than five (5) days after Developer's completion of the Punch List. Neither final payment nor any remaining retainage shall be paid to the Developer until the Owner has received an affidavit in a form sufficient to the Owner that all indebtedness in connection with the performance of the Work for which the Owner or the Owner's Project Site may be held liable or encumbered. have been fully paid or otherwise satisfied; a certification in a form acceptable to the Owner which establishes that all required insurance will remain in full force and effect after final payment and will not be cancelled or allowed to expire until at least 30 days prior written notice has been provided to the Owner; consent of the surety to final payment; and any other certifications reasonably required by the Owner establishing full payment or satisfaction of any obligations. In the event the Developer fails to furnish such certifications as the Owner reasonably requires to satisfy the Owner that there are no outstanding Claims by any Persons whomsoever, the Owner may require the Developer as a condition of final payment and at the Developer's expense, to furnish a bond in a form and amount satisfactory to the Owner to indemnify the Owner against such Claims. Final payment by the Owner to the Developer shall not constitute a waiver of any Claim the Owner may have against the Developer for Work not in compliance with the Construction Documents. However, acceptance of final payment by the Developer, a subcontractor or material supplier shall constitute a waiver of any Claims by that entity or individual, except those previously made in writing and clearly identified as unsettled at the time of the final Application for Payment and payment thereon. Neither the acceptance of the Work nor payment by Owner shall be deemed to be a waiver of Owner's right to enforce any obligations of Developer hereunder or to the recovery of damages for defective Work not discovered by Owner at the time of final inspection.

#### 9.7 Intentionally Omitted.

9.8 Notwithstanding anything contained herein to the contrary, to the extent Owner fails or refuses to make payments hereunder, Owner acknowledges and agrees that Developer's obligation to pay its subcontractors and/or suppliers under this Agreement is expressly conditioned upon and limited to the payments made by the Owner to the Contractor.

#### ARTICLE 10: DEVELOPER'S SAFETY PROGRAM

- 10.1 The Developer shall be responsible for initiating, maintaining and supervising a safety program in connection with its Work under the Agreement and Construction Documents. A copy of the Developer's Safety Program shall be submitted to Owner upon written request of City Manager or his designee.
- 10.2 In the event the Developer encounters on the site material reasonably believed to be a Hazardous Material, Substance or Waste which has not been rendered harmless, the Developer shall immediately stop Work in the area affected and report the condition to the Owner in writing. Developer shall thereafter as soon as reasonably possible conduct a thorough investigation to determine if the suspected Hazardous Material, Substance or Waste in the affected area is in fact a Hazardous Material, Substance or Waste and shall certify to Owner that such material is not a Hazardous Material. Substance of Waste or if such material is in fact a Hazardous Material, Substance or Waste that has been abated and that it is safe to return to the affected area and resume Work. Owner may require Developer to furnish copies of reports of tests conducted by a qualified testing laboratory verifying the absence of such Hazardous Material, Substance or Waste before Developer will resume Work. By way of a Construction Change Directive the Contract Time may be equitably adjusted to account for the time lost due to the encountering of the Hazardous Material, Substance or Waste and the reasonable cost associated therewith, pursuant to the procedure for making a claim set forth in Article 4.
- 10.3 The Developer shall not be required pursuant to the changes clause herein to perform without consent any Work relating to Hazardous Material, Substance or Waste.
- 10.4 The Developer shall take all reasonable precautions for the safety of and shall provide all reasonable protection to prevent damage, injury or loss to employees and other persons on the Project Site and Staging Areas, the Work and all materials and equipment to be incorporated into the Work, other property utilized in conjunction with the Work on the Project, including public rights-of-way or properties adjacent thereto, and any other property of the Owner, whether or not forming part of the Work located at the Project Site or adjacent thereto and areas to which the Developer has access.

- 10.5 The Developer shall erect and maintain all reasonable safeguards for safety and protection, including signs and other warnings as appropriate, in its performance of the Agreement. In the event the Work requires the use or storage of explosives or other hazardous materials, equipment, or means or methods, the Developer shall exercise the utmost care and carry on such activities under the continuous supervision of properly qualified individuals.
- 10.6 The Developer shall promptly remedy any and all damage and loss to property referred to above. In the event the damage or loss is due in whole to the Developer's negligence, the Developer shall bear the entire cost of the loss or damage, if the damage or loss is due in part to the Developer's negligence, the Developer shall bear the cost that is determined, by a neutral third party, mutually selected by the parties.
- 10.7 The Developer shall designate its superintendent on the Project as its safety program representative.
- 10.8 The Developer shall provide and maintain in good, operating condition suitable and adequate fire protection equipment and services, and shall comply with all reasonable recommendations regarding fire protection made by the representatives of the fire insurance company carrying insurance on the Work or by the local fire chief or fire marshal. The area within the site limits shall be kept orderly and clean, and all combustible rubbish shall be promptly removed from the site.
- 10.9 The Developer shall at all times protect excavations, trenches, buildings and materials, from rain water, ground water, backup or leakage of sewers, drains and other piping, and from water of any other origin and shall remove promptly any accumulation of water. The Developer shall provide and operate all pumps, piping and other equipment necessary to this end.
- 10.10 During the progress of the Work and at all times prior to the date of Substantial Completion, the Developer shall provide temporary heat, ventilation, and enclosure, adequate to permit the Work to proceed in a timely fashion, and to prevent damage to completed Work or Work in progress, or to materials stored at the Project Site or other location utilized for Staging of Materials or Equipment. The permanent heating and ventilation systems may be used for these purposes when available unless otherwise provided in the Contract Documents.
- 10.11 In the event of an emergency affecting the safety of Persons or property, the Developer shall utilize its judgment and discretion to prevent or minimize any threatened damage, injury or loss. In the event of such an emergency, the Developer shall immediately notify the Owner, and coordinate and cooperate in the resolution of all such emergencies.

# ARTICLE 11: INSURANCE AND BOND REQUIREMENTS

- 11.1 The Developer shall deliver the proofs of insurance to the Owner prior to the commencement of any Work, and in no event any later than five (5) days after the execution of this Agreement; with the exception of the required bonds and Builder's Risk Insurance, which will be delivered within ten (10) days of the earlier of issuance of all permits or start of the construction phase.
- 11.2 The Developer shall, throughout the performance of its services and Work under this Agreement and throughout the term of this Agreement maintain and provide to the Owner the insurance coverages listed in this Article 11. The insurance policies shall be issued and underwritten by a licensed insurer, licensed as such in the State of Florida. The Developer shall provide insurance that may not be reduced, terminated, or cancelled unless 30 days prior written notice thereof is furnished to the Owner. Certificates of insurance and certified, true copies of all policies (if required by the Owner) shall be furnished to the Owner within five (5) days after the execution of this Agreement. In the event of any cancellation or reduction in insurance coverage, the Developer shall obtain substitute coverage, maintaining required coverage without any lapse of coverage whatsoever. The insurance policies shall name the Owner as additional insured (except for the professional liability and worker's compensation insurance).

Developer shall provide Additional Insured Endorsement ISO Form CG 20 10 11 85, ISO Form 20 37 10 01, or their equivalent acceptable to Owner, at no expense to Owner, its officers, directors or employees as additional insureds.

- 11.3 The insurance required by in this Article shall include all major divisions of coverage, and shall be on a commercial general basis including premises and operations (including explosion, collapse, and underground), Independent Contractor Hired Products and Completed Operations, and Owned, Nonowned, and Hired Motor Vehicles. Such insurance shall be written for not less than any limits of liability required by law or others set forth in the Contract Documents, whichever is greater. All insurance shall be written on an occurrence basis, unless the Owner approves in writing coverage on a claims-made basis. Coverages, whether written on an occurrence or claims-made basis, shall be maintained without interruption from date of commencement of the work until date of final payment and termination of any coverage required to be maintained after final payment.
- 11.4 The Developer shall furnish to the Owner copies of any endorsements that are subsequently issued amending limits of coverage.
- 11.5 The insurance required by this Article shall be written for not less than the following, or greater if required by law.

11.5.1 Workers' Compensation: The Developer shall provide and maintain workers' compensation insurance for all employees in the full amount required by statute and full compliance with the applicable laws of the State of Florida. Said policy must include Employers' Liability insurance with limits of no less than:

•	Each Accident	\$500,000.00
٠	Disease – Policy Limit	\$500,000.00
•	Disease – Each Employee	\$500,000.00

- 11.5.2 Commercial General Liability: The Developer shall provide minimum limits of \$5,000,000 each occurrence, \$5,000,000 annual aggregate combined single limit for bodily injury and property damage liability. This shall include premises/operations (including explosion, collapse, and underground), independent contractors, products, completed operations, broad form property damage, personal and advertising injury, and specifically confirming contractual liability, and insurina the indemnification and hold harmless clause of the contract. This policy of insurance shall be considered primary to and not contributing with any insurance maintained by Owner and shall name Owner as an additional insured. The policy of insurance shall be written on an "occurrence" form, Maximum deductible will be \$25,000 each claim.
- 11.5.3 Builder's Risk: Coverage must be provided as follows:
  - Form All Risk Coverage Coverage is to be no more restrictive than that afforded by the latest edition of Insurance Services Office Forms CP 0020.
  - Amount of insurance is to be 100% of the completed value of such additions(s), building(s) or structures(s).
  - Waiver of Occupancy Clause or Warranty Policy must be specifically endorsed to eliminate any "Occupancy Clause" or similar warranty or representation that the building(s), addition(s) or structure(s) in the course of construction shall not be occupied without specific endorsement of the policy. The policy must be endorsed to provide that the builder's risk coverage will continue to apply until final acceptance of the building(s), addition(s) or structure(s) by the Owner.
  - Maximum Deductible \$25,000 each claim.
  - Named Additional Insured The Owner must be included as a named additional insured.
  - Notice of Cancellation and/or Restriction The policy must be specifically endorsed to provide the City of Fort Lauderdale with a thirty (30) day notice of cancellation and/or restriction.
  - Wind and Hail Deductible shall be between 3% and 5%, not to exceed 5%.

- 11.5.4 **Commercial Automobile Liability**: The Developer shall provide minimum limits of liability of \$2,000,000 each accident, combined single limit for bodily injury and property damage. This shall include coverage for owned, hired, and non-owned automobiles. The State of Florida has no-fault automobile insurance requirements. The Developer shall be certain coverage is provided which conforms to any specific stipulation in the law.
- 11.5.5 Professional Liability: The Designers retained by the Developer shall, during the Term of this Agreement, provide the Owner with evidence of professional liability insurance for its legal liability arising out of the performance of professional services under this Agreement. Such insurance for the Architect of Record shall have minimum limits of \$2,000,000 per occurrence, \$2,000,000 aggregate with respect to acts, errors or omissions in connection with professional services to be provided under this Agreement. Additional Designers shall have minimum limits as described in **Exhibit I.** Designer shall waive its right of recovery against Owner as to any claims under this insurance. The Designer further agrees to maintain like coverage for a minimum of four (4) years following the latest date of the following: completion of this Agreement, completion of construction or issuance of a certificate of occupancy. Designer shall promptly submit certificates of insurance providing for an unqualified written notice to Owner of any cancellation of coverage or reduction in limits. In addition, Designer shall also notify Owner by certified mail, within twenty-four (24) hours after receipt, of any notices of expiration, cancellation, non-renewal or material change in coverages or limits received by Designer from its insurer. Designer shall promptly submit a certified, true copy of the policy and any endorsements issued or to be issued on the policy if requested by Owner.
- 11.5.6 **Umbrella Liability**: The Developer shall provide umbrella coverage with limits of no less than \$2,000,000 excess of Commercial General Liability, Automobile Liability, and Employer's Liability.
- 11.6 The Developer shall procure property insurance for that portion of the Work stored off Site or in transit, and the cost for such shall be borne by the Developer.
- 11.7 <u>Performance and Payment Bonds</u>. The Developer shall furnish bonds covering the faithful performance of the Agreement and payment of any and all obligations arising under the Agreement as required by Florida law. Upon request, the Developer shall furnish a copy to any person or entity requesting a copy. Such bonds shall be in conformance and compliance with Section 255.05, Florida Statutes, and shall contain the information and provisions set forth in the referenced section.

- 11.7.1 Developer shall provide Owner with Performance and Payment Bond, in the form prescribed in **Exhibits F and G**, in the amount of 100% of the total sum of the GMP, the costs of which are to be paid by Developer. The Performance and Payment Bonds must comply with the following provisions and must be otherwise acceptable to Owner:
  - 11.7.1.1 The Bonds must be underwritten by a surety company which has a currently valid Certificate of Authority issued by the State of Florida, Department of Insurance, authorizing it to write surety bonds in the State of Florida.
  - 11.7.1.2 The surety company shall have currently valid Certificate of Authority issued by the United States Department of Treasury under Sections 9304 to 9308 of Title 31 of the United States Code.
  - 11.7.1.3 The surety company shall be in full compliance with the provisions of the Florida Insurance Code.
  - 11.7.1.4 The surety company shall have at least twice the minimum surplus and capital required by the Florida Insurance Code at the time this Agreement is executed.
  - 11.7.1.5 The Bonds must be fully performable in Florida, with service and venue in Broward County, Florida.
- 11.7.2 If the surety for any bond furnished by Developer is declared bankrupt, becomes insolvent, its right to do business is terminated in the State of Florida, or it ceases to meet the requirements imposed by the Contract Documents, Developer shall, within five (5) days thereafter, substitute another bond and surety, both of which shall be subject to the minimum requirements noted above and Owner's approval. The cost of any such replacement bond shall be paid by Developer.
- 11.7.3 Within five (5) days prior to commencing any construction Work on the Project Site, Developer shall execute, deliver to the Owner and cause to be recorded in the Public Records of Broward County, Florida, a copy of the Performance and Payment Bonds for the Project. Developer, prior to commencing any construction Work on the Project Site, shall deliver within five (5) days to Owner evidence, reasonably acceptable to Owner, of the recording of said Bonds along with recording data. The delivery of such evidence is a condition precedent to Owner's obligation to make any progress payments to Developer hereunder. Developer shall deliver the original, recorded Performance and Payment Bonds to Owner within two (2) Business Days after receipt of same from the Broward County Clerk of the Circuit Court.

- 11.7.4 Notwithstanding anything to the contrary, Owner agrees that upon issuance of a final Certificate of Occupancy for the Project, the Surety shall be released and discharged on the Performance Bond provided to Owner hereunder solely as to any and all liability, claims, obligations, causes of action, costs and damages of any nature and/or kind, at law and/or in equity, relating to, or arising out of, any Design Services provided under this Agreement, but not as to any other obligations of the Surety arising under the Performance Bond. In such event, among other causes of action Owner may have, at law or in equity, as to costs, damages or services arising from any Design Services, Owner agrees to look to the professional liability insurance provided by the Designers under the terms of this Agreement for defects in Design Services and not to the Performance Bond.
- 11.8 <u>Insurance as Additional Remedy</u>. Compliance with the insurance requirements of this Agreement shall not limit the liability of the Developer, its subcontractors, sub-subcontractors, material suppliers, employees, or agents to the Owner or others. Any remedy provided to the Owner, or the Owner's officers, employees, agents or assigns, by the insurance shall be in addition to and not in lieu of any other remedy available under the Agreement or otherwise.
- 11.9 <u>No Waiver by Approval/Disapproval</u>. Neither approval by the Owner nor failure to disapprove the insurance furnished by the Developer shall relieve the Developer of its full responsibility to provide the insurance as required by this Agreement.
- 11.10 During the term of this Agreement Developer shall provide, pay for, and maintain, with companies satisfactory to Owner, the types of insurance described in this Article 11. All insurance shall be from responsible companies duly authorized to do business in the State of Florida. All insurance carriers shall be rated (A) or better by the most recently published A.M. Best Rating Guide. The Owner reserves the right to accept or reject the insurance carrier. Certified, true and exact copies of all insurance policies required shall be provided to Owner, on a timely basis, if requested by Owner. Certificates of insurance and policies shall contain provisions that thirty (30) days' written notice by registered or certified mail shall be given Owner of any cancellation, intent not to renew, or reduction in the policies' coverages. Developer shall also notify Owner, in a like manner, within ten (10) days after receipt, of any notices of expiration, cancellation, nonrenewal or material change in coverages or limits received by Developer from its insurer, and nothing contained herein shall relieve Developer of this requirement to provide notice. All insurance coverages of Developer shall be primary to any insurance or self-insurance program carried by Owner applicable to this Agreement.
- 11.11 The terms "The City of Fort Lauderdale, Florida" and "City" shall include The City of Fort Lauderdale, Florida, a municipal corporation, and all Authorities, Boards,

Bureaus, Commissions, Divisions, Departments and offices thereof and individual members and employees thereof in their official capacity, and/or while acting on behalf of The City of Fort Lauderdale, Florida.

- 11.12 All insurance policies, other than the Professional Liability policy and the Workers Compensation policy, provided by Developer to meet the requirements of this Agreement shall name The City of Fort Lauderdale, Florida, as that name is defined in section 11.11 above, as an additional insured as to the operations of Developer under the Contract Documents and shall contain a severability of interests provisions.
- 11.13 Companies issuing the insurance policy or policies shall have no recourse against Owner for payment of premiums or assessments for any deductibles which all are at the sole responsibility and risk of Developer.
- 11.14 For all policies of insurance required in this Article 11, deductibles shall be no greater than \$5,000 each claim, unless otherwise specified. Any self-insured retention or deductible greater than \$5,000 will need to be pre-approved by the Owner and may be subject to review of appropriate financial documents of Developer.
- 11.15 All insurance coverages of Developer shall be primary to any insurance or selfinsurance program carried by Owner applicable to this Agreement, and the "Other Insurance" provisions of any policies obtained by Developer shall not apply to any insurance or self-insurance program carried by Owner applicable to this Agreement.
- 11.16 The Certificates of Insurance, which are to be provided pursuant to paragraph 11.2 above, must identify this Agreement.
- 11.17 All insurance policies shall be fully performable in Broward County, Florida, and shall be construed in accordance with the laws of the State of Florida.
- 11.18 All insurance policies to be provided by Developer pursuant to the terms hereof will accept service of process in Broward County, Florida and that the exclusive venue for any action concerning any matter under those policies shall be in the appropriate state court situated in Broward County, Florida.
- 11.19 The acceptance by Owner of any Certificate of Insurance pursuant to the terms of this Agreement evidencing the insurance coverages and limits required hereunder does not constitute approval or agreement by Owner that the insurance requirements have been met or that the insurance policies shown on the Certificates of Insurance are in compliance with the requirements of this Agreement.
- 11.20 Before starting and through completion of all Work required hereunder and thereafter as required to cover claims arising from the period through final

acceptance and the warranty period, Developer and their Designers shall procure and maintain insurance of the types and to the limits specified in this Article 11. Developer shall require each of its Additional Designers to procure and maintain, until the completion of that Additional Designer's work or services, insurance of the types and to the limits specified in **Exhibit 1.** Prior to work being performed by any Additional Designers, the Developer shall provide to Owner copies of all Certificates of Insurance and certified, true copies of underlying insurance policies, of the types and limits specified above, that the Developer has required from all Additional Designers.

- 11.21 If any insurance provided pursuant to this Agreement expires prior to the completion of the Work required hereunder, renewal Certificates of Insurance and, if requested by Owner, certified, true copies of the renewal policies, shall be furnished to Owner thirty (30) days prior to the date of expiration.
- 11.22 Should at any time Developer not maintain the insurance coverages required in this Agreement, Owner may, following five (5) Business Days' written notice to Developer and Developer's failure to cure same within such period of time, cancel this Agreement or at its sole discretion shall be authorized to purchase such coverages and charge Developer for such coverages purchased. If Developer fails to reimburse Owner for such costs within thirty (30) days after demand, Owner has the right to offset these costs from any amount due Developer under this Agreement. Owner shall be under no obligation to purchase such insurance, nor shall it be responsible for the coverages purchased or the insurance company/companies used. The decision of Owner to purchase such insurance coverages shall in no way be construed to be a waiver of its rights under this Agreement.

#### ARTICLE 12: UNCOVERING AND CORRECTION OF THE WORK

- 12.1 In the event the Developer covers a portion of the Work contrary to the request of the Owner or prior to an inspection by the Owner, the Developer must if requested in writing uncover the Work, and then shall be required to replace the work at the Developer's sole expense, without any change to the Contract Time.
- 12.2 In the event the Developer covers a portion of the Work which the Owner has not specifically requested to inspect, the Owner may request the Developer to uncover the Work. If such Work is in compliance with the Contract Documents, the costs of uncovering the Work and replacement of the Work shall be reimbursed to the Developer by the Owner. If such Work is not in compliance with the Contract Documents, the Developer shall bear such costs.
- 12.3 The Project is subject to and shall be constructed in accordance with the Florida Building Code, and to all applicable codes referenced therein. This facility will be designed and constructed in accordance with Section 255.2575, Florida Statutes on energy-efficient and sustainable buildings.

- 12.3.1 The Developer shall promptly correct any and all Work rejected by the Owner and any and all Work which fails to comply with the requirements of the Contract Documents. The Developer shall bear all of the costs for correcting such Work.
- 12.4 If within one (1) year after the date of issuance of the Certificate of Final Completion, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Developer shall promptly return to the Project and correct the deficient Work upon receipt of written notification from the Owner to do so. The obligation set forth in this subparagraph shall survive acceptance of the Work under this Agreement and any termination of the Agreement. The Owner shall give such written notice promptly after discovery of the deficient Work. This one (1) year Developer warranty described in this subparagraph does not impact or impair any manufacturer's warranty or the Owner's ability to make any other claim against the Developer as allowed under Florida law.
- 12.5 The Developer shall, within five (5) Business Days after notice from Owner, remove from the Project Site any and all portions of the Work which are not in compliance with the requirements of the Contract Documents (provided, however, that in the event such removal or correction cannot be completed within five (5) Business Days, then if Developer commences such work within five (5) Business Days and diligently pursues same to completion, then Developer shall have satisfied this Agreement). In the event the Developer fails to correct any non-conforming Work within the time period provided above, the Owner may correct such non-conforming Work in accordance with this Article 12.6. In the event the Owner corrects the non-conforming Work, the Owner may remove and store any salvageable materials or equipment at the Developer's expense. If the Developer fails to reimburse the Owner for such expenses, within seven (7) days after written notice, the Owner may take any and all action it deems appropriate in order to obtain reimbursement of its expenses under this subparagraph. Any action taken by the Owner under this subparagraph shall not in any respect serve to limit in law or equity the Owner's ability to place any other Claim against the Developer.
- 12.6 The Owner has the discretion to accept Work that is not in compliance with the requirements of the Contract Documents. In this event, the Owner shall reduce its decision to writing, which shall include any reduction, if any, to the Contract Sum as a result of the acceptance of non-conforming Work. Any such adjustment to the Contract Sum shall apply whether or not final payment has been made under this Agreement.

## ARTICLE 13: TERMINATION OF THE AGREEMENT

#### Termination by Developer

- 13.1 The Developer may terminate the Agreement if the Work is stopped for a period of 30 days, for any of the following reasons:
  - (a) The issuance of an order of a Court or any other public authority having jurisdiction;
  - (b) An act of government, such as a declaration of national emergency, making materials unavailable;
  - (c) In the event the Owner has not made payment on an uncontested certificate for payment within the time stated in the Contract Documents and after written notice from the Developer of the failure to make timely payment and a period of twenty (20) days after notice within which Owner may make payment and prevent termination. However, if Owner in response to this written notice initiates a legal action for declaratory judgment or other appropriate action then the funds due, if any, shall be paid within thirty-five (35) days of the rendition of a final judgment or if an appeal is taken then promptly upon completion of all appellate and trial court proceedings.
- 13.2 In the event the Developer elects to terminate the Agreement for one of the above enumerated reasons, the Developer shall provide the Owner with fourteen (14) days prior written notice, and thereafter terminate the Agreement and receive from the Owner payment for any non-deficient Work executed up to the date of such termination notice; including the Design Fee and Developer's Fee through such date of termination.
- 13.3 In the event the Work is stopped for a period of sixty (60) days through no act or fault of the Developer, a subcontractor, sub-subcontractor or material supplier as set forth in § 13.1 above, but solely because the Owner has persistently failed to fulfill the Owner's obligations under the Contract Documents with respect to substantive matters relating to the progress of the Work, the Developer may, upon giving the Owner a 14 days prior written notice and opportunity to cure, terminate the Agreement and recover from the Owner payment for any non-deficient Work executed up to the date of said termination notice, including the Design Fee and Developer's Fee through such date of termination.

# **Termination by Owner for Cause**

- 13.4 The Owner may terminate the Agreement for cause for any of the following reasons:
  - (a) Repeated refusal or failure of the Developer to supply sufficient, properly skilled workers, or proper materials to the Work Site;

- (b) Failure to make payment to subcontractors for materials or labor in accordance with the required agreements between the Developer and subcontractors; provided, however, that Developer does not have a legitimate dispute with such subcontractor or supplier, and has so notified Owner of same;
- (c) Disregard by the Developer of any applicable laws, ordinance, or rules, regulations or orders of a public authority having jurisdiction; or
- (d) Any substantive and material breach of a provision of this Agreement or the Contract Documents.
- If any of the above enumerated causes exist, the Owner may, without prejudice to 13.5 any other rights or remedies, and after giving the Developer and the Developer's surety, if any, twenty-one (21) days' advance written notice, terminate this Agreement with the Developer and may, subject only to any prior rights of the surety, take possession of the Project Site and of all materials, equipment, tools, and construction equipment and machinery thereon owned by the Developer; accept assignment of subcontracts pursuant to this Agreement; and finish the Work by whatever reasonable method the Owner may deem appropriate. The Developer shall not be entitled to receive any further payment under this In the event the costs for completing the Work, including Agreement. compensation for any Owner services and reasonable expenses, exceed the unpaid balance of the Contract Sum, the Developer shall pay the difference to the Owner. In the event the Agreement is terminated, the Owner shall pay the Developer for any non-deficient Work executed up to the date of such termination, including, without limitation, de-mobilization costs, General Conditions, together with the Design Fee and Developer's Fee up to the date of termination.

### Suspension or Termination by Owner Without Cause

13.6 The City Manager may, without cause, require the Developer to suspend or delay the Work in whole or in part for such period of time as the City Manager may determine necessary. Such requirement shall be in writing, authorized and signed by Owner's City Manager which will also constitute a change in Contract Time equal to the number of days Work is suspended by the Owner without cause. Termination of this Agreement can only be authorized by the City Commission. An adjustment will be made for any increases in the cost of performance of the Agreement, caused by the suspension, delay or termination, including, without limitation, de-mobilization and re-mobilization costs, extended General Conditions and escalation (or credit for decrease in costs) on materials, together with the Design Fee and Developer's Fee up to the date of suspension or termination. No adjustment shall be made to the extent that the Developer is responsible for any cause which would also have suspended, delayed, or interrupted the work or to the extent that the Developer has previously requested an equitable adjustment under another provision of this Agreement and such request has either been granted or denied. This provision shall not be construed to modify or expand the limitations on the Developer's remedies or compensation provided in this Agreement.

13.7 In the event the Work is temporarily suspended or terminated under this subparagraph, the Owner shall pay the Developer for any non-deficient Work executed up to the date of said temporary suspension or termination, including, without limitation, de-mobilization and re-mobilization costs, extended General Conditions and escalation (or credit for decrease in costs) on materials, together with the Design Fee and Developer's Fee up to the date of suspension or termination.

### ARTICLE 14: TAX EXEMPT OWNER DIRECT MATERIAL/EQUIPMENT PURCHASE PROGRAM

14.1 The Owner shall appoint the Developer as the Owner's authorized representative with respect to any matter arising out of the purchase orders under this program. The Developer will cooperate fully with the Owner with respect to the implementation of a tax exempt direct material/equipment purchase program involving the direct purchase of various construction materials, supplies and equipment that is currently part of this Contract.

The Owner Direct Purchase Program, is attached hereto as **Exhibit C**, controls the Direct Purchase Program for the Project. The Direct Purchase Program will be operated in accordance with the following provisions:

- 14.1.1 The Owner will issue its own purchase orders directly to the third party vendor or supplier of material and equipment purchased under the Direct Purchase Program. The purchase order will be accompanied by a Certificate of Entitlement in accordance with the Florida Department of Revenue Rule 12A-1.094, F.A.C. Public Works Contracts, and the Owner's Exemption Certificate which includes its name, address, and the exemption number with issuance and expiration date.
- 14.1.2 All material and equipment purchased under the Direct Purchase Program is sold directly to the Owner and is directly to the Owner and is directly invoiced by the vendor or supplier.
- 14.1.3 The Owner takes title and possession of all materials and equipment purchased under the Direct Purchase Program from the vendor or seller before they are incorporated into the Project.

- 14.1.4 The Owner makes direct payment to the third party vendor or seller for all purchases from its own funds or accounts for all purchases under the Direct Purchase Program.
- 14.2 The Owner agrees to process its purchase orders so that the progress of construction is not jeopardized. Should the Owner fail to process the purchase orders within a time frame so as not to delay the construction, the Developer shall, at its sole discretion, void the Owner purchase order and purchase the item direct thereby waiving any rights the Owner may have for a direct purchase tax savings and Developer shall be entitled to an extension of the Contract Time to address any impact on the critical path. If the items included in the purchase order represent any materials, supplies or equipment that is part of a subcontractor's scope of Work, then any terms and conditions that the subcontractor deems to be warranted to protect their interest, shall also be included and/or substituted. Vendors and suppliers must be approved by the Owner prior to the processing of purchase orders.
- 14.3 The items being purchased shall be purchased from the vendors and suppliers selected by the Developer and/or the subcontractor for prices negotiated by the Developer and/or subcontractors.
- 14.4 The Developer has established an accounting system that will adequately track and monitor the direct purchases made by the Owner. The system developed by the Developer shall track and monitor that materials purchased (and shall adequately identify the same), costs, tax savings, and such other charts of accounts or information as may be reasonable requested by the Owner. The Developer shall submit a monthly accounting report of this information with the Developer's Application for Payment.
- The Developer shall provide all rough drafts of purchase orders to the Owner for 14.5 processing in such time and sequence that the Work will not be impeded or delayed in any manner. Notwithstanding anything in this Article 14.5 to the contrary, the Developer remains fully responsible under its Contract with the Owner, and the implementation of this direct purchase program shall not be used in any manner by the Developer to justify any delay unless such delay is a direct result of the Owner's failure to comply timely with the terms of the Direct Owner Purchase Program through no fault of the Developer, whereupon Developer shall be entitled to an extension of the Contract Time with respect to any such delays. Should a delay be incurred that is not the result of the Owner's failure, as stated above, the Developer shall be held accountable for such a delay. The Developer, for \$10.00 and other valuable consideration, the adequacy of receipt of which is hereby acknowledged and deemed to be sufficient, does hereby release, waive and hold harmless the Owner from and against any claim for damages, acceleration damages, or any other matter, claim or damage that may arise from or be related to in any way the owner's Direct Purchase Program to the extent stated herein.

- 14.6 The Developer shall be responsible for all purchases in the same manner as if the Developer had purchased the items, inclusive of managing the warranties for the Owner. The Developer shall cooperate with the Owner and take all action necessary to assure that all warranties with respect to any materials or equipment which may be available from any vendor are passed-through to the Owner.
- 14.7 Intentionally Omitted.
- 14.8 The Developer and its surety hereby agree that the performance bond penal amount shall be unaffected by any direct purchase deductive change order which is made pursuant to this program.
- 14.9 The Developer agrees that its builder's risk insurance coverage amount shall be unaffected by any direct purchase deductive change order implemented pursuant to this program.
- 14.10 Payment shall be directly made by the Owner to the vendor for any Direct Purchases.
- 14.11 To the extent authorized under Florida law, Owner agrees to indemnify and hold harmless the Developer, its subcontractors and suppliers of and from any claims, liability, or responsibility to the State of Florida for any action the State may take against any of them for the payment of any sales or use taxes as a result of Owner's direct purchase of such materials, supplies or equipment.
- 14.12 Intentionally Omitted.
- 14.13 Owner shall not withhold retainage on any payments made to the vendor.

## ARTICLE 15: MISCELLANEOUS PROVISIONS

- 15.1 <u>Recitals</u>. The recitals set forth in the Whereas Clauses are incorporated herein by reference and made a part of this Agreement.
- 15.2 <u>Defining Terms</u>. Unless otherwise noted, the terms used in the Agreement shall have their ordinary and customary meanings as used in the industry.
- 15.3 <u>Disclaimer of Liability</u>. The Owner shall not at any time, be liable for injury or damage occurring to any person or property from any cause, whatsoever, arising out of Developer's fulfillment of this Agreement.
  - 15.3.1 <u>Indemnification</u>. Subject to the provisions of Section 725.06(2), Florida Statutes, and for other and additional good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Developer agrees as follows:

- 15.3.2 The Developer shall, at its sole cost and expense, indemnify and hold harmless the Owner, including, but not limited to, its officers, agents, contractors and subcontractors, representatives, employees, volunteers and elected and appointed officials, successors and assigns from or on account of all claims, damages, losses, liabilities and expenses, direct or indirect including, but not limited to, fees and charges of engineers, architects, attorneys, experts, consultants and other professionals and court costs arising out of or from (a) any negligence, recklessness or intentional, wrongful misconduct of the Developer, including, but not limited to, its agents, officers, servants, representatives and employees as well as its subcontractors and their agents, officers, servants representatives and employees (hereafter the Developer); (b) any and all bodily injury, sickness, disease or death caused by any negligent recklessness or intentional wrongful conduct on the part of the Developer or as a result of Developer's failure to act; (c) injury to or destruction of property (other than the Work), including any resulting loss of use; (d) other such damages, liabilities, or losses received or sustained by any person or persons during or on account of any of Developer's negligent actions connected with the construction of this Project including the warranty period; (e) the violation of any federal, state, county or other laws, ordinances or regulations by Developer, its subcontractors, agents, servants, independent contractors or employees; (f) any construction defect including both patent and latent defects; ; (g) the breach or alleged breach by Developer of any term of the Agreement, including the breach or alleged breach of any warranty or guarantee; (h) the use of any improper materials.
- 15.3.3 Developer agrees to indemnify, defend, save and hold the Owner harmless from any type of Claim or demand whatsoever (including, but not limited to, damages, liabilities, losses, claims, fines, costs, expenses and fees, and from any and all suits and causes of actions of every name, or description) that may be brought against Owner, on account of any claims, fees, royalties, or costs for any invention or patent and/or for the infringement of any and all copyrights or patent rights claimed by any person, Developer, or corporation pertaining to the acts or omissions of Developer, its agents, servants, independent contractors, subcontractors or employees under this Agreement;
- 15.3.4 Except as otherwise provided herein, Developer shall pay all Claims, losses, liens, settlements or judgments of any nature in connection with the foregoing indemnifications including, but not limited to, reasonable attorney's fees and suit costs for trials and appeals.
- 15.3.5 If any Subcontractor, supplier, laborer, or materialmen of Developer or any other person directly or indirectly acting for or through Developer files or

attempts to file a mechanic's or construction lien against the real property on which the Work is performed or any part thereof or against any personal property or improvements thereon or make a claim against any monies due or to become due from the Owner to Developer or from Developer to a Subcontractor, for or on account of any Work, labor, services, material, equipment, or other items furnished in connection with the Work or any change order, Developer agrees to satisfy, remove, or discharge such lien or claim at its own expense by transfer to bond, payment, or otherwise within fifteen (15) days from receipt of written notice from the Owner.

Additionally, until such time as such lien or claim is satisfied, removed or discharged by Developer, 150% of the amount of the lien shall be held by, Owner as security for the satisfaction, removal and discharge of such lien and any expense that may be incurred while obtaining the discharge. If Developer shall fail to do so, Owner shall have the right, in addition to all other rights and remedies provided by this Agreement or by law, to satisfy, remove, or discharge such lien or claim by whatever means Owner chooses at the entire and sole cost and expense of Developer which costs and expenses shall, without limitation, include attorney's fees, litigation costs, fees and expenses and all court costs and assessments, and which shall be deducted from any amount owing to Developer. In the event the amount due Developer is less than the amount required to satisfy Developer's obligation under this, or any other article, paragraph or section of this Agreement, the Developer shall be liable for the deficiency due the Owner.

The Developer and the Owner agree that Section 725.06(2), Florida Statutes controls the extent and limits of the indemnification and hold harmless provisions of this Agreement, if any, and that the parties waive any defects in the wording of this Article that runs afoul of said statutory section.

15.4 <u>Limitation of Liability</u>. The Owner desires to enter into this Agreement only if in so doing the Owner can place a limit on the Owner's liability for any cause of action arising out of this Agreement, so that the Owner's liability for any breach never exceeds the Developer's Fee minus any amount paid to the Developer. For other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Developer expresses its willingness to enter into this Agreement with the knowledge that the Developer's recovery from the Owner to any action or claim arising from the Agreement is limited to the GMP minus any amount paid to the Developer, including the Developer's Fee. Accordingly, and notwithstanding any other term or condition of this Agreement that may suggest otherwise, the Developer agrees that the Owner shall not be liable to the Developer for damages in an amount in excess of the Developer's Fee minus any amount paid to the Developer for any action or claim arising out of this

Agreement. Nothing contained in this paragraph or elsewhere in this Agreement is in any manner intended either to be a waiver of the limitation placed upon the Owner's liability as set forth in Section 768.28, Florida Statutes, or to extend the Owner's liability beyond the limits established in said Section 768.28.

15.5

If the Developer commits a default due to its insolvency or bankruptcy, the following shall apply:

15.5.1 Should this Agreement be entered into and fully executed by the parties, and funds have been released to the Developer by the Owner (Debtor) files for bankruptcy, the following shall occur:

> In the event the Developer files a voluntary petition under 11 U.S.C. 301 or 302, or an order for relief is entered under 11 U.S.C. 303, the Developer shall acknowledge the extent, validity, and priority of the lien recorded in favor of the Owner. The Developer further agrees that in the event of this default, the Owner shall, at its option, be entitled to seek relief from the automatic stay provisions in effect pursuant to 11 U.S.C. 362. The Owner shall be entitled to relief from the automatic stay pursuant to 11 U.S.C. 362(d)(1) or (d)(2), and the Developer agrees to waive the notice provisions in effect pursuant to 11 U.S.C. 362 and any applicable Local Rules of the United States Bankruptcy Court. The Developer acknowledges that such waiver is done knowingly and voluntarily.

15.5.1.2

15.5.1.1

Alternatively, in the event the Owner does not seek stay relief, or if stay relief is denied, the Owner shall be entitled to monthly adequate protection payments within the meaning of 11 U.S.C. 361. The monthly adequate protection payments shall each be in an amount determined in accordance with the Note and Mortgage executed by the Developer in favor of the Owner.

15.5.1.3

In the event the Developer files for bankruptcy under Chapter 13 of Title 11, United States Code, in addition to the foregoing provisions, the Developer agrees to cure any amounts in arrears over a period not to exceed twenty-four (24) months from the date of the confirmation order, and such payments shall be made in addition to the regular monthly payments required by the Note and Mortgage. Additionally, the Developer shall agree that the Owner is over secured and, therefore, entitled to interest and attorney's fees pursuant to 11 U.S.C. 506(b). Such fees shall be allowed and payable as an administrative expense. Further, in the event the Developer has less than five (5) years of payments remaining on the Note, the Developer agrees that the treatment afforded to the claim of the Owner under any confirmed plan of reorganization shall provide that the remaining payments shall be satisfied in accordance with the Note, and that the remaining payments or claim shall not be extended or amortized over a longer period than the time remaining under the Note.

15.5.2 Should this Agreement be entered into and fully executed by the parties, and the funds have not been forwarded to Developer, the following shall occur:

15.5.2.1

In the event the Developer files a voluntary petition pursuant to 11 U.S.C. 301 or 302, or an order for relief is entered under 11 U.S.C. 303, the Developer acknowledges that the commencement of a bankruptcy proceeding constitutes an event of default under the terms of this Agreement. Further, the Developer acknowledges that this Agreement constitutes an executory contract within the meaning of 11 U.S.C. 365, The Developer acknowledges that this Agreement is not capable of being assumed pursuant to 11 U.S.C. 365(c)(2), unless the Owner expressly consents in writing to the assumption. In the event the Owner consents to the assumption, the Developer agrees to file a motion to assume this Agreement within ten (10) days after receipt of written consent from the Owner, regardless of whether the bankruptcy proceeding is pending under Chapter 7, 11, or 13 of Title 11 of the United States Code. The Developer further acknowledges that this Agreement is not capable of being assigned pursuant to 11 U.S.C. 365(b)(1).

- 15.6 Should the Developer's Work be terminated by the Owner, the termination shall not affect any rights of the Owner against the Developer then existing or which may thereafter accrue. Any retention or payment of monies due the Developer by the Owner will not release the Developer from liability.
- 15.7 The Developer has no right, authority or ability to terminate the Work except for wrongful withholding of any payments due the Developer from the Owner or for the reasons set forth in Section 13.1.
- 15.8 Unless the context of this Agreement otherwise clearly requires, references to the plural include the singular, references to the singular include the plural, the term "including" is not limiting, and the terms "hereof", "herein", "hereunder", and similar terms in the Contract Documents refer to the Contract Documents as a whole and not to any particular provision thereof, unless stated otherwise.

- 15.9 <u>Gender</u>. Unless the context clearly indicates to the contrary, pronouns having a neuter, masculine or feminine gender shall be deemed to include the others.
- 15.10 <u>Entire Agreement</u>. This Agreement and the Construction Documents incorporated herein by reference constitute the entire Agreement between the parties with respect to the matters covered by this Agreement. All prior negotiations, representations and agreements not incorporated in this Agreement are cancelled. This Agreement can be modified or amended only by a written document duly executed by the parties or their duly appointed representative.
- 15.11 <u>Binding Effect</u>. Each and all of the covenants, terms, provisions and agreements contained in this Agreement shall be binding upon and inure to the benefits of the parties and their respective assigns, successors, subsidiaries, affiliates, holding companies and legal representatives, as allowed in this Agreement.
- 15.12 <u>Notices</u>. All notices shall be in writing, and may be served by (a) depositing the same in the United States mail addressed to the party to be notified, postpaid, and registered or certified with return receipt requested, (b) delivering the same to such party, by (i) personal delivery, or (ii) overnight courier, or (c) facsimile transmission provided that a copy is sent on the same day, by 5 p.m. (local time of party giving notice), by any of the methods described in (a) or (b). Notice deposited in the mail shall be deemed to have been given on the third day next following the date postmarked on the envelope containing such notice, or when actually received, whichever is earlier. All notices to be given to the parties shall be sent to or delivered at the addresses or facsimile numbers set forth below:

If to the Owner:

Lee R. Feldman, City Manager City of Fort Lauderdale 100 North Andrews Avenue Fort Lauderdale, Florida 33301

With a Copy:

City Attorney City of Fort Lauderdale 100 North Andrews Avenue Fort Lauderdale, Florida 33301

If to Developer:

Joseph Cerrone, III, President Recreational Design & Construction, Inc.

### 3990 North Powerline Road Oakland Park, Florida 33309

With a Copy to:

Steven W. Deutsch, Esquire Frank, Weinberg & Black, P.L. 7805 S.W. 6<sup>th</sup> Court Plantation, Florida 33324

By giving the other party at least 15 days written notice, each party shall have the right to change its address and specify as its new address any other address in the United States of America.

- 15.13 <u>Waiver</u>. No consent or waiver, express or implied, by either party to this Agreement to or of any breach or default by another in the performance of any obligations shall be deemed or construed to be consent or waiver to or of any other breach or default by that party. Except as otherwise provided in this Agreement, failure on the part of any party to complain of any act or failure to act by another party or to declare the other party in default, irrespective of how long such failure continues, shall not constitute a waiver of the rights of that party.
- 15.14 <u>Captions</u>. The headings used for the various portions of this Agreement and the Construction Documents are inserted only as a matter of convenience, and for reference, and in no way define, limit or describe the scope or the intent of this Agreement, any section of this Agreement, or any section of the Contract Documents.
- 15.15 <u>Severability</u>. In the event the provisions of this Agreement are determined by a Court of competent jurisdiction to be illegal or unenforceable, then such unenforceable or unlawful provision shall be excised from this Agreement, and the remainder of this Agreement shall continue in full force and effect. Notwithstanding the foregoing, if the result of the deletion of such provision will materially and adversely affect the rights of a party, such party may elect, at its option, to terminate this Agreement in its entirety.
- 15.16 <u>Cumulative Remedies</u>. All rights, powers, remedies, benefits, and privileges are available under any provision of this Agreement to any party, is in addition to and cumulative of any and all rights, powers, remedies, benefits and privileges available to such party under all other provisions of this Agreement, at law or in equity.
- 15.17 <u>Approval</u>. Whenever any review or approval is required by any party, such party agrees that such review or approval will be promptly and expeditiously

prosecuted to conclusion, subject to any stricter time frames specifically set forth in the Contract Documents.

- 15.18 <u>Further Assurances</u>. The parties agree to execute any and all further instruments and documents, and take all such action as may be reasonably required by any party to effectuate the terms and provisions of this Agreement and the transactions contemplated in this Agreement.
- 15.19 <u>No Partnership or Joint Venture</u>. It is understood and agreed that nothing contained in this Agreement shall be deemed or construed as creating a partnership or joint venture between the parties or any third party, or cause any party to be responsible in any way for the debts and obligations of the other party.
- 15.20 <u>No Construction Against Drafter</u>. Each of the parties has been represented by legal counsel who have had ample opportunity to, and have, participated in the drafting of this Agreement. Therefore, this Agreement shall not be construed more favorably or unfavorably against any party.
- 15.21 <u>Third Party Beneficiary</u>. This Agreement has been made and entered into for the sole protection and benefit of the Owner, and its respective successors, and no other person or entity shall have any right or action under this Agreement.
- 15.22 <u>No Assignments</u>. This Agreement is for the personal services of the Developer, and may not be assigned by the Developer in any fashion, whether by operation of law or by conveyance of any type including without limitation, transfer of stock in the Developer, without the prior written consent of the Owner, which consent the Owner may withhold in its sole discretion.
- 15.23 Force Majeure. With regard to the performance under this Agreement, a party shall not be deemed to be in default of this Agreement, or have failed to comply with any term or conditions if, for reasons beyond the parties reasonable control, including without limitation acts of God, natural disaster, labor unrest, strikes, lockouts, terrorist activities, material shortages, embargos or blockages, war, declared or undeclared, the existence of injunctions or requirements for obtaining licenses, easements, permits or other compliance with applicable laws, rules and regulations, unusually severe weather conditions by comparison with the ten-year Broward County, Florida average, such performance is not reasonably possible within such time periods, then the time for such performance shall be extended until removal of such reasons beyond the parties reasonable control, provided that the party commences such performance as soon as reasonably possible and diligently pursues such performance. As used herein, the term "acts of God" shall include a tropical storm or hurricane, making landfall within fifty (50) miles from the Project Site or any weather event that the Emergency Management Response enacts an evacuation of the coastal areas.
- 15.24 <u>Governing Law and Venue</u>. This Agreement shall be governed and construed under the laws of the State of Florida. Broward County, Florida shall be the proper for all suits to enforce this Agreement. Any legal proceeding arising out of or in connection with this Agreement shall be brought in the Circuit Courts of Broward County, Florida. Developer (which term for the purposes of this subparagraph shall include Developer's surety) consents and submits to the exclusive jurisdiction of any such court and agrees to accept service of process from the State of Florida in any matter to be submitted to any such court. Notwithstanding any other provision of the Contract Documents, the Owner does not agree to, nor shall the parties arbitrate in any manner whatsoever any issue arising out of this Agreement, the Contract Documents or the performance thereof. In connection with a dispute, legal action or litigation arising out of this Agreement, the Contract Documents or the performance thereof, the prevailing party shall be entitled to receive its attorneys' fees and costs from the nonprevailing party.
- 15.25 <u>Waiver of Jury Trial</u>. THE PARTIES EXPRESSLY WAIVE THE RIGHT TO A JURY TRIAL.
- 15.26 <u>Dispute Resolution</u>. Prior to initiating any litigation arising out of the Agreement, the parties agree to submit the dispute to non-binding mediation by a mediator who is certified in Florida in an effort to resolve disputes in an expedient manner. Each party shall bear their own attorneys' fees, and the cost of the mediator shall be split between the parties.
- 15.27 <u>Right to Enter this Agreement</u>. Each party warrants and represents, with respect to itself, that neither the execution of this Agreement nor the performance of its obligations under this Agreement shall violate any legal requirement, result in or constitute a breach or default under any indenture, contract, or other commitment or restriction to which it is a party or by which it is bound. Each party also warrants and represents, with respect to itself, that the execution of this Agreement and the performances and obligations under this Agreement shall not require any consent; vote, or approval which has not been obtained, or at the appropriate time shall not have been given or obtained. Each party agrees that it has or will continue to have throughout the term of this Agreement the full right and authority to enter into this Agreement and to perform its obligation under this Agreement. Upon written request, each party agrees to supply the other party with evidence of its full right and authority.
- 15.28 <u>Conduct While on Owner Property</u>. The Developer acknowledges that its employees and agents must behave in an appropriate manner while on the premises of any Owner facility and shall at all times conduct themselves in a manner consistent with City of Fort Lauderdale policies and subject to the City Manager or designee. It will be considered a breach of this Agreement for any agent or employee of the Developer to behave in a manner which is inconsistent with good conduct or decorum, or to behave in any manner which will disrupt

the Owner's business or constitute any level of threat to safety, health, and wellbeing of any citizen, visitor, volunteer or employee of The City of Fort Lauderdale. The Developer agrees to immediately remove any agent or employee if directed to do so by the City Manager or designee.

- 15.29 <u>Owner Transfer of Interest</u>. If the Owner conveys its interest in the Project to a third party, any rights which the Owner may have against the Developer arising from this Agreement shall automatically transfer to such third party without the necessity of a written document or consent from the Developer.
- 15.30 <u>Public Entity Crime Information Statement and Debarment</u>. Section 287.133(2)(a), Florida Statutes states: "A person or affiliate who has been placed on the convicted vendor list following a conviction for a public entity crime may not submit a bid on a contract to provide any goods or services to a public entity, may not submit a bid on a contract with a public entity for the construction or repair of a public building or public work, may not submit bids on leases of real property to a public entity, may not be awarded or perform work as a contractor, supplier, subcontractor, or consultant under a contract with any public entity, and may not transact business with any public entity in excess of the threshold amount provided in Section 287.017, for CATEGORY TWO for a period of thirtysix (36) months from the date of being placed on the convicted vendor list."

By signing this Agreement, Developer certifies, to the best of its knowledge and belief, that it and its principals:

- (a) Are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from covered transactions by a federal department or agency.
- (b) Have not, within a five-year period preceding the issuance of RFP #105-10408 been convicted of or had a civil judgment rendered against them for commission of fraud or a criminal offense in connection with obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under public transaction; violation of federal or state antitrust statutes or commission of embezzlement, theft, forgery, bribery, falsification or destruction of records, making false statements or receiving stolen property.
- (c) Are not presently indicted or otherwise criminally charged by a governmental entity (federal, state or local) with commission of any of the offenses enumerated in the preceding paragraph (b).
- (d) Have not within a five-year period preceding the issuance of RFP #105-10408 had one or more public transactions (federal, state or local) terminated for cause or default.

Developer agrees to notify the City of Fort Lauderdale within 30 days after the occurrence of any of the events, actions, debarments, proposals, declarations, exclusions, convictions, judgments, indictments, information, or terminations as described in Article 15.30 (a) – (d) above, with respect to Developer or its principals.

- 15.31 <u>No Waiver of Sovereign Immunity</u>. Nothing herein is intended to serve as a waiver of sovereign immunity by any agency or political subdivision to which sovereign immunity may be applicable.
- 15.32 <u>Non-Discrimination</u>. The parties shall not discriminate against any employee or participant in the performance of the duties, responsibilities and obligations under this agreement because of race, color, religion, gender, age, marital status, disability, political or religious beliefs, national or ethnic origin.
- 15.33 <u>Waiver of Consequential Damages</u>. Notwithstanding anything contained herein to the contrary, Developer and Owner hereby waive all Claims against each other for consequential damages arising out of or relating to this Agreement or the other Contract Documents.
- 15.34 WAIVER OF <u>CHAPTER 558 NOTICE</u>. THE PARTIES AGREE THAT ANY CLAIMS FOR CONSTRUCTION DEFECTS SHALL NOT BE SUBJECT TO THE NOTICE AND CURE PROVISIONS OF CHAPTER 558, FLORIDA STATUTES.

#### [REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS OF THE FOREGOING, the parties have set their hands and seals the day and year first above written.

CITY

WITNESSES:

Print Name

(SEAL)

CITY OF FORT LAUDERDALE, a municipal corporation of the State of Florida:

Bν

CK" SEILER, Mayor Ρ

LEE R. FELDMAN City Manager

ATTEST By . JOND'A JOSEPH, City Clerk JEFFREY MODARCEUT, ASST CITY CLERCK

Approved as to form:

CARRIE L. SARVER Assistant City Attorney

L:\CLS\FLAC\Developer's Agreement - CLEAN FINAL Revised Agreement 9-24-12.doc

WITNESSES

Fr. s. T. Slores Print Name

Kavin HIPPolite Print Name

RECREATIONAL DESIGN & CONSTRUCTION, a Florida corporation;

В٧ rint Name/Title

ATTEST

Βv

Print Name/Title

(CORPORATE SEAL)

STATE OF FLORIDA: COUNTY OF BROWARD:

(SEAL)

Notary Public, State of Florida (Signature of Notary taking Acknowledgment)

Name of Notary Typed, Printed or Stamped

My Commission Expires:

Commission Number\_\_\_\_\_



#### Exhibit A

#### The City of Fort Lauderdale, Florida Truth-In-Negotiation Certificate

The wage rates and other factual unit costs supporting the compensation under the Developer's Agreement between the City of Fort Lauderdale, Florida and Recreational Design & Construction, Inc. dated <u>2</u>/<u>2</u>, 2012 are accurate, complete and current as of the time of entering into the contract. This Certificate is executed in Compliance with Section 287.055(5)(a), Florida Statutes.

DATED this  $\mathscr{A}$  day of 2012. hertion; 14C DEVELOPE B١ STATE OF COUNTY OF The foregoing instrument was acknowledged before me this d day of TOTYDITT ,2012, by (11)92 *X*)( as .Th( of KALIEUT 12600n · (ORGT Developer, on behalf of Developer. He/She is personally known to me OR has produced as identification.

My Commission Expires:

(AFFIX NOTARY SEAL)

Notary Public (Signature)

CYNTHIA GLUNT MY COMMISSION # EE 080736 EXPIRES: August 4, 2015 Bonded Thru Notary Public Underwriters

(Title or Rank)

(Printed Name)

(Serial Number, if any)

#### The City of Fort Lauderdale, Florida GMP

Pursuant to Paragraph 3.10 of the Developer's Agreement between the City of Fort Lauderdale, Florida and Recreational Design & Construction, Inc., for the Project known as Fort Lauderdale Aquatic Center Project, the Owner and Developer establish the Guaranteed Maximum Price and Contract Time for the Work as set forth herein below.

#### I. Guaranteed Maximum Price (GMP)

The Developer's Guaranteed Maximum Price for the Work (GMP), including the Cost of the Work and the Developer's Fee is (1) Twenty-Four Million Eight Hundred Sixty Four Thousand Nine Hundred and Fifty Dollars (\$24,864,950.00) for the public improvements (as delineated on the Project Program submitted to the City of Fort Lauderdale on March 28, 2012) and (2) Seven Million Five Hundred Seventy Two Thousand Four Hundred and Eighty Four Dollars (\$7,572,484.00) for the parking garage, both elements inclusive of the cost for Design Services, the Design Fee and Developer's Fee for a total GMP of Thirty Two Million, Four Hundred Thirty Seven Thousand, Four Hundred Thirty Four Dollars (\$32,437,434.00) The Developer's Fee is hereby established as Four Million, Seven Hundred Ten Thousand, Five Hundred Seventy Seven Dollars, Eighty Cents (\$4,710,577.80) based on a fixed percentage of 17.78% for each of the parking garage component and for the swimming pool component of the Project, calculated on all amounts reflected in the GMP Sum, excluding permits, payment and performance bond, builder's risk insurance, project specific E & O policy, Developer's Contingency and Owner's Contingency.

The Above GMP is for the **FULL AND FINAL GUARANTEED MAXIMUM PRICE FOR THE WORK**, subject to the terms set forth in the Agreement.

**II. Contract Time** – The date of Substantial Completion of the construction of the Work shall be on or before three hundred thirty one (331) Business Days from the issuance of written authorization by Owner to commence construction. The date of Final Completion for the Work established by this **Exhibit B** shall be sixty (60) days after Substantial Completion, by which the entire Work shall be fully completed and ready for acceptance.

#### III. Enumeration of GMP Exhibits/Attachments

Work shall be in conformance with the Contract Documents and the Contract. Exhibits include the following attachments, which further delineate and itemize pertinent elements of the GMP and the associated project Scope of Work. Said Exhibits are as follows Exhibits 1 through 11 are required at 90% Construction Document approval by the City and Final GMP **Exhibit B**. Exhibits 12 through 16 are required within 15 days after the issuance of all permits.

Exhibit 1	Final GMP Summary of Cost by Division/Trade Package
Exhibit 2	GMP General Condition Cost Itemization
Exhibit 3	GMP Cost of Work Exclusions, overhead staff and any lump sum costs (include descriptions of formulas used to allocate the cost to the project; as well as third party evidence for the rates or unit prices used in the allocation).
Exhibit 4	GMP Developer List of onsite staff charged to Cost of Work
Exhibit 5	GMP Assumptions and Clarifications
Exhibit 6	GMP Allowances (including unit prices and quantity amounts)
Exhibit 7	GMP Accepted Cost Savings
Exhibit 8	Schedule of List of Drawings, as Signed/Dated by A/E of Record and Developer
Exhibit 9	Schedule/List of RFI/ASI's (asked and answered during Bid Portion of the Work that is included in the Cost of the Work)
Exhibit 10	Developer Insurance Provided Affidavit, Bond & Insurance Rate and Cost (include descriptions of formulas used to allocate the cost to the project, as well as third party evidence for the rates used or unit prices in the allocation).
Exhibit 11	Developer Surety Form of Bond (indicating all language of Bond document)
Exhibit 12	Developer List of Sub-contractors and suppliers with License Nos., as dated
Exhibit 13	Developer Affidavit, Attesting to Subcontractor/Vendor Licensure Verification, as dated
Exhibit 14	Project Master Delivery Schedule (w/summary milestone delivery items)
Exhibit 15	Project Schedule (CPM Delivery schedule with itemized breakdown of Work scope by Building, and resource loaded)
Exhibit 16	Master Schedule (Project Submittals with submittal due dates/responsible Party)

#### IV. Signature

IN WITNESS THEREOF, the above parties have executed this instrument, the name of each party being affixed and these present duly signed by its undersigned representative, pursuant to authority of its governing body.

**OWNER:** City of fort Lauderdale, Florida By:

Print Name: Lee R. Feldman

Title: <u>City Manager</u> Date: <u>10-15-12</u>

ATTEST By: Name/Title:

RECREATIONAL DESIGN & CONSTRUCTION, INC. By: Print Name: HIFF C. CLIFOM Title: 1710/2017 Date: 9/27/12-

ATTEST: By: [ Name/Title:

r

.

ć

### Exhibit 1 - Final GMP Summary of Cost by Division/Trade Package

1

#### Exhibit 2 - GMP General Condition Cost Itemization

The Contract Documents should include a certified statement that the labor costs represent those amounts that are actually paid to the Persons that are to be working on the Project. Provide the labor burden for the each of the submitted staffing

#### Exhibit 3 - GMP Cost of Work Exclusions

OH staff and any lump sum costs (include descriptions of Formulas used to allocate the cost to the project as well as third party evidence for the rates or unit prices used in the allocation)

#### Exhibit 4 - GMP Developer

List of onsite staff charged to Cost of Work, provide the labor burden for the submitted staff.

· • · · ·

## Exhibit 5 – GMP Assumptions and Clarifications

Exhibit 1 CAM 15-1467 Page 87 of 158

Exhibit 6 – GMP Allowances, including unit prices and quantity amounts

Exhibit 1 CAM 15-1467 Page 88 of 158

Exhibit 7 – GMP Accepted Cost Savings

# Exhibit 8 – Schedule of List of Drawings, as Signed/Dated by A/E of Record and Developer

Exhibit 1 CAM 15-1467 Page 90 of 158

## Exhibit 9 – Schedule/List of RFI/ASI's asked and answered during Bid Portion of the Work that is included in the Cost of the Work

٢

. (

 $\frac{1}{2} \left\{ \frac{1}{2} \left$ 

.

Exhibit 1 CAM 15-1467 Page 91 of 158

#### Exhibit 10 – Developer Insurance Provided Affidavit, Bond & Insurance Rate and Cost

Include descriptions of formulas used to allocate the cost to the Project, as well as third party evidence for the rates used or unit prices in the allocation

## Exhibit 11 – Developer Surety Form of Bond, indicating all language of Bond document

Exhibit 1 CAM 15-1467 Page 93 of 158

## Exhibit 12 – Developer List of Subcontractors and suppliers with License Nos., as dated

Exhibit 1 CAM 15-1467 Page 94 of 158

## Exhibit 13 – Developer Affidavit, Attesting to Subcontractor/Vendor Licensure Verification, as dated

Exhibit 1 CAM 15-1467 Page 95 of 158

## Exhibit 14 – Project Master Delivery Schedule, w/summary milestone delivery items

Exhibit 1 CAM 15-1467 Page 96 of 158

## Exhibit 15 – Project Schedule, CPM Delivery schedule with itemized breakdown of Work scope by Building, and resource loaded

## Exhibit 16 – Master Schedule, Project Submittals with submittal due dates/responsible party

Exhibit 1 CAM 15-1467 Page 98 of 158

#### The City of Fort Lauderdale, Florida Exhibit C Owner Direct Material/Equipment Purchase Program

1.

2.

The Subcontractor has included Florida State Sales and other applicable taxes in his bid for material, supplies and equipment. The Owner, being exempt from sales tax, reserves the right to make direct purchases of various construction equipment, materials or supplies included in the Subcontractor's bid and/or contract, substantially in accordance with the form of Purchase Order attached herewith.

Any equipment, materials or supplies directly purchased by the Owner that are included in the Subcontractor's contract shall be referred to as Owner-Purchased Materials and the responsibilities of both Owner and Subcontractor relating to such Owner-Purchased Materials shall be governed by the terms and conditions of the procedures. The Owner will own and hold full title to all Owner-Purchased Materials.

Material suppliers shall be selected by the Subcontractor awarded the subcontract.

The Subcontractor has included the price for all construction materials in his bid. Owner purchasing of construction materials, if selected, will be administered on a deductive Change Order basis.

3. Subcontractor shall provide Developer a list of all intended suppliers, vendors, and material men for consideration as Owner-Purchased Materials. This list shall be submitted at the same time as the preliminary schedule of values. The Subcontractor shall submit a description of the materials to be supplied, estimated quantities and prices.

4. Upon request from Developer, and in a timely manner, Subcontractor shall prepare a standard Purchase Order Requisition Form in a form acceptable to the Owner and the Developer, to specifically identify the materials which Owner had, at its sole option, elected to purchase directly. The Purchase Order Requisition Form shall include:

A. The name, address, telephone number and contact person for the material supplier.

B. Manufacturer or brand, model or specification number of the item.

C. Quantity needed as estimated by the Subcontractor.

- D. The price quoted by the supplier for the materials identified therein.
- E. Any sales tax associated, with such quote.
- F. Delivery dates as established by Subcontractor.

Subcontractor shall include reference to any terms and conditions which have been negotiated with the vendors; i.e., payment terms, warranties, retainage, etc.

Such Purchase Order Requisition Forms are to be submitted to Developer's designated representative no less than fifteen (15) days prior to the need for ordering such Owner-Purchased Materials, in order to provide sufficient time for Owner review and approval and to assure that, such Directly Purchased Materials may be directly purchased by Owner and delivered to the Project site so as to avoid any delay to the Project.

- 5. After receipt of the Purchase Order Requisition Form, Owner shall prepare its Purchase Orders for equipment, materials or supplies which the Owner chooses to purchase directly. Pursuant to the Purchase Order, the vendor will provide the required quantities of material at the price established in the vendor's quote to the Subcontractor, less any sales tax associated with such price. Promptly upon receipt of each Purchase Order, Subcontractor shall verify the terms and conditions of the Purchase Order prior to its issuance to supplier and in a manner to assure proper and timely delivery of items. Owners Procurement Manager or his designated representative shall be the approving authority for the Owner on Purchase Orders in conjunction with Owner-Purchased Materials. The Purchase Order shall require that the supplier provide the required shipping and handling insurance. The Purchase Order shall also require the delivery of the Owner-Purchased Materials on the delivery dates provided by the Subcontractor in the Purchase Order Requisition Form and shall indicate F.O.B. jobsite.
- 6. In conjunction with the execution of the Purchase Orders by the suppliers, the Subcontractor shall execute and deliver to the Owner, through the Developer, one or more deductive Change Orders, referencing the full value of all Owner-Purchased Materials to be provided by each supplier from whom the Owner elected to purchase material directly, plus all sales tax savings associated with such materials in Subcontractor's bid to Developer.
- 7. All shop drawings and submittals shall be made by the Subcontractor in accordance with the Project Specifications.

Subcontractor shall be fully responsible for all matters relating to the receipt of materials furnished by Owner in accordance with these Procedures, including, but not limited to, verifying correct quantities, verifying documentation of orders in a timely manner, coordinating purchases, providing and obtaining all warranties and guarantees required by the Contract Documents, inspection and acceptance of the goods at the time of delivery, and loss, or damage to equipment and materials following acceptance of items by the Owner due to the negligence of the Subcontractor. The Subcontractor shall coordinate delivery schedules, sequence of delivery, loading orientation, and other arrangements normally required by the Subcontractor for the particular materials furnished. The Subcontractor agrees to indemnify and hold harmless the Owner and Developer from any and all claims of whatever nature resulting from non-payment of goods to suppliers arising from the actions or directions of Subcontractor. Owner purchased materials shall be stored at the construction site.

8.

9. As Owner-Purchased Materials are delivered to the Project Site, the Subcontractor and the Developer, as City's Representative, shall visually inspect all shipments from the suppliers, and approve the vendor's invoice of material delivered. The Subcontractor shall assure that each delivery of Owner-Purchased materials is accompanied by adequate to identify the Purchase Order against which the purchase is made. This documentation may consist of a delivery ticket and an invoice from the supplier conforming to the Purchase Order, together with such additional information as the Owner or Developer may require. The Developer, as Owner's Representative, shall verify in writing to the Owner the accuracy of the delivery ticket. The Subcontractor will then forward the invoice to the Owner through the Developer for payment. The invoice shall be thereupon furnished to the Finance Department for processing and payment in the manner as all other City of Fort Lauderdale invoices are processed. The Owner shall have the right to assign personnel to verify and audit the accuracy of all Director Purchase Documents.

10. The Subcontractor shall insure that Owner-Purchased Materials conform to the Specifications, and determine prior to incorporation into the Work if such materials are patently defective, and whether such materials are identical to the material ordered and match the description on the bill of lading. If the Subcontractor discovers defective or non-conformities in the Owner-Purchased Material upon such visual inspection, the Subcontractor shall not utilize such non-conforming or defective materials in the Work and instead shall promptly notify the vendor of the defective or nonconforming condition in order to pursue repair or replacement of those materials without any undue delay or interruption to the Project. Additionally, the Subcontractor shall notify the Owner, through the Developer, of such occurrence. If the Subcontractor fails to perform such inspection and otherwise incorporated Owner-Purchased Materials, the condition of which it either knew or should have known by performance of an inspection, Subcontractor shall be responsible for all damages to Owner and/or Developer resulting from Subcontractor's incorporation of such materials into the Project, including liquidated or delay damages. In the event that materials furnished are found to be defective or noconforming, the Subcontractor shall promptly take action to remedy the defect or non-conformance so as not to delay the work.

- 11. The Subcontractor shall maintain records of all Owner-Purchased Materials it incorporates into the work from the stock of Owner-Purchased Materials in its possession. The Subcontractor shall account monthly to the Owner, through the Developer, for any Owner-Purchased Materials delivered into the Subcontractor's possession, including portions of all such materials which have been incorporated into the work.
- 12. The Subcontractor, as the Owner's agent, shall be responsible for obtaining and managing all warranties and guarantees for all material and products as required by the Contract Documents. All repair, maintenance or damage-repair calls shall be forwarded to the Subcontractor for resolution with the appropriate supplier or vendor.
- 13. Notwithstanding the transfer of Owner-Purchased Materials by the Owner to the Subcontractor's possession, the Owner shall retain title and assume risk to any and all Owner-Purchased Materials.
- 14. The transfer of possession of Owner-Purchased Materials from the Owner to the Subcontractor shall constitute a bailment for the mutual benefit of the Owner and the Subcontractor. The Owner shall be considered the bailor and the Subcontractor the bailee of the Owner-Purchased Materials. Owner-Purchased Materials shall be considered returned to the Owner for the purposes of its bailment at such time as they are incorporated into the Project or consumed in the process of completing the Project. All Owner-Purchased Materials shall be stored at the construction site.
- 15. The Owner shall in no way be liable for interruption or delay in the Project, for any defects or other problems with the Project, or for any extra costs or time resulting from delay in the delivery of, or defects in, Owner-Purchased Materials when such delay is a result of the failure of the Subcontractor's performance.
- 16. On a monthly basis, Subcontractor shall be required to review invoices submitted by all suppliers of Owner-Purchased Materials delivered to the Project site during that month and either concur or object to the Owner's

issuance of payment to the suppliers; based upon Subcontractor's records of material delivered to the site and any defects in such materials.

17. In order to arrange for the prompt payment to the supplier, the Subcontractor shall provide to the Owner, through the Developer, a list indicating the acceptance of the goods or materials in accordance with the established monthly Payment Request Schedule. The list shall include a copy of the applicable Purchase Order, invoices, delivery tickets, written acceptance of the delivered items, and such other documentation as may be reasonably required by the Owner. Upon receipt and verification of the appropriate documentation, the Owner shall prepare a check drawn to the supplier based upon the receipt of data provided. This check will be released, delivered, and remitted directly to the supplier. The Subcontractor agrees to assist the Owner and Developer to immediately obtain a partial or final release of lien waiver as appropriate.

18. Salvage materials shall be the property of the Owner and stored or removed from the site by the Subcontractor at the Owner's direction.

19. The Owner's direct purchase of equipment, materials or supplies, as provided herein does not relieve the Developer or any Subcontractor of any obligation required pursuant to the contract or subcontract pertaining to the performance of work, except as to the Owner's obligation to make direct payments to such vendors and may reduce the bonds to the extent permitted by Section 255.05, Florida Statutes.

20. All obligations and duties imposed on the Developer contained in the Agreement are incorporated herein by reference.

#### The City of Fort Lauderdale, Florida Exhibit D Scope of Design Services

#### 1. DESCRIPTION OF PROJECT:

The project consists of providing design services to the City of Fort Lauderdale for the design and construction of the Fort Lauderdale Aquatic Center (FLAC).

The FLAC project currently includes the following aspects, although project elements may be added or deleted through the negotiation process prior to establishment of the Guaranteed Maximum Price (GMP) and the schedule. Once the GMP and schedule are established, any further changes to the Project and/or Developer's obligations must be done by Change Order signed by both parties.

- Final Design, permitting, and construction of an aquatic complex including, at a minimum, the following components:
  - o Bath House Facility (4 levels 30,450 sq. feet) which includes
    - Office Space
    - o Swim Team Offices
    - o Dive Team Offices
    - o Swim and Dive Meet Operations Center
    - Weight Training Rooms
    - o Recreational Rooms
    - o Meeting/Conference Rooms
    - o Locker Rooms
    - o **Restrooms**
    - o Equipment Space
    - o Concession Area
  - International Swimming Hall of Fame Facility (10,150 sq. ft. of two story shell space)
  - o FINA Competition Pool (53m x 25m)
  - Competition Pool Grandstand Seating (2,040 seats) with Canopy Shade Structure
  - o Lockers, Restrooms and Storage to be located under Grandstands
  - o Dive Pool (25m x 25 yd) with Bleacher Seating (540 seats)
  - o Dive Platforms, Springboards, Acclimation Spa
  - o Instructional Pool (20 ft x 40 ft)
  - Renovation of existing (lower level) Pool (50m x 25 yd) including the installation of a movable floor
  - o Lower-level Seating (240 seats)

- • Parking Structure (minimum of 529 spaces)
- Surface parking (uncovered) (54 spaces)
- Waterfront Public Park, Landscaping and perimeter walkway
- New Public Plaza (Seabreeze)

1.1.1.1

- The FINA Competition Pool and lower-level pool shall meet the requirements for one-time certification for competition as established by USA Swimming.
- The FINA Dive Pool and associated dive platforms and springboards shall meet the requirements for certification for competition as established by USA Diving.
- Conceptual drawings were submitted to the Owner by the Developer on March 28, 2012 for the proposed site layout and proposed building plan and exterior elevations.
- Site improvements including, but not limited to, clearing and grubbing, utility installation and connections, stormwater conveyance and detention/retention system, landscaping, and other improvements as required to complete the facility.
- Building Improvements including the construction of the Bath House Facility, and all other required components such as dumpster enclosures, equipment screens, and retaining walls as required.

The Developer will be expected to recommend value engineering concepts that can reduce initial and life cycle costs and/or add value to the project.

The project is expected to be in compliance with USGBC Standards, for sustainable buildings, but will not be LEED certified for new construction.

The Owner intends to pursue sales tax recovery to the greatest degree practicable.

- 2. PROGRAM VERIFICATION PHASE:
- 2.1. During the various design phases of this Agreement, Developer will provide Owner with value engineering suggestions and other services with respect to Developer's design. Those services shall include but are not limited to the preparation of cost estimates and comments concerning the constructability of the design.

- 2.2. Developer shall review the Owner's Central Beach Master Plan (revised November 30, 2009) and will either submit a written statement that the design documents will be prepared in accordance with the requirements set forth therein or Developer will submit a written request for variance, identifying specific exceptions. Owner may, at its discretion, accept or reject the request for variance.
- 2.3. Developer shall review the Owner's Construction Standards and Specifications and will either submit a written statement that the design documents will be prepared in accordance with the requirements set forth therein or Developer will submit a written request for variance, identifying specific exceptions. Owner may, at its discretion, accept or reject the request for variance.
- 2.4. At the conclusion of the Program Verification and during the various design phases of this Agreement, Developer will provide Owner with cost estimates and comments concerning the constructability of the design.

Developer shall work with the Owner's Director of Public Works, Owner's Director of Parks and Recreation, Owner's Project Manager and other users of the Project to meet design requirements and identify the areas within the facility design, which offer the greatest potential for the elimination of unnecessary costs. Owner's Standards shall not be eliminated as value engineering items.

2.5. As part of the Design Services, Developer will be required to provide Owner with a cost estimate as part of the Program Verification Phasetogether with a written explanation for all variances between that budget estimate and Owner's approved Project construction budget. The cost estimate format shall be subject to Owner's approval and may require electronic submission of cost estimate information. If Developer's budget estimate or any other estimate prepared by or for Owner indicate that construction costs will exceed the GMP, Owner may elect to modify its budget and/or require Developer to revise the design elements to bring them within the GMP. Developer shall be solely responsible for all costs and expenses which it may incur in revising the design elements to bring them within the GMP. Notwithstanding anything contained herein to the contrary, in no event will Developer be obligated to commence construction of the Project unless and until all costs for the Design Services and Construction Services (together with the Design Fee and Developer's Fee) under the approved Construction Documents do not exceed the GMP and the Contract Time shall be extended accordingly, if necessary, with the parties having all rights as provided in Section 3.8 of the Agreement.

#### 3. PHASE 1 - DESIGN DEVELOPMENT PHASE:

After Owner's review and written approval of the Phase I – Program Verification and issuance of Owner's written Notice to Proceed. Developer shall commence the Phase II – Design Development Phase services and perform the following:

- 3.1. Develop design documents to a level of definitiveness and detail to fix and describe the size and character of the various Project components and each Project discipline and system as may be appropriate for this stage of development, including long lead and special order materials and equipment, which will permit determination of whether the facility can be satisfactorily constructed in all task areas by all disciplines within the GMP.
- 3.2. Continue developing the architectural, structural, mechanical, electrical, security, and other discipline's responsibilities to establish the final scope and details for that discipline's work.
- 3.3. Perform materials research and prepare specifications specific to Project requirements in draft form.
- 3.4. Identify and properly coordinate the requirements of the various utility services that have an impact upon the Project Design. Drainage investigations and drainage designs shall be coordinated with storm water management district having jurisdiction on the site.
- 3.5. Continue to develop Project construction cost estimates and the overall Project schedule.
- 3.6. Generate alternative ideas through Value Engineering Workshops with the Owner to provide the identified primary function for the Project.
- 3.7. Evaluate alternative ideas in terms of their feasibility to construct, time and cost.
- 3.8. Develop selected alternative ideas in detail with emphasis on their technical durability, constructability and life cycle cost.
- 3.9. Developer's final Phase II Design Development submittal and presentation shall include +/- 75% design documents.
- 3.10. Developer shall prepare and submit with Phase II Design Development Documents a quality assurance/quality control (QA/QC) itemized checklist, conferring that the Phase II – Design Development Documents submittal is in compliance with the Owner's program submission requirements and that all project design disciplines have been coordinated. 3.11. Developer shall conduct a pre-submittal document review meeting with the Owner's

Project Manager prior to submission of the Phase II - Design Development Documents. Developer, may be required, to conduct at least one presentation at completion of this Phase II to demonstrate how Owner's previously submitted comments were incorporated into the design documents.

- 3.11. All Phase II Design Development Documents prepared by or for Developer are subject to Owner's review and approval. At completion of the Design Development Phase, Developer shall submit the Phase II -Design Development Documents to the Project Manager for review and comment. Developer shall respond in writing to the review comments within 7 days of receipt. Responses shall be forwarded directly to the Owner's Project Manager. Developer shall revise the Phase II – Design Development Documents as required by Owner's Project Manager in order to obtain Owner's written approval and authorization to proceed to the Phase III- Construction Documents Phase; provided, however, in no event will Developer be required to incorporate Owner's comments into the Project if to do so would result in the GMP being exceeded, with the parties having all rights as provided in Section 3.8 of the Agreement.
- 3.12. As part of the Design Services, Developer will be required to further develop and update the budget estimate it initially prepared as part of its Program Verification Phase services and bring to Owner's attention in writing any variances between that updated budget estimate and the GMP. If Developer's updated budget estimate or any other estimate prepared by or for Owner based upon the Design Development Documents indicate that construction costs will exceed the GMP. Owner may elect to modify its budget and/or require Developer to revise the Design Development Documents to bring them within Owner's approved Project construction budget. Developer shall be solely responsible for all costs and expenses which it may incur in revising the Design Development documents to bring them within Owner's approved Project construction budget. Notwithstanding anything contained herein to the contrary and any rights of Owner to review and request changes to the Design Development Documents, in no event will Developer be obligated to commence construction of the Project unless and until the costs for the Design Services and Construction Services (together with, the Design Fee and Developer's Fee) under the approved Construction Documents do not exceed the GMP, in Developer's sole discretion, and the Contract time shall be extended accordingly, if necessary, with the parties having all rights as provided in Section 3.8 of the Agreement.
- 4. PHASE III CONSTRUCTION DOCUMENTS PHASE:
After Owner's review of the Phase II – Design Development Documents and issuance of Owner's written authorization to proceed, Developer shall commence the Phase III - Construction Documents Phase services and perform the following:

- 4.1. Prepare final calculations, Construction Documents setting forth in detail each discipline's requirements into a cohesive whole based upon the approved Phase II – Design Development Documents and consult with Owner.
- 4.2. Prepare final quality assurance/quality control (QA/QC) itemized checklist, confirming that the Phase III Construction Documents submission is in compliance with the Owner's program submission requirements and that all project design disciplines have been coordinated.
- 4.3. Complete the Project manual in accordance with Owner's Construction Standards and Specifications.
- 4.4. Review and advise Owner regarding final Project schedule and GMP.
- 4.5. Prepare and file all applications, data and documents required to obtain the approval of all authorities having permit jurisdiction over the Project Developer shall advise Owner and schedule the necessary contacts and liaison with all authorities having permit jurisdiction over the Project, and shall furnish, on a timely basis, such plans, data and information as may be necessary to secure approval of the required permits. Developer shall, at no additional cost to Owner, make all reasonable and necessary construction plan revisions required to obtain the necessary permit approvals for construction of the Project.
- 4.6. As part of the Design Services, Developer will be required to further develop and update its budget estimate as part of the Phase III -Construction Documents, and notify Owner in writing of any variances between that updated budget estimate and the GMP. If Developer's updated budget estimate or any other estimate prepared by or for Owner based upon the Phase III – Construction Documents indicate that construction costs will exceed the GMP, Owner may elect to modify its budget and/or require Developer to revise the Construction Documents to bring them within Owner's approved Project construction budget. Developer shall be solely responsible for all costs and expenses which it may incur in revising the Construction Documents to bring them within Owner's approved Project construction budget. Notwithstanding anything contained herein to the contrary, and any rights of Owner to review and request changes to the Construction Documents, in no event will Developer be obligated to commence construction of the Project unless and until the costs for the Design Services and Construction Services

(together with the Design Fee and Developer's Fee) under the approved Construction Documents do not exceed the GMP, and the Contract Time shall be extended accordingly, if necessary, with the parties having all rights as provided in Section 3.8 of the Agreement.

- 4.7. All Phase III - Construction Documents prepared by or for Developer are subject to Owner's review and approval. At completion of the Construction Documents Phase, Developer shall submit the Phase III -Construction Documents to the Project Manager for review and comment. Developer shall respond in writing to the review comments within 14 days of receipt. Responses shall be forwarded directly to the Owner's Project Manager. Developer shall revise the Phase III-Construction Documents as required by Owner to obtain Owner's written approval of such documents. Notwithstanding anything contained herein to the contrary, and any rights of Owner to review and request changes to the Construction Documents, in no event will Developer be obligated to commence construction of the Project unless and until the costs for Design Services and Construction Services (together with the Design Fee and Developer's Fee) under the approved Construction Documents do not exceed the GMP, and the Contract Time shall be extended accordingly, if necessary, with the parties having all rights as provided in Section 3.8 of the Agreement.
- 4.8. As used herein, the term "Construction Documents" refers to all documents to be prepared by and for Developer pursuant to this Agreement with respect to the construction of the Project, including, but not limited to, all drawings, specifications, bid documents, Project Manual, contract conditions, and Addenda.
- 5. PHASE IV BUILDING PERMIT PHASE:
- 5.1. Developer is responsible for obtaining all necessary and required permits for the Project.
- 5.2. As part of the Building Permit application package, the Developer shall provide the applicable Building Permit office with the number of complete sets of signed and sealed Construction Documents and all other bidding documents prepared by the Developer as indicated in the Deliverable Schedule set forth in the Guidelines. Each of the drawings and the cover sheet of the Project Manual shall be signed, sealed, and dated by the Developer per Florida Statutes.
- 5.3. As part of the Building Permit application process, the Building Official shall review and provide comments to the Developer on the submitted Construction Documents. Developer shall revise the Construction Documents by incorporating necessary revisions to address mandatory

inclusions made by the Building Official. The revised final Construction Documents shall be signed and sealed by the Developer and re-submitted to Owner in the quantities indicated in the Deliverable Schedule. Once this revised set of Construction Documents is approved by Owner, it will be deemed to be the final approved set of Construction Documents upon which the construction of the Project is to be based.

#### 6. SERVICES RELATING TO ALL PHASES:

- 6.1. Developer shall investigate and confirm in writing to Owner, to the best of Developer's knowledge, conformance of the Plans and Specifications with all applicable Building Code provisions together with local public and utility regulations.
- 6.2. Developer shall ensure compliance with USGBC Standards for sustainable buildings, but will not be certified as LEED.
- 6.3. Developer shall furnish check prints for every Project phase including ten (10) sets at the completion point of each phase. Developer shall furnish to Owner one (1) electronic set at the completion of each phase. The electronic set to be provided to Owner at 100% Construction Documents shall be fully conformed to incorporate all addenda issued prior to bidding.
- 6.4. At all phases of design, Developer's design documents must be consistent with Owner's Construction Standards and Specifications unless expressly authorized otherwise in writing by Owner.
- 6.5 Developer shall submit to Owner design notes and computations to document the design conclusions reached during the development of the Project design as requested by Owner.
  - 6.5.1. When the plans are submitted for final review, the design notes and computations corrected for any Owner comments shall be resubmitted. At the Project completion, a final set of the design notes and computations, properly certified by Developer, shall be submitted with the record set of plans and tracings.
  - 6.5.2. The design notes and calculations shall include, but not be limited to, the following data:
    - 6.5.2.1 Design criteria used for the Project;
    - 6.5.2.2 Structural calculations;
    - 6.5.2.3 Drainage calculations;

- 6.5.2.4 Calculations as required by provisions of the Florida Energy Conservation Manual (Department of General Services), latest revision;
- 6.5.2.5 Calculations showing probable cost comparisons of various alternatives considered;
- 6.5.2.6 Documentation of decisions reached resulting from meetings, telephone conversations or site visits; and
- 6.5.2.7 Other Project related correspondence as appropriate.
- 6.6. All drawing documents for the Project shall be accurate, legible, complete in design and drawn to scales acceptable to Owner. The completed drawing documents shall be furnished on reproducible material and in a size and format which is reasonably acceptable to Owner.
- 6.7. Owner in no way obligates itself to check Developer's work, and further, is not responsible for maintaining the Design Schedule, except with respect to providing timely responses and/or approvals as provided in the Contract Documents.
- 6.8. Owner's approval or acceptance of any service in any phase does not relieve Developer of any of its duties, obligations or responsibilities under this Agreement.
- 6.9. All design services performed by Developer and any documents prepared by Developer and submitted to Owner shall meet Section 255.2575, Florida Statutes on energy-efficient and sustainable buildings.
- 6.10. Developer acknowledges that Owner may establish construction budgets for any particular portions of the Project, which budgets may be subject to adjustments, as mutually approved by Owner and Developer in writing.
- 6.11. Developer represents to Owner that it has expertise, or has engaged professional firms who have expertise, in the type of professional architectural and/or engineering services that will be required for the Project. Drawings shall be prepared in electronic format acceptable to the Owner and the Project Manual shall be prepared as an electronic Microsoft Word document per Owner's standards. By execution of this Agreement, Developer acknowledges it has received the most recent version of the Owner's Standards & Specifications as of the date of this Agreement and will follow, observe and design in accordance with the standards, requirements and conventions set forth therein. Developer agrees that all Work to be provided by Developer pursuant to this Agreement shall be subject to Owner's reasonable review and approval and shall be in accordance with all applicable published laws, statutes, ordinances, codes

(including City's ULDR), rules, regulations (including utility regulations), local and state fire marshal requirements and the Florida Building Code, as well as the requirements of any governmental agencies which regulate or have jurisdiction over the Project or the Work to be provided and performed by Developer hereunder. In the event of any conflicts in these requirements, Developer shall promptly notify Owner of such conflict in writing and utilize its best professional judgment to resolve the conflict. Owner's approval of the design documents in no way relieves Developer of its obligation to deliver complete and accurate documents necessary for successful construction of the Project.

## The City of Fort Lauderdale, Florida Exhibit E Scope of Construction Services

#### PHASE V - Construction Services

Developer shall provide the following services in addition to all other Construction Services required by the terms of this Contract:

- 1.1 Prepare a list of required submittals for shop drawings, product data, samples, warrantees, and other submittals required by Contract Documents, in tabular form which will indicate specification section number and section name per Project Manual Table of Contents.
- 1.2 Process, review, respond and distribute shop drawings, product data, samples, substitutions and other submittals required by the Construction Documents within ten (10) Business Days of receipt.
- 1.3. Maintenance of a master file of all submittals, including submittal register, made to Developer, with duplicates for Owner. Owner's copy shall be in electronic/CD format and submitted at time of Substantial Completion.
- 1.4 Prepare, reproduce and distribute supplemental drawings, specifications and interpretations in response to requests for clarification by Owner as required by construction exigencies. Developer's response to any such request must be received by Owner within ten (10) Business Days. Developer will review and respond to all submittals, including but not limited to shop drawings, within a reasonable period of time so as not to delay the progress of the Work, but in no event, more than ten (10) Business Days, unless Owner expressly agrees otherwise in writing.
- 1.5. Developer shall submit to the applicable Building Permit office the number of sets of drawings and/or documents reflecting the approved changes in the Work as may be required by that office. Code compliance issues must be approved by the applicable Building Permit office prior to inspection of the subject Work.
- 1.6. Developer shall arrange for all Project Site facilities as required by Owner and necessary to enable Developer to perform its respective duties and to accommodate any representatives of Owner which Owner may choose to have present on the job, the description of such facilities to be included in the GMP.
  - 1.6.1. Tangible personal property, otherwise referred to as Project Site facilities, include, but are not limited to such things as trailers, toilets, typewriters, computers and any other equipment necessary to carry on the Work.

- 1.7. Developer's administration of the Work shall include the following:
  - 1.7.1. Maintain a log of daily activities, including manpower records, weather, delays, major decisions, etc.
  - 1.7.2. Maintain a roster of companies on the Project with names and telephone numbers of key personnel.
  - 1.7.3. Establish and enforce job rules governing parking, clean-up, use of facilities and worker discipline.
  - 1.7.4. Provide labor relations management for a harmonious, productive Project.
- 1.8. Developer also shall provide job site administration functions during construction to assure proper documentation, including but not limited to the following:
  - 1.8.1. Job Meetings: Developer shall attend meetings such as preconstruction conferences, progress meetings, job conferences, precloseout meetings, and other Project-related meetings, as may be directed by Owner. Developer's Architect is also required to attend any such meetings. Developer shall provide meeting minutes for these meetings. Conduct a preconstruction conference with each subcontractor after award of the subcontract and prior to the start of its portion of the Work. Hold weekly progress and coordination meetings, or more frequently if required by Work progress, to provide for the timely completion of the Work. In addition, Developer shall arrange and conduct regular Project status meetings with Owner at intervals to be determined by the Owner.
  - 1.8.2. Developer shall use the job site meetings as a tool for the preplanning of Work and enforcing schedules, and for establishing procedures, responsibilities, and identification of authority for all parties to clearly understand. During these meetings, Developer shall identify the party or parties responsible for following up on any problems, delay items or questions, and Developer shall note the action to be taken by such party or parties. Developer shall revisit each pending item at each subsequent meeting until resolution is achieved. Developer shall attempt to obtain from all present any problems or delaying event known to them for appropriate attention and resolution.
  - 1.8.3. <u>Shop Drawing Submittals/Approvals</u>: Provide staff to review and approve shop drawings and other submittals and to implement procedures for transmittal to Developer of such submittals for

action, and closely monitor their review process. Owner reserves the right to review the shop drawings and other submittals and require approval on such shop drawings and other submittals; with any review and approval by Owner being within ten (10) Business Days after receipt.

- 1.8.4. <u>Material and Equipment Expediting</u>: Provide staff to closely monitor material and equipment deliveries, check and follow-up on supplier commitments for all subcontractors and maintain a material and equipment expediting log.
- 1.8.5. <u>Payments to Subcontractors</u>: Develop and implement a procedure for the review, processing and payment of applications by subcontractors for progress and final payments.
- 1.8.6. Document Interpretation: Respond to all questions for interpretation of the Contract Documents made by subcontractors.
- 1.8.7. <u>Reports and Project Site Documents</u>: Record the progress of the Work. Submit written progress reports to Owner, including information on subcontractors' Work, and the percentage of completion. Keep a daily log available to Owner and any permitting authority inspectors.
- **1.8.8.** <u>Subcontractors Progress</u>: Prepare periodic punch lists for subcontractors' work including unsatisfactory or incomplete items and schedules for their completion.
- 1.8.9. <u>Substantial Completion</u>: Notify Owner in writing when the Work or designated portions thereof are ready for the Substantial Completion inspections. From the Punch Lists of incomplete or unsatisfactory items prepared by Developer and reviewed and supplemented by Owner, prepare a schedule for their completion indicating completion dates for Owner's review.
- 1.8.10. <u>Final Completion</u>: Monitor the subcontractors' performance on the completion of the Work and provide notice to Owner when the Work is ready for final inspection. Secure, review and certify compliance with the Contract Documents, then transmit to Owner all required guarantees, warranties, affidavits, releases, bonds, waivers, manuals, record drawings, and maintenance books.
- **1.8.11.** <u>Start-Up</u>: With Owner's personnel, direct the check-out of utilities, operations, systems and equipment for readiness and assist in their initial start-up and testing by the subcontractors.

- 1.8.12. <u>Record Drawings</u>: Developer shall monitor the progress of its own forces and its subcontractors on marked up field prints which shall be developed by Developer into the final record drawings.
- 1.9. Developer shall maintain at the Developer's office or field office, originals or copies of, and on electronic file copy of, on a current basis, all Project files and records, including, but not limited to, the following administrative records:
  - **1.9.1.** Subcontracts and Purchase Orders
  - 1.9.2. Shop Drawing Submittal/Approval Logs
  - 1.9.3. Equipment Purchase/Delivery Logs
  - 1.9.4. Contract Drawings and Specifications with Addenda
  - 1.9.5. Warranties and Guarantees
  - 1.9.6. Cost Accounting Records
  - 1.9.7. Labor Costs
  - 1.9.8. Material Costs
  - 1.9.9. Equipment Costs
  - 1.9.10. Cost Proposal Request
  - 1.9.11. Payment Request Records
  - 1.9.12. Meeting Minutes
  - 1.9.13. Cost-Estimates
  - 1.9.14. Intentionally Omitted
  - 1.9.15. Lab Test Reports
  - 1.9.16. Insurance Certificates and Bonds
  - 1.9.17. Contract Changes
  - 1.9.18. Permits
  - 1.9.19. Material Purchase Delivery Logs
  - 1.9.20. Technical Standards
  - 1.9.21. Design Handbooks
  - 1.9.22. "As-Built" Marked Prints
  - 1.9.23. Operating & Maintenance Instruction
  - 1.9.24. Daily Progress Reports
  - 1.9.25. Monthly Progress Reports
  - 1.9.26. Correspondence Files
  - 1.9.27. Transmittal Records
  - 1.9.28. Inspection Reports
  - 1.9.29. Bid/Award Information

1.9.30. Bid Analysis and Negotiations

1.9.31. Punch Lists

- 1.9.32. Intentionally Omitted
- 1.9.33. Intentionally Omitted
- 1.9.34. Subcontractor licenses

The Project files and records shall be available at all times to Owner or its designees for reference, review or copying.

- 1.10. Developer shall provide the following services with respect to the Work, to facilitate the smooth, successful and timely occupancy of the Project by Owner:
  - 1.10.1. Developer shall provide consultation and Project management to facilitate Owner's occupancy of the Project and provide transitional services to place the Work "on line" in such conditions as will satisfy Owner's operations requirements. The services include Developer's coordination of the delivery of Owner supplied furniture, fixtures and equipment for the Project prior to Substantial Completion.
  - 1.10.2. Developer shall catalog operational and maintenance requirements of equipment to be operated by maintenance personnel and convey these to Owner in such a manner as to promote their usability. Developer shall provide Owner's operations and maintenance personnel with operations and maintenance training with respect to the equipment and systems being provided as part of the Work. This training may be videotaped by Owner for subsequent presentation to Owner's operations and maintenance personnel.
  - 1.10.3. Developer shall secure required guarantees and warranties, and shall assemble and deliver same to Owner in the manner required by Owner
- 1.11 Developer will revise the final approved Construction Documents to incorporate all "As-Built" information contained in the Developer's marked-up "As-Built" drawings and specifications, as well as to reflect all addenda, contract changes and field changes (sometimes referred to herein as the "Record Documents"). Developer shall provide Owner with one (1) electronic copy on compact disk (CD) of the Record Documents, two sets of the conformed, signed and sealed drawings and prints, and two sets of the conformed Project Manual/Specifications.
  - 1.11.1 The electronic copy on CD of the Record Documents shall be provided in "pdf" or "dwg" format.
  - 1.11.2 Throughout the Construction Phase, Developer shall mark-up "As-Built" drawings and Project Manual/Specifications, on a regular basis to reflect all "As-Built" conditions at the Site, maintaining such "As-Built" drawings and specifications is a condition precedent to Developer's entitlement to payment hereunder.

1.12 Developer, upon final acceptance of the Project, shall issue and sign the Certificate of Final Inspection in a form approved by the Owner.

## [REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

 $(x_1,y_1)$ 

## The City of Fort Lauderdale, Florida Exhibit F Form of Payment Bond

BOND

NO.\_\_\_\_\_

#### PUBLIC PAYMENT BOND

KNOW ALL MEN BY THESE PRE	ed at (Business Address) are held and actually bound to, as Obligee in the sum of
as Principal, and	, as Surety,
located at	(Business Address) are held and
Contractually bound to	as Obligee in the sum of
(\$) for the payment	whereof we bind ourselves, our heirs,
executors, personal representatives, suc	cessors and assigns, jointly and severally.

**WHEREAS,** Principal has entered into a Design-Build contract dated as of the \_\_\_\_\_ day of \_\_\_\_\_\_, 201\_, with Obligee for the design and construction of:

in accordance with drawings and specifications, which contract is incorporated by reference and made a part hereof, and is referred to as the Contract.

#### **THE CONDITION OF THIS BOND** is that if Principal:

1. Promptly makes payment to all claimants as defined in Section 255.05(1), Florida Statutes, supplying Principal with labor, services, materials or supplies, used directly or indirectly by Principal in the prosecution of the Work provided for in the Contract, then this bond is void; otherwise it remains in full force.

Any changes in or under the Contract and compliance or noncompliance with any formalities connected with the Contract or the changes do not affect Surety's obligation under this Bond.

The provisions of this bond are subject to the time limitations of Section 255.05(2). In no event will the Surety be liable in the aggregate to claimants for more than the penal sum of this Payment Bond, regardless of the number of suits that may be filed by claimants.

**IN WITNESS WHEREOF,** the above parties have executed this instrument this \_\_\_\_\_ day of \_\_\_\_\_\_, 201\_, the name of each party being affixed and these presents duly signed by its undersigned representative, pursuant to authority of its governing body.

Signed, sealed and delivered in the presence of:

#### **PRINCIPAL:**

Ву:\_\_\_\_\_ Witnesses as to Principal Name: lts: STATE OF \_\_\_\_\_ COUNTY OF The foregoing instrument was acknowledged before me this \_\_\_\_ day of \_\_\_\_\_\_ 201\_\_\_ by \_\_\_\_\_\_ as \_\_\_\_\_\_ of \_\_\_\_\_, a \_\_\_\_, a \_\_\_, a \_\_, a \_\_\_, a \_\_\_, a \_\_\_, a \_\_\_, a \_\_, a \_\_\_, a \_\_, a \_\_\_, a \_\_\_, a \_\_\_, a \_\_\_, a \_\_, a \_\_\_, a \_\_\_, a \_\_\_, a \_\_\_, a \_\_, a \_\_\_, a \_\_\_, a \_\_, a \_\_\_, a \_\_\_, a \_\_\_, a \_\_\_, a \_\_, a \_\_\_, a \_\_, a \_\_\_, a \_\_, a \_\_\_, a \_\_, a has produced \_\_\_\_\_\_ as identification. My Commission Expires: Notary Public (Signature) (AFFIX NOTARY SEAL) (Printed Name) (Title or Rank) (Serial Number, if any) SURETY: ATTEST: (Printed Name) (Business Address)

Witnesses as to Surety

(Authorized Signature)

(Printed Name)

# OR

Witnesses

As Attorney in Fact (Attach Power of Attorney)

(Business Address)

(Printed Name)

(Telephone Number)

STATE OF \_\_\_\_\_\_

The	fore	going	instrument	Wa	as acknowl	edged	befo	re n	ne tl	his _	day	of
		_, 201_	_, by							, a:	s	
		, of _	-								Surety,	
behalf	of	Surety	. He/She		personally as identifica		to	me	OR	has	produ	ced

My Commission Expires:

(AFFIX NOTARY SEAL)

Notary Public (Signature)

(Printed Name)

(Title or Rank)

(Serial Number, if any)

## The City of Fort Lauderdale, Florida Exhibit G Form of Performance Bond

BOND NO.

## PERFORMANCE BOND

KNC	DW ALL	MEN B	Y THESE	PRES	ENTS	S: Tha	ıt <u> </u>	81. <u> </u>			
as Principa	l, and _				N			a	s Su	irety, lo	cated
at			· 4 · -				(Bu	isiness	A A	ddress)	are
held and	contra	ctually	bound	to _		i				(″Owne	er")as
Obligee	in '	the	sum	of							
(\$		) for	the pay	ment	wh	ereof	we bin	d ours	selve	s, our	heirs,
executors,	persona	l repre	sentativ	es, suc	cess	ors ar	nd assigr	is, join	tly aı	nd sever	ally.
WH	ereas,	Princip	al has ei	nterec	int int	o a De	esign Bu	ild cor	ntrac	t dated	as of
the	day of _			_, 20_		with	Obligee	for	the	design	and
constructio	on of			<del></del>				,			

in accordance with drawings and specifications, which contract is incorporated by reference and made a part hereof, and is referred to as the Contract.

THE CONDITION OF THIS BOND is that if Principal:

- 1. Performs the Contract at the times and in the manner prescribed in the Contract; and
- 2. Promptly makes payment to all claimants, as defined in Section 255.05 (1), Florida Statutes, supplying Principal with labor, materials, or supplies, used directly or indirectly by Principal in the prosecution of the work provided for in the contract; and
- 3. Pays Obligee any and all losses, damages, expenses, costs and attorneys' fees, including appellate proceedings, that Obligee sustains because of any default by Principal under the Contract, including, but not limited to, all delay damages, whether liquidated or actual, incurred by Obligee; and
- 4. Performs the guarantee of all work and materials furnished under the Contract for the time specified in the Contract,

then this bond is void; otherwise it remains in full force.

Any changes in or under the Contract and compliance or noncompliance with any formalities connected with the Contract or the changes do not affect Surety's obligation under this Bond. The Surety, for value received, hereby stipulates and agrees that no changes, extensions of time, alterations or additions to the terms of the Contract or other Work to be performed hereunder, or the specifications referred to therein shall in anyway affect its obligations under this bond, and it does hereby waive notice of any such changes, extensions of time, alterations or additions to the terms of the Contract or to Work or to the specifications.

This instrument shall be construed in all respects as a common law bond. It is expressly understood that the time provisions and statute of limitations under Section 255.05, Florida Statutes, shall not apply to this bond.

In no event will the Surety be liable in the aggregate to Obligee for more than the penal sum of this Performance Bond regardless of the number of suits that may be filed by Obligee.

> Exhibit 1 CAM 15-1467 Page 124 of 158

. ;

**IN WITNESS WHEREOF,** the above parties have executed this instrument this \_\_\_\_\_ day of \_\_\_\_\_\_, 20\_\_\_, the name of each party being affixed and these presents duly signed by its undersigned representative, pursuant to authority of its governing body.

Signed, sealed and delivered in the presence of:

## PRINCIPAL:

Witnesses as to Principal

By:		 
Name:		
lts:	:	 

STATE OF \_\_\_\_\_\_ COUNTY OF \_\_\_\_\_\_

The foregoing instrument was acknowledged before me this \_\_\_\_ day of \_\_\_\_\_\_, 20\_\_, by \_\_\_\_\_\_, as \_\_\_\_\_\_ of \_\_\_\_\_, a \_\_\_\_\_\_, a \_\_\_\_\_\_, corporation, on behalf of the corporation. He/she is personally known to me **OR** has produced \_\_\_\_\_\_ as identification.

My Commission Expires:

Notary Public (Signature)

(AFFIX NOTARY SEAL)

(Printed Name)

(Title or Rank)

(Serial Number, if any)

Witnesses

Witnesses as to Surety

SURETY:

(Printed Name)

(Business Address)

(Authorized Signature)

(Printed Name)

. OR

As Attorney in Fact (Attach Power of Attorney)

(Business Address)

(Printed Name)

(Telephone Number)

Exhibit 1 CAM 15-1467 Page 126 of 158 STATE OF \_\_\_\_\_\_ COUNTY OF \_\_\_\_\_\_

and the second second

· .

The foregoing instrument was acknowledged before me this \_\_\_\_ day of \_\_\_\_\_, 201\_\_\_, by \_\_\_\_\_\_

of \_\_\_\_\_\_, Surety, on behalf of Surety. He/She is personally known to me **OR** has produced \_\_\_\_\_\_as identification.

My Commission Expires:

(AFFIX NOTARY SEAL)

Notary Public (Signature)

(Printed Name)

(Title or Rank)

(Serial Number, if any)

Exhibit 1 CAM 15-1467 Page 127 of 158

. .

## The City of Fort Lauderdale, Florida Exhibit H

## Release and Affidavit

STAT	E OF FLO	RIDA		)			
cou	NTY OF	-		)			
	Before			undersigned , who after			
····-	Before	me,		undersigned , who after			
<b>1</b> .	\$ and wai assigns, contraci corpora in any v	ves for all cla or in te exist vay to f	itself ims d tort, a ing un he pe	the Contract paid, and its subcor emands, dama against The Cit der the laws of rformance of th , 201,	ntractors, ma iges, costs a y of Fort La the State of ne Agreemen	("Develope terialmen, suc nd expenses, uderdale, Flor Florida ("Own t between De	r") releases cessors and whether in ida, a body er") relating veloper and
2.	success lands, li which a	ors and censes lien or	d assig and o a der	for itself ar gns, that all c ther expenses nand against a nd paid, except	harges for l for which Ow ny payment l	abor, materia /ner might be oond might b	ls, supplies, sued or for e filed, have
3.	demand	ls or su	its, act	indemnify, defitions, claims of g out of the p	liens or othe	r charges fil <mark>e</mark> d	or asserted

against Owner arising out of the performance by Developer of the Work covered by this Release and Affidavit; provided, however, that Owner has timely and fully satisfied all of its payment obligations under the Contract Documents.

- 4. Except as otherwise provided herein, Developer certifies that it has paid all its subcontractors and materialmen in full all amounts owed them from any previous payments received by Developer from Owner and has not withheld any such amounts.
- 5. This Release and Affidavit is given in connection with Developer's [monthly/final] Application for Payment No.\_\_\_\_.

•			·
Witnesses:		Developer:	
		Ву:	
	<b>-</b>	lts:	
Print Name			
·		Date:	
hint Nomo			
Print Name			
[Corp	orate Seal]	• •	
	•		
TATE OF	_		· .
201, by a a corporation. He/She is personal (state) driver's license no.	cor ly known to me o	poration, on behal r has produced a	f of th 
Ay Commission Expires:		· · · · · · · · · · · · · · · · · · ·	
· · ·	Notary P	ublic (Signature)	
AFFIX NOTARY SEAL)			
	۲) (۲	rinted Name)	
	. <b>(Т</b>	itle or Rank)	
	(S	erial Number, if any)	
	( <b>)</b>	enarivanioer, a ariy)	
		• •	
		•	

## Exhibit I

.

د الموجه بالمربعية العربية المحادث المراجع المحادثة المراجع المحاد المحار محمر موادف المراجع والمحادي المراجع المراجع الم			-	
	The Archield and a second and a	CERENE Provension	NONY SEVENCE	
Recreational Design & Construction,	954-566-3885	Developer/Design/Builder	Lexington Insurance Co	2,000,000 Each Claim
3990 N. Powerline Road				2,000,000 Aggregate Limit
Oakland Park, FL 33309		-		
Zyscovich	305-372-5222	Master Planner & Architect	Lexington Insurance Co	3,000.00 Each Claim
100 N Biscayne Blvd 27th Floor				5,000,000 Annual Aggregate
Miami, Florida 33132	· · · · · · · · · · · · · · · · · · ·			50,000 Deductible Each Claim
Counsliman-Hunsker	314-894-1245	Aquatics Engineering	Evanston Insurance Company	2,000,000 Each Claim
10733 Sunset Office Drvie 4th Floor	LindaDuerst@chh2o.com			2,000,000 Aggregate Limit
St Louis, Mo. 63127		<u></u>		
Corzo Castella Carbello Thompson Salman	954-565-2113	Civil Engineering	Continental Casualty Company	2,000,000 Per Occ
3996 NW 9 Avenue				2,000,000 Aggregate
Fort Lauderdale, Florida 33309			· · · · · · · · · · · · · · · · · · ·	
Loucks Engineering, Inc	954-938-1881	Mechancial Engineering	The Camden Fire Insurance	1,000,000 Per Claim Limit
4529 NE 18 Avenue Suite 200	rita@louckseng.com			1,000,000 Annual Aggregate
Fort Lauderdale, Florida 33334			·	
Lakdas/Yohalem Engineering, Inc	954-771-0630	Structural Engineering	RLI Insurance Company	1,000,000 Per Claim Limit
2211 NE 54 Street	lye@lyengineering.com			2,000,000 Aggregate Limit
Fort Lauderdale, Florida 33308				
Construction Consultants of Florida, Inc	954-584-7904	Electrical Engineering	Colony Insurance Company	250,000 / 2500 Ded
1445 SW 21 Avenue				
Fort Lauderdale, Florida 33312	- -			

.

Exhibit 1 CAM 15-1467 Page 130 of 158

## Schedule 3.10.3 Basis of the GMP

#### Re: Redevelopment of the Fort Lauderdale Aquatic Complex.

Recreational Design and Construction, Inc. is pleased to submit this Scope of work for the Public portion of the Redevelopment of the Fort Lauderdale Aquatic Complex. This scope of work outlines our understanding of the project requirements and was used to establish the GMP for Design-Build Services to include all design, engineering, project management and construction for the proposed Aquatic Facility.

The Scope of Work & cost estimate is in accordance with the Schematic design for the FINA competition pool (53m x 25m), the renovated 50-meter competition pool, 20' x 40' instructional pool, 25m x 25y Dive Pool, Bath House Facilities including a shell space for the International Swimming Hall of Fame, Parking structure, surface parking and Waterfront Greenway.

The GMP is based on our team's extensive experience with similar aquatic complexes and the past 28 months of working with city staff, mayor and commission. We have outlined below our understanding of the project requirements and the associated cost. The cost of the Public improvements, as of March 28, 2012, shall be \$24,864,950.00 for the pools and Bath House Facility, with a cost of \$7,572,484.00 for the parking structure. Both elements combined make up the total GMP of \$32,437,434.00.

#### We understand the work to include the following:

#### Site Work

- Temporary construction barrier shall be erected to protect the existing 50m pool from demolition/construction activities.
- Walkways and new entry points shall be designed and constructed in compliance with Federal ADA requirements.
- A main loop stabilized earth, emergency vehicle accessway, shall be designed and constructed around the entire site.
- All existing building structures, bleachers, dive pool, dive tower and elevated 50m pool shall be demolished.
- The area within the limits of construction shall be leveled, graded and shaped to create the proper building pads, asphalt paved areas, storm water retention and drainage areas, parking garage and lower-level pool deck area.

#### Site Electrical.

• The electric power shall be connected to the existing service provider's transformer located within the limits of construction.

- The electrical service shall be constructed to provide power for the Bath House, Parking Structure, Pool Equipment, and site lighting.
- Site lighting shall be provided to illuminate the pedestrian and service areas throughout the site as required by the City of Fort Lauderdale Building/Zoning code, utilizing concrete poles and Kim Lighting "The Archetype" fixtures.

#### Site Utilities

- Existing municipal water service lines shall be utilized to provide water service to the facility.
- A new fire line with double-detector check valve shall be installed to provide fire sprinkler protection to the facility
- A new 2" potable water line with backflow preventer shall be constructed to service the pool fill lines, hose-bibs, and deck showers.
- The building sanitary sewer systems & filter discharge piping shall be connected to the existing 8" service line located within the limits of construction.
- The building potable water systems, collector tank auto-fill lines and pool deck hose bibs shall be connected to existing 2" services located within the limits of construction.
- The existing gas service shall be utilized to provide hot water via conventional gas fired hot water heaters to the food prep areas, showers, locker rooms and janitor rooms, as required by code. Gas service is located within the limits of construction.

## FINA Competition Pool 53m x 25m x 2.5m in Depth

- FINA Competition Pool shall be constructed with Myrtha laminated stainless steel walls, standard membrane floor liner, and 14" deep Myrtha Ceramic II deck level gutter system.
- Competition Pool shall be constructed in accordance with Florida Building Code 424.1 requirements, Chapter 64E-9 Public Swimming Pool Code and FINA standards.
- Competition Pool shall have a minimum depth of 2.5m at the turn walls with required slope to the main drains.
- No Underwater lighting will be provided or installed.
- Grabrails with recessed steps will be provided for accessing the pool.
- Long course swim lanes (x10) shall be provided with Myrtha PVC marking material.
- Short course swim lanes (x20) shall be a provided with Myrtha PVC marking material. This shall be accomplished by repositioning the bulkheads.
- Two (2) fiberglass movable bulkheads by Nemato or approved equal.
- Anti-wave lane lines with storage reels shall be provided and shared for long and short course swim lanes. (twenty (20) sets of anti-wave lines total)
- Myrtha competition starting blocks (x20) shall be provided.
- Colorado Time System timing system with touch-pads (x20) and a multi-sport capability scoreboard shall be provided with an allowance of \$250,000.
- Pool piping systems shall be installed in schedule 40 PVC.

- Four (4) 48" FRP, 52.0 sq ft Nemato high rate sand filters.
- Two (2) 30hp ITT Marlow circulation pumps.
- Two (2) vinylester resin strainers with clear acrylic viewing windows and stainless steel baskets.
- Two (2) 30hp, Square D, Nema I, motor control centers with integrated variable frequency drives to operate circulation pumps and interface with chemical controller.
- One (1) BECSys7 chemical control center with ethernet and motor control center compatibility.
- A surge / collector tank below deck with tockable hatch and automated fill device will be provided.
- Eight (8) Symbiont PH215 Geo-Thermal units with titanium heat exchangers for heating and cooling pool water to FINA standards.
- ADA access shall be provided as per Department of Justice requirements.
- Two (2) portable lifeguard chairs shall be provided for use along the perimeter of the competition pool.
- Pool safety equipment and signage shall be furnished / installed as per Florida Building Code 424.1 and Chapter 64E-9 Public Swimming Pool Code.
- One (1) portable Hammerhead vacuum device to be shared for all bodies of water.

#### Renovation of the Existing 50m x 25y

- Renovation of the existing Competition Pool shall be constructed utilizing Myrtha "Renovaction System" laminated stainless steel walls, standard membrane floor liner and 8" deep Myrtha Ceramic II deck level gutter system.
- Renovation of the pool shall be constructed in accordance with Florida Building Code 424.1 requirements, Chapter 64E-9 Public Swimming Pool Code and USA Swimming standards.
- The renovated pool will have a minimum depth of 4'-0" at the turning wall and a maximum depth of 7'-0" at the starting end.
- Provide and install a 75'-0" x 45'-0" Vario Pool movable floor.
- No underwater lighting will be provided or installed.
- Grabrails with recessed steps shall be provided for access to the pool.
- Long course swim lanes (x8) shall be provided with Myrtha PVC marking material.
- Short course swim lanes (x23) shall be provided with Myrtha PVC marking material.
- Existing anti-wave lane lines for swim lanes shall be utilized.
- Eight (8) Spectrum "Bighorn" starting blocks and sixteen (16) deck anchors shall be installed. Starting blocks shall have the ability to be re-located as required to accommodate the long and short course training.
- Pool piping systems shall be installed in schedule 40 PVC.
- Four (4) 48" FRP, 52.0 sq ft Nemato high rate sand filters.
- Two (2) 30hp ITT Marlow circulation pumps.

- Two (2) vinylester resin strainers with clear acrylic viewing windows and stainless steel baskets.
- Two (2) 30hp, Square D, Nema I, motor control centers with integrated variable frequency drives to operate circulation pumps and interface with chemical controller.
- One (1) BECSys7 chemical control center with ethernet and motor control center compatibility.
- Pre-fabricated surge/collector tank with lockable hatch and automated fill device.
- Seven (7) Symbiont PH215 Geo-Thermal units with titanium heat exchangers for heating and cooling pool water.
- ADA access shall be provided as per Department of Justice requirements.
- Two (2) movable lifeguard chairs will be provided for use along the perimeter of the renovated pool.
- Pool safety equipment and signage shall be furnished / installed as per Florida Building Code 424.1.
- One (1) portable Hammerhead vacuum device to be shared for all bodies of water.
- Re-locate existing Colorado scoreboard and utilize existing touch pads and timing system.

#### New 25m x 25y FINA Dive Pool

- Dive Pool will be constructed with Myrtha System laminated stainless steel walls, standard membrane floor liner, and 14" deep Myrtha Ceramic II deck level gutter system.
- Dive Pool will be constructed in accordance with Florida Building Code 424.1 requirements and USA Diving standards.
- Dive Pool will have a maximum depth of 16' 4-7/8" (5m)
- Three (3) three meter springboards, Durafirm dive board assemblies shall be mounted to a concrete pedestal.
- One (1) one meter springboard, Durafirm dive board assembly shall be mounted to a concrete pedestal.
- Four (4) removable one meter springboards, Durafirm aluminum dive board assemblies with brass deck anchors shall be located on adjacent walls to tower.
- Three centerline dive tower with 3m, 5m, 7.5m and 10m platforms with non-slip surfacing shall be located on the South Dive Pool wall. The 3m, 5m, & 7.5m platforms to be 9'-10" (3m) wide and the 10m shall be 10'-6" (3.2m) for synchronized training and competition.
- Myrtha sparger / aeration system shall be installed at four (4) diving locations in the pool floor.
- 12V 300W in pool lighting will be provided and installed.
- Two (2) recessed stairways descending to 3'-0" water depth shall be located on opposite ends of the South Dive Pool wall.
- Ladders and/or recessed steps shall be provided for access to the pool.
- Eight (8) 25m and six (6) 25y swim lanes shall be provided with Myrtha PVC marking material.

- Nine (9) anti-wave lane lines shall be provided with storage reel to be shared for long and shorts courses.
- Eight (8) Spectrum "Bighorn" starting blocks and sixteen (16) deck anchors shall installed. Starting blocks shall have the ability to be re-located as required to accomplish long and short course training.
- Pool piping systems shall be installed in schedule 40 PVC.
- Four (4) 48" FRP, 52.0 sq ft Nemato high rate sand filters.
- Two (2) 30hp ITT Marlow circulation pumps.
- Two (2) vinylester resin strainers with clear acrylic viewing windows and stainless steel baskets.
- Two (2) 30hp, Square D, Nema I, motor control centers with integrated variable frequency drives to operate circulation pumps and interface with chemical controller.
- One (1) BECSys7 chemical control center with ethernet and motor control center compatibility.
- Deck level surge / collector tank with lockable hatch and automated fill device.
- Four (4) Symbiont PH215 Geo-Thermal units with titanium heat exchangers for heating and cooling pool water.
- ADA access shall be provided as per Department of Justice requirements.
- Two (2) movable lifeguard chairs shall be provided for use along the perimeter of the competition pool.
- Pool safety equipment and signage shall be furnished / installed as per Florida Building Code 424.1.
- 1200 square foot Dry Land Training area shall be provided on the pool deck in close proximity to the dive pool.
- Seven (7) person concrete acclimation pool with ceramic tile and quartz aggregate finish shall be provided adjacent to the dive pool.

#### Instructional Pool

- 20' x 40' Instructional Pool shall be constructed with Myrtha laminated stainless steel walls, standard membrane floor liner, and 8" deep Myrtha Ceramic II deck level gutter system.
- Instructional Pool shall be designed in accordance with Florida Building Code 424.1 requirements.
- Instructional pool will have a minimum depth of 1m at walls with required slope to the main drains.
- No Underwater lighting will be provided or installed.
- Ladders and/or recessed steps shall be provided for accessing the pool.
- Pool piping systems shall be installed in schedule 40 PVC.
- One (1) 34" FRP, 15.3.0 sq ft Nemato high rate sand filter.
- One (1) 7.5hp ITT Marlow circulation pump.
- One (1) vinylester resin strainers with clear acrylic viewing window and stainless steel basket.

- One (1) 7.5hp, Square D, Nema I, motor control center with integrated variable frequency drives to operate circulation pump and interface with chemical controller.
- One (1) BECSys7 chemical control center with Ethernet and motor control center compatibility.
- Deck level surge / collector tank with lockable hatch and automated fill device.
- Two (2) Symbiont PH090 Geo-Thermal units with titanium heat exchangers for heating and cooling pool water.
- ADA access shall be provided as per Department of Justice requirements.
- Pool safety equipment and signage shall be furnished / installed as per Florida Building Code 424.1.

#### Competition Pool Support Area and Bleachers

- A competition pool support area of approximately 8,200 square-feet shall be designed and constructed, adjacent to the competition pool, to include men's and women's locker rooms, men's and women's Restrooms, pool storage areas, swim team/coach office, meet management space, lifeguard/first-aid office and classroom space.
- Elevated aluminum stadium style bench seating for 2040 spectators will be constructed above the support space, overlooking the 53m competition pool.
- The spectator seating area shall include support spaces of approximately 2,025 square-feet, including men's and women's restrooms and concession space.
- Spectator seating area shall include dedicated accessible seating spaces in compliance with the American's with Disabilities Act.
- A shade fabric canopy shall be installed over the main spectator areas.

#### **Dive Pool Support Area and Bleachers**

- A dive pool support area shall be designed and constructed, adjacent to the dive pool, to include men's and women's restrooms, pool storage areas and dive team/coach office.
- Elevated aluminum stadium style bench seating for 540 spectators will be constructed above the support space, overlooking the dive pool.
- Spectator seating area shall include dedicated accessible seating spaces in compliance with the American's with Disabilities Act.

#### Lower-Level Seating Area

- A tiered, open-plan seating area shall be designed and constructed, adjacent to the renovated 50m pool, to accommodate approximately 240 spectators.
- This spectator area shall be serviced by restrooms and concessions located in the adjacent Bath-House facility

Pool Deck

- A combined total of approximately 60,000 square-feet of pool deck area shall be provided, not including the deck-level open area of the Bath-House Building.
- All pool deck areas to receive a broom finish.
- Column showers will be installed at multiple locations on the pool deck, within 20'-0" of the water's edge, as required by Florida Building Code Section 424.1.
- Deck drainage shall be provided

#### Pool Deck Lighting

- High mast sports lights will be provided to meet Health Department code for night use and FINA standards, 600 lux min., at starting platforms and turning ends, measured at the water surface.
- High mast sports lights will be provided to meet Health Department code for night use and FINA standards, 600 lux min. measured at 1 meter above water surface

#### Bath House Building

- A (4) level reinforced masonry building, will be designed and constructed to serve as the main entry and control point of the facility, having approximately 31,950 s.f. of enclosed space and 14,150 s.f. of open area.
- The Bath House Facility shall include a Men's and Women's Locker rooms, Men's and Women's restrooms, offices, Laundry, Lifeguard and First Aid Office, Concession Space, Conference Room, Fitness Center, Banquette Space, Lobby/Reception Areas, mechanical/electrical rooms, storage space and pool equipment room. The Bath House will also include an unfinished shell space for future Swimming Hall of Fame.
- Approximate square footages shall be:
  - o Ground Level- 10,150 s.f.
  - Second Level- 10,150 s.f. two-story unfinished shell space for future International Swimming Hall of Fame, potentially buildable to 20,300 s.f.
  - o Third Level- 1,500 s.f. enclosed, and 10,900 s.f. of outside deck area
  - o Fourth Level- 10,150 s.f. enclosed and 3,250 s.f. of balcony area
- Electrical, water, sewer and fire protection service shall be extended to the second level shell space for future Hall of Fame design and expansion/installation
- The structure shall be constructed with 8" reinforced concrete masonry units, on grade beams over auger-cast piles.
- Aluminum intake louvers with insect screens will be installed at exterior walls as required to provide proper ventilation. The louvers will meet Miami-Dade product approval.
- Hollow metal exterior doors/frames will be constructed of galvanized exterior grade material. The doors will meet Miami-Dade product approval.
- Glass storefront window and door systems will be constructed of high impact resistant tinted glass with anodized aluminum framing. The windows will meet Miami-Dade product approval.
- The floors in the Men's and Women's Locker room, and the Men's and Women's restrooms, will be finished with a seamless epoxy floor system.

- The floors in the offices, fitness center, Conference Room, Banquette Space, Lobby and Reception Areas, Concessions Space and Lifeguard/First Aid Office will be finished with a VCT and vinyl base floor system.
- The floors in the fitness center shall receive a rubberized flooring membrane.
- Floors in concession area and lifeguard/first-aid office shall receive a vinyl composition tile finish.
- The ceilings in the offices, fitness center, Conference Room, Banquette Space, Lobby/Reception Areas, Concessions Space and Lifeguard/First Aid Office will be finished with an acoustical tile in a ceiling grid.
- The ceilings in the Men's and Women's Locker Room and Men's and Women's restroom will be an exposed, painted, concrete finish.
- All other Interior ceilings will have an exposed and/or painted finish.
- The office will be provided with solid surfaced transaction counter with an observation window between the office and ticket area.
- Benches will be provided in the Men's and Women's Locker Rooms.
- Solid plastic (HDPE) 3-tiered padlocked lockers (12"Wx15"D), mounted on a 4" high base, will be installed along the interior walls of the Men's and Women's Locker Rooms (78 individual lockers total).
- Toilet and urinal floor mounted partitions will be constructed of 3/4" thick highdensity polyethylene plastic with natural anodized aluminum hardware and headrail bracing.
- Toilet accessories to be stainless steel as manufactured by Bobrick (or equal).
- The Concession area will be provided with floor drains and hot/cold water for the sink(s).
- The Men's and Women's locker rooms and restrooms will include toilets, urinals, sinks, hose-bibs, floor drains and showers as required by code.
- Domestic hot water piping will be provided to the locker room showers and concession area sinks.
- "Bi-Level" ADA compliant stainless steel electric chilled drinking fountains will be installed on the exterior of the building at the pool deck area.
- The locker rooms, pool-level restrooms, pool equipment room and concession area will be provided with mechanical ventilation.
- A HVAC System will be designed and installed to service offices, fitness center, Conference Room, Banquette Space, Lobby/Reception Areas, upper level restrooms and Lifeguard/First Aid Office..
- Power outlets and general lighting will be provided in quantities as required by the Florida Building Code of this use and occupancy, for all rooms planned throughout the facility.

#### Parking Structure

- A multi-level parking structure will be designed and constructed to include drive aisles, parking stops, striping, signage, ramps and traffic control devices, accommodating between 520 and 535 parking spaces.
- Final parking space count will be determined by final design, code and programming requirements.

• Dedicated Handicap Parking spaces shall be provided as required by code.

## Surface Parking

• Approximately 54 un-covered parking spaces will be designed and constructed to include drive aisles, parking stops, striping, signage and traffic control devices.

## Waterfront Park and Public Plaza

- A waterfront park and plaza will be constructed along the perimeter of the property and at the West end of the peninsula.
- Landscaping will be designed and provided.
- Irrigation system including automated controls will be provided for landscaped areas
- DC Alexander Park shall remain "As-Is." Any landscape areas disturbed by construction activities will be restored to original conditions.

## General Provisions

- The Design-Build Team will provide all civil, aquatic, architectural, structural, mechanical, plumbing and electrical plans necessary to permit and construct the proposed project.
- The Design-Build Team will provide construction stakeout as necessary to complete the proposed project.
- A boundary and topographic survey, reflecting elevations and topographic information on the NAVD, as well as underground utilities, easements and restrictions, shall be provided by The Owner.

## General Exclusions

- Removal of Unsuitable Soils and/or contaminated materials
- Impact Fees, Transit oriented fees and Trip charges
- Florida Power & Light Electrical Connection Fees
- Florida Power & Light relocation of existing primary and/or new primary services
- Utility Services (Telephone, Gas & Cable) Connection and primary service upgrades, including fees.
- Water and Sewer meters and connection Fees
- Water and Sewer main service modifications and upgrades
- Site Drainage and Site Water Management Control
- Repairs to the existing seawall and retaining walls.

## Assumptions and Clarifications

- Recreational Design & Construction assumes the role of Developer, Design/Build General Contractor and Pool sub-contractor as it pertains to this contract.
- RDC assumes that there are no environmental issues of concerns with the site.
- Permits are to be paid for with the allowance set forth in the GMP and will be treated as a reimbursable item.

- The project was designed with maintained average light levels of 600 lux over the pool's surface area and the first five feet of the pool deck. RDC assumes that this facility will not hold events requiring lighting levels in excess of those required by FINA, and that should these events take place, then temporary lights will be provided by the owner.
- The city will retain a Media Consultant to assist the design team through the city's desired media goals.
- RDC and its design team have been working on the behalf of the project for nearly three years and the effort to date is included in the total GMP. It is also understood that RDC and its design team are approximately 20% complete with the design effort upon acceptance of this contract.
- RDC has not made any provisions to accommodate the boat show; any additional expenses caused by or created by the boat show are not included in the GMP.
- A line item savings or expenses from the buy-out will be listed in the contingency log, and it's from there that the savings formula will be applied. Betterments can come from the contingency as long as it does not affect the portion of the savings factor that is RDC's.
- Preliminary design efforts required the design team to make certain assumptions regarding code interpretation. In particular, the design team has made assumptions regarding toilet fixture counts, and building use. Code interpretation by the authority having jurisdiction may alter those requirements. A breakdown of intended fixtures is included for reference
- Basis of GMP has included selection of pool equipment, and is included for reference.

		Pool Water Surface Area	Men's Urinais	Men's WC	Men's Lavatory	Women's WC	Women's levatory	Drinking Fountains
	Competition Pool	14,259	4	4	4	12	4	
	Dive Pool	6,150	2	2	2	6	2	
	Spa	70	1	1	1	· 1	1	1
	Training Pool	950	1	1 : ;	1	5	1	
	Existing Comp Pool	12,393	4	. 4		12	4	
Level 1 (Required FL								
	Training Pool	13,343	4	4	4	12	4	
	Existing Comp Pool	NING WAR	RA VISI	88 <b>1</b> 987	-989-984	SI NORM	RAAR SA	RAA.
Level 2 (Required FL Aquatic Code)			a	0	D	a	a	-
Lavel 2 (Dry Area)			·	ľ		<u>۲</u>	Ť	
Total - Level 2			0	0	o	D	٥	
Levél 3 Int(Required FL Aquatic Code)			0	0	D	0	0	
Level 3 (Dry Area) Total - Level 3			0	0	Ð	0	a	
Level 3 (Required FL Aquatic Code) Level 3 (Dry Area &	Competition Pool Dive Pool Spa	20,479	5	5	5	15	5	
2,540) (2004)			KS COL	1250133		NAMO NA	<b>资料</b> 销货	ROR S
Level 4 (Required FL Aquatic Code)			, o	a			a	
level 4 (Dry Area)			-			ľ	-	
Total - Level 4		L	0	0	0	0	0	

A provide a set of the set of t

Classification	Occupancy		Men's Urinals	Men's WC	• Men's Lavatory	Women's WC	Women's Levatory	Drinking Founteins
	1		1p	er 75	1 per 200	1 per 40	1 per 150	1 per 1,000
Assembly	A-S	Bleachers (Comp Pool)	9	l s	6	26	7	3
Assembly	A-5	Bleachers (Dive Pool)	2	2	2	7	2	1
			2536563		cieste de la	Section	ille Section	\$55. A.S.
	1	4th ficer -	19	er 75	1 per 200	1 per 75	1 per 200	1 per 500
Assembly	A-2	Ath Floor - BangueVFitness	1	1	1	2	t	1
		1 per 25 - 1st 50		1 per 40 - 1st 80	1 per 25 - 151 \$0	1 per 40 - 1st 80	1 per 100	
Business B	В	4th Floor - Admin	0	1	1	1	1	1
	Τ.		1 1 2	er 125	1 per 200	1 per 65	1 per 200	1 per 500
Assembly	A-3	2nd-3rd Floor -15HOF	2	i	2	6	2	2
	1		1 per 2	5 - 1st 50	1 per 40 - 1st 80	1 për 25 - 1st 50	1 per 40 - 1st 80	1 per 100
Business	8	1st Floor - Office	0	1	1	1	1	1
			1 ре	er 500	1 per 750	1 per 500	1 per 750	1 per 1,00
Mercantile	м	1st Floor - Gift Shop	0	1	1	1	1	1
	1		1.6	er 75	1 per 200	1 per 40	1 per 150	1 per 1,00
Assembly	A-4	1st Floor - Therapy Pool	1	1	1	4	1	1
			1 р	er 100	1 per 100	1 per 100	1 per 100	1 per 1,00
Storege	5-2	1st Floor	1	L	L	l	L	
	<del></del>	- <u>r</u>	10	er 100	1 per 100	1 per 100	1 per 100	1 per 1,00

1

## FINA Competition Pool

## 53m x 25m / 14,292sq ft / 876,942gal

- 2 Filters / National 48", 52sq ft / NFS-48-50B-520S
- 2 Filters / National 48", 52sq ft / NFS-48-50B-520E
- 4 8" Mini-Links Single Lever B/W Valve Auto / NFS-60-MINP-0800A
- 4 60" Mini-Link Hardware Kit 50B / 575-688ML
- 1 Gauge Panel / Auto System / NFS-GP-EFT-ASYS
- 1 12" Primary Valve / Nemato / NFS-12-PV-1200
- 1 1.5" Backwash Site Glass / SGS-150
- 904 Filter Media
- 1 BECSys7 Chemical Controller / BECSYS7-OS1TPXO
- 1 Vacuum Transducer w/50' Cable / 8660034
- 1 Pressure Transducer w/50' Cable / 8660034
- 1 Signet 8" PVC Saddle w/adaptor / PV8S080
- 1 Paddlewheel Lo Flow Sensor (5-8") Blue Cap / 3-2536-P1
- 2 Pumps / Marlow 30hp, 1150rpm, 230/460v, 3ph / 6-13SC 30HP 1150
- 2 Strainer / Nemato 12" x 8" / FRP 30psi / NSS-16-030-1208
- 2 MCC-VFD / Square D, 30hp, 460v, Nema 1 / EF-30-1-460
- 2 Stainless Steel Flex Connector
- 2 Rubber Flex Connector
- 2 Butterfly Valves
- 2 Concentric Reducer
- 1 High Capacity Sodium Hypochlorite Feeder / Solution X-1 / 22152-01
- 1 Muriatic Acid Feeder / Solution X-2 / 22152-02
- 1 Sodium Hypochlorite Storage Tank / Sharred
- 1 Muriatic Acid Storage Tank / Sharred
- 1 Symbiont Geo-Thermal Heat/Cool System
- 8 PH215 Heat/Cool Units
- 1 8" Supply well
- 1 10" Injection Well
- 1 20hp Submersable Well Pump
- 2 Fiberglass Bulkheads / Nemato or Equal
- 1 Timing System / Colarodo Timing Systems
- 20 Touch Pads / Colarodo Timing Systems
- 2 Guard Chairs / KDI Paragon 6'
- 2 Life Rings with throw rope
- 2 Life Hooks & Poles
- 2 Life Ring Holder
- 1 Vac Head pole hose
- 1 Test Kit
- 1 Pool Rules Sign
- 8 Grab Rails for recessed steps / Myrtha
- 12 Brass Anchors
- 12 Backstroke Poles
- 8 Starting Blocks / Myrtha OSB12
- 16 Starting Block Anchors / Myrtha
- 9 Lane Lines
- 2 Lane Line Storage Real

Page 12 of 27 Exhibit 1 CAM 15-1467 Page 142 of 158

## **FINA Dive Pool**

#### 25m x 25y / 6,150sq ft / 758,400gal

- 2 Filters / National 48", 48sq ft / NFS-48-50B-480S
- 2 Filters / National 48", 48sq ft / NFS-48-50B-480E
- 4 8" Mini-Links Single Lever B/W Valve Auto / NFS-60-MINP-0800A
- 4 60" Mini-Link Hardware Kit 50B / 575-688ML
- 1 Gauge Panel / Auto System / NFS-GP-EFT-ASYS
- 1 12" Primary Valve / Nemato / NFS-12-PV-1200
- 1 1.5" Backwash Site Glass / SGS-150

840 Filter Media

- 1 BECSys7 Chemical Controller / BECSYS7-OS1TPXO
- 1 Vacuum Transducer w/50' Cable / 8660034
- 1 Pressure Transducer w/50' Cable / 8660034
- 1 Signet 8" PVC Saddle w/adaptor /PV8S080
- 1 Paddlewheel Lo Flow Sensor (5-8") Blue Cap / 3-2536-P1
- 2 Pumps / Marlow 30hp, 1150rpm, 230/460v, 3ph / 6-13SC 30HP 1150
- 2 Strainer / Nemato 10" x 8" / FRP 30psi / NSS-16-030-1008
- 2 MCC-VFD / Square D, 30hp, 460v, Nema 1 / EF-30-1-460
- 2 Stainless Steel Flex Connector
- 2 Rubber Flex Connector
- 2 Butterfly Valves
- 2 Concentric Reducer
- 1 High Capacity Sodium Hypochlorite Feeder / Solution X-1 / 22152-01
- 1 Muriatic Acid Feeder / Solution X-2 / 22152-02
- 1 Sodium Hypochlorite Storage Tank / Sharred
- 1 Muriatic Acid Storage Tank / Sharred
- 1 Symbiont Geo-Thermal Heat/Cool System
- 4 PH215 Heat/Cool Units
- 1 8" Supply Well
- 1 10" Injection Well
- 1 7.5hp Submersable Weil Pump
- 2 Guard Chairs / KDI Paragon 6'
- 4 Durafirm Short Stands / 70-231-520
- 4 Anchor Sets (6 each) 70-231-900
- 4 Durafirm 1m Stands 70-231-400
- 4 Anchor Sets (8 each) 70-231-900
- 8 Maxiflex Model B Dive Boards / 66-231-330
- 2 Life Rings with throw rope
- 2 Life Hooks & Poles
- 2 Life Ring Holder
- 1 Test Kit
- 1 Pool Rules Sign
- 2 Handrails
- 4 Ladders
- 8 Starting Blocks
- 16 Starting Block Anchors
- 9 Lane Lines
- 2 Lane Line Storage Real

## **Renovation of Existing (lower level) Pool**

#### 50m x 25y / 12,326sq ft / 786,582gal

- 2 Filters / National 48", 48sq ft / NFS-48-50B-480S
- 2 Filters / National 48", 48sq ft / NFS-48-50B-480E
- 4 6" Mini-Links Single Lever B/W Valve Auto / NFS-60-MINP-0800A
- 4 60" Mini-Link Hardware Kit 50B / 575-688ML
- 1 Gauge Panel / Auto System / NFS-GP-EFT-ASYS
- 1 12" Primary Valve / Nemato / NFS-12-PV-1200
- 1 1.5" Backwash Site Glass / SGS-150
- 840 Filter Media
- 1 BECSys7 Chemical Controller / BECSYS7-OS1TPXO
- 1 Vacuum Transducer w/50' Cable / 8660034
- 2 Pressure Transducer w/50' Cable / 8660034
- 1 Signet 10" PVC Saddle w/adaptor /PV8S100
- 1 Paddlewheel Lo Flow Sensor (5-8") Blue Cap / 3-2536-P2
- 2 Pumps / Marlow 30hp, 1150rpm, 230/460v, 3ph / 6-13SC 30HP 1150
- 2 Strainer / Nemato 10" x 6" / FRP 30psi / NSS-16-030-1006
- 2 MCC-VFD / Square D, 30hp, 460v, Nema 1 / EF-30-1-460
- 2 Stainless Steel Flex Connector
- 2 Rubber Flex Connector
- 2 Butterfly Valves
- 2 Concentric Reducer
- 1 High Capacity Sodium Hypochlorite Feeder / Solution X-1 / 22152-01
- 1 Muriatic Acid Feeder / Solution X-2 / 22152-02
- 1 Sodium Hypochlorite Storage Tank / Sharred
- Muriatic Acid Storage Tank / Sharred
- 1 Symbiont Geo-Thermal Heat/Cool System
- 7 PH215 Heat/Cool Units
- 1 8" Supply Well
- 1 10" Injection Well
- 1 20hp Submersable Well Pump
- 1 75' x 45' Movable Floor / Vario Pools
- 2 Guard Chairs / KDI Paragon 6'
- 2 Life Rings with throw rope
- 2 Life Hooks & Poles
- 2 Life Ring Holder
- 1 Pool Rules Sign
- 8 Ladders
- 28 Brass Anchors
- 12 Backstroke Poles
- 8 Starting Blocks
- 16 Starting Block Anchors
- 9 Lane Lines
- 2 Lane Line Storage Real
### Instructional Pool

### 20' x 40 / 800sq ft / 19,900gal

- 1 Filters / National 34", 15.3sq ft / NFS-34-50B-153E
- 1 4" Mini-Links Single Lever B/W Valve Auto / NFS-34-SFUP-0400A
- 1 34" Mini-Link Hardware Kit 50B / 575-343SF
- 1 Gauge Panel / Auto System / NFS-GP-EFT-ASYS
- 1 1.5" Backwash Site Glass / SGS-150
- 48 Filter Media
- 1 BECSys7 Chemical Controller / BECSYS7-OS1TPXO
- 1 BECSys Solid Gold Band ORP Snsor Cable / 9660038
- 1 Vacuum Transducer w/50' Cable / 8660034
- 1 Pressure Transducer w/50' Cable / 8660030
- 1 Signet 3" PVC Saddle w/adaptor /PV8S030
- 1 Paddlewheel Lo Flow Sensor (5-8") Blue Cap / 3-2536-PO
- 1 Pumps / Marlow 7.5hp, 1750rpm, 230/460v, 3ph / 6-13SC 30HP 1150
- 1 Strainer / Nemato 4" x 3" / FRP 30psi / NSS-06-030-0403
- 1 MCC-VFD / Square D, 7.6hp, 460v, Nema 1 / EF-7.5-1-460
- 1 Stainless Steel Flex Connector
- 1 Rubber Flex Connector
- 1 Butterfly Valve
- 1 Concentric Reducer
- 1 Sodium Hypochlorite Feeder / Stenner 85 GPD / 85MJL5A1S
- 1 Muriatic Acid Feeder / Stenner 85 GPD / 85MJL5A1S
- 1 Sodium Hypochlorite Storage Tank / Sharred
- 1 Muriatic Acid Storage Tank / Sharred
- 1 Symbiont Geo-Thermal Heat/Cool System
- 2 PH090 Heat/Cool Units
- 1 4" Supply well
- 1 1.5hp Submersable Well Pump
- 2 Life Rings with throw rope
- 2 Life Hooks & Poles
- 2 Life Ring Holder
- 1 Test Kit
- 1 Pool Rules Sign
- 2 Handrails
- 2 Ladders
- 8 Brass Anchors

		Developer : Recreational Design & Construc Owner : City of Fort Lauderdale	tion	I	FORT	LAUDE	RDALE	AQUA'	TIC CE	NTER Sheel1 of 1		
		B/	AS/S	OF G	MP 3.1	0.3.4: A	QUACT	IC FAC	LITY			
A TEM NO. '	<b>N</b> Q	B. ITEM DESCRIPTION	QTY	UNIT	Li 1. Unit Cost	ABOR 2. Total	F. MA	PERIAL 2. Total	EQUIPMENT	SUB CONSULTANTS	TOTAL LINE ITEM PRICE (E.2 + F.2)	GRAND TOTAL
		Design Fees				(E.1 X C)		(F.1 X C)				
2 0	222	Master Planning Architectural Design Aquatic Design	1 1	Ls Ls						\$ 85,000 \$ 295,000 \$ 325,000	\$ 326,000,00	
5 0	N 1 N 1 N 1	Structural Design Chvil Engineer Machanical Engineer	1	L0 L5 Ly						\$ 130,000 \$ 145,000 \$ 174,600	\$ 130,000.00 \$ 145,000.00 \$ 174,500.00	
7 0	2₩1 2₩1	Electrical Engineer Environmentel Engineer Landacape		- Li - Li						\$ 65,000 \$ 20,000 \$ 35,000	\$ 85,000.00 \$ 20,000.00	
	51¥1 51⊻1	Infontion Engineering Traffic Engineering	1	Ls Ls						\$ 30,000 \$ 20,000	\$ 30,600.00 \$ 20,000.00	
12 [	ועג זעג זעז	Pre Construction Director Project Manager Assistant Pre Con	48 560 60	ուրջ	\$ 1,960.00 \$ 53.76 \$ 1,000.00	\$ 30,100.00					8 94,080.00 8 30,100.00 1 80,000.00	
14 E		Sprvays and Boundaries Reconstruct Geo Technical Testing Asbestos Testing and Reports	1	5 5 5						\$ 18,000 \$ 15,000 \$ 15,000	\$ 18,000.00 \$ 15,000.00	
17	1 12	Print and Copies Pastage and Couries	10	miha			\$ 1,200 \$ 700	\$ 12,000.00 \$ 7,000,00			12,000.00 7,000.00	
		AQUATIC FASTLE CHBIGN LINE TOTAL										1,878,690.00
		General Conditions										
20 0	01V1 01V1	Surialnable Building Reviews Submittal Reviewa/ Gonakycybon Admin Threshold kupector	1 1800	Ls mhs	\$ \$5.00					45,000 4 138,000 5 55,000		
22 0	1 VIC 0 VI	Quality Assurance and Control Sonedvling	520 18 1700	hrs ea	\$ 85.00 \$ 55.00	\$ 44,200.00 \$ \$ 03,600.00				25,000	\$ 44,200.00	
25 28 0	<u>) № 1</u>	EaUmating and Cost Control Sector Project Manager Sector Project Manager	n	Wks	\$ 2,150.00	\$ 165,550.00					\$ 168,680.00 \$ 127,050,00	
28 0	0N1 0N1 0N1	Assistant Project Manager Field Engineer Safety / Fire Watch Officer	77 71 710	Wxs HRS	\$ 65.00	\$ 115,600.00 \$ 50,050.00					\$ 115,500.00 \$ 50,080.00	
31 0	2×1 2×1	General Super Intendent Pool Foreman Butiding Foreman	77 48 77	Wks Wks	3 1,600,00 3 1,600,00 3 1,600,00	\$ 123,200.00 \$ 72,000.00 \$ 115,800.00 \$ 82,500.00					123,200,00 72,000,00 115,500,00	
33 (	5 <b>1</b> √1	Bookkeeping & Accounting Construction Experies and Surveys	1600	Hrs Ls	\$ 05.00	\$ 62,500.00		· .		\$ 45,000	\$ 82,500,00 \$ 45,000.00	
35 E	DIV 1 DIV 1	Concrete Teating Compaction Teating	160 20	<b>\$</b> 8						\$ 50 \$ 105	\$ 7,500.00 \$ 2,100.00	
	<u>2N1</u>	Weiding and Structural Testing	16	Le mths			\$ 225			\$ 6,000	\$ 8,000.00 \$ 4,050.00	
39 0	2N/1 2N/1	Mobilization and Salup Staging Area	120	LS	\$ 25.00	\$ 3,000.00	\$ 2,000				\$ 3,000.00 \$ 2,000.00	
41 0		Office Trails Set up Project Signage	2	L3 #2			\$ 1,700 \$ 600				\$ 1,700.00 \$ 1,000.00	
42 0		Construction Signape	1 18				\$ 2,000 \$ 300				\$ 2,000.00 \$ 8,450.50	
45 E	DIN T	Sub Contractor Filed Office Field Office Furnishings Storage Containers	17 1 15	Le Mths			\$ 225 \$ 3,600 \$ 450				\$ 3,825.00 \$ 3,800.00 \$ 6,750.00	
48 [	DIV 1 DIV 1 DIV 1	Tellet Facilities Temporary Fencing Barricedes	18 1 18	Mths Le mths			\$ 260 \$ 12,600 \$ 300				\$ 4,680.00 \$ 12,500.00 \$ 5,400.00	
81 E	DIV 1 DIV 1	Temporary Water Connection Temporary Electric		ել			\$ 2,000 \$ 5,000				\$ 2,000.00 \$ 6,000.00	1
52 C 53 C 54 C	DIV 1 DIV 1 DIV 1	Fire Department Connection Telephone Connection Water Monthly	16				\$ 7,600 \$ 500 \$ 500				\$ 7,600.00 \$ 500.00 \$ 9,000.00	
55 0	DIV 1 DIV 1	Electrical Monthly Phone Mardily	10	Mihs	<u> </u>		5 400 5 300				\$ 7,200.00 \$ 5,400.00	1
		Dumpsters Trash Chyles	10		[		3 400 3 400				\$ 25,000.00 \$ 4,000.00	
80 C		Temporary Lebor Daily Clean Up	2080	mhs mhs	\$ 15.00 \$ 15.00 \$ 6,500.00	\$ 31,200.00 \$ 10,800.00					\$ \$1,200.00 \$ 10,800.00 \$ 5,600.00	
62 1		観察はかかい かいじゅう たいとう やがらす					\$ <u>840</u> \$ 400		<b></b>		\$ 13,440,00 \$ 0,400.00	
84 C		Water , ice { Workman Daily} Bafaty Materiale		Mth Mth			\$ 600 \$ 450 \$ 1,200				\$ 14,400.00 \$ 7,660.00 \$ 20,400.00	
67 6		Security Watchman	7(	Wks	<u>\$ 1,200.00</u> \$ 400.00	· · ··	* 1,200				\$ <u>26,400.00</u> \$ <u>84,000.00</u> \$ <u>28,000.00</u>	
69 1	ow €	All the Mileson and Travel	16	MDh	a 400.00		1 1 200				\$ 21,600.00	
	DIV 1	RDC Trucks Equipment Fuel	16 16		<b> </b>		\$ 600 \$ 670				\$ 10,800.00 \$ 12,080.00	

Γ

	72		Puttch List	4	wks	\$ 1,750.00 }	\$ 7,000.00					5 7,003.00	
PA         Distance         PA         Distance         PA	74	DJV 1	As Buills	1	L\$						\$4,000	4,009.00	
South Tube         South Local	76	DIV 1	Printing and Binding, Turn Over Docs		Ls			\$ 1,200				\$ 1,200.00	
Image: constraint of the second sec	<u> </u>			'			- A.L.	• 000				\$ <u>500.00</u>	
Image: second			<b>G</b> IN FAUX		3. 1. 5	21. M.		的现在	金利用	928.59	网络合同语言	的形式动物	
To         Dot         Dot <thdot< th="">         Dot         <thdot< th=""> <thdot< th=""> <thdot< th=""></thdot<></thdot<></thdot<></thdot<>			CONDITIONS TOTAL COST	<u>e</u>	1. 11	32,038	9 - 18 - 18 - 18 - 18 - 18 - 18 - 18 - 1	\$4 <b>5.4</b> 3.43	-1348/678)	8-5-1 <b>8</b> -3	纪尔的家畜	85.867.51 <del>9.</del> 752.	1,717,808.00
To         Dot         Dot <thdot< th="">         Dot         <thdot< th=""> <thdot< th=""> <thdot< th=""></thdot<></thdot<></thdot<></thdot<>													1
1         100		<b>D</b> N 2									E 45.000	45.000.00	
1       1	78	DIV 2	Demplition	1	L1 -						\$ 335,130	\$ 335,430.00	
Col. 1001         Direction of sections         Li. 1         Col. 1002         Direction of sections           Col. 2001         Control Contro		DIV 2	Drainage	1							\$ 262,100	\$ 252,100.00	
10         100         Constrained out of constr			Bewer Piping and Connections	1									
Color         Color <th< th=""><th>83</th><th><b>0</b>₩ 2</th><th>Concrete Welkwelke</th><th>1</th><th>t</th><th></th><th></th><th></th><th></th><th></th><th>\$ 32,960</th><th>\$ 32,960.00</th><th></th></th<>	83	<b>0</b> ₩ 2	Concrete Welkwelke	1	t						\$ 32,960	\$ 32,960.00	
Cold         Cold <thcold< th="">         Cold         Cold         <thc< th=""><th></th><th></th><th>Asphalt paving , Curbs and Marking</th><th></th><th></th><th></th><th></th><th><b></b></th><th></th><th></th><th>3 09,405 3 158,760</th><th>\$ 189,750.00</th><th></th></thc<></thcold<>			Asphalt paving , Curbs and Marking					<b></b>			3 09,405 3 158,760	\$ 189,750.00	
Bit         Dir         Software         Softw			Lendscaping and Trees, Planters	1							\$ <u>144,425</u> \$ <u>95,000</u>	\$ 144,426.00 \$ 98,000,00	
IDTAL STE WORK TOTAL 20201.         IDEA         IDEA <t< th=""><th></th><th></th><th>Electrical Primary Service and Power Vaults</th><th></th><th>الو</th><th></th><th></th><th></th><th></th><th></th><th>\$ 475,000</th><th>\$ 475,000.00</th><th></th></t<>			Electrical Primary Service and Power Vaults		الو						\$ 475,000	\$ 475,000.00	
SO Metar Compatibility Pool         SO         SO         SO         SO         SO           8         CV3         Section prime         C         SO	-		Sive Electrical Distribution										
10     2010     And constrained and allow     100     1     2010     1000000       10     10000     1000000     1000000     10000000     10000000       10     10000     1000000     10000000     10000000       10     1000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     100000000       10     10     10000000     100000000       10     10     <			TOTAL SITE WORK TOTAL COSTS	Seal of	100	1. 20	19986	S. B. Cak	Cost of	13.38 694	<b>新新派派</b>	e na de la compañía d	2,472,830.00
10     2010     And constrained and allow     100     1     2010     1000000       10     10000     1000000     1000000     10000000     10000000       10     10000     1000000     10000000     10000000       10     1000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10000000     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     10000000       10     10     10000000     100000000       10     10     10000000     100000000       10     10     <			50 Mates Campatilian Real										
Bit Bit S         Calification and the factor of the f	- 89	DIV 13	Pool Concrete Floor Stab	1200							\$ 330	\$ \$98,000.00	
The Difference of Reng.     Li     Monthly Constrained Provided Action of Constrained Pro	90	DIV 13	Pool Slab Finish	12870	\$F			\$ 25.700			0.45	\$ 4,600.00	
Image: State 1         Image:	92	DIV 13	Pool Filtration and Pumps	1	Ls			\$ 185,750				\$ 165,750.00	
B         Dots         Apple Alament         I         Processor	94	DIN 13	Install Mytha Panela	<u> </u>	Ls			\$ 165,000				\$ 165,000.00	
27       0001       10001       10000       10000       10000       100000       100000       100000       100000       100000       100000       100000       100000       100000       100000       100000       100000       100000       100000       100000       100000       1000000       1000000       1000000       1000000       1000000       1000000       1000000       1000000       1000000       10000000       10000000       10000000       10000000       10000000       1000000000000000000000000000000000000	93	DIV 13	Pool Bulkheads	1				\$ 300,000				\$ 300,000.00	
EX         DVID         The Neuroid stars are stars         0         1         0         78000000         78000000         78000000         78000000         78000000         78000000         78000000         78000000         780000000         780000000         780000000         780000000000         7800000000	97	DIV 13	Aquatic Lift	1	L9			\$ 10,200				\$ 10,200.00	
100       1000       Peak I face the final final and the final set of the set of se	99	DIV 13	Geo Thermal Heaters		Ls				·····			\$ 290,000.00	
100       DOTO       Fight Constraint ()       0       1       10000       1000000         100       Part Bendingsharp       1       0       0       100000       1000000         100       DOTO Bendingsharp       1000000       10000000       10000000       10000000         100       DOTO Bendingsharp       000000       10000000       10000000       1000000         100       DOTO Bendingsharp       000000       1000000       1000000       1000000         100       DOTO Bendingsharp <td< td=""><td></td><td></td><td>Pool Vile and Rece lane finishes</td><td>1</td><td></td><td><u> </u></td><td>\$ 90 250 00</td><td></td><td></td><td></td><td></td><td>\$ 40,690.00</td><td></td></td<>			Pool Vile and Rece lane finishes	1		<u> </u>	\$ 90 250 00					\$ 40,690.00	
101         0170         First Breinfall stamp         101         101         101         600         100         600         100         600         100	102	DIV 13	Pool Equipment Installation	1	La		\$ 25,850.00			[		\$ 25,850.00	
Image: state			Pool Scoreboard (Allowance) Pool Elactrical Bonding	- 1				<u>\$176,000</u>			§6,000	\$ 175,000.00 \$ 5,000.00	
Image: state in the s	105		WAN (IN LINE (-I- WATAVA) PATAVA I POPUAGA - AND	- 		A	NOTES A DESCRIPTION	19094 193/1984	128920 2550	di taka kata sa			
General Control Prior Base 200 (2)	<u> </u>		2 YAND SILLOOKIN SHOWAL 24		<u>. 8</u>	08342232	8. 1827 3. Ash	DYFR WARKS	ALC: CONTRACTOR	6.200826	Contraction MD	9-9738-32, 27683	2.52
General Control Prior Base 200 (2)			Dhia Wall										
Control       Peak Big France       Peak Big	108	DIV 13		1034	CY						\$ 330	\$ 341,220,00	
148       01103       Port Taunto sel futura       1       1       10000       1       10000       100000       1000000       10000000       1000000000000000000000000000000000000	107	DIV 13	Poel Step Finish	6725	8F			¢ 20.000				\$ 2,176.00	
111       BY 10       Div B tradi       1       1       2550       1       5.6607         111       BY 10       Brad Bart       1 </td <td>109</td> <td>DIV 13</td> <td>Pool Filtration and Pumps</td> <td></td> <td>La</td> <td></td> <td></td> <td>\$ 150,000</td> <td></td> <td></td> <td></td> <td>\$ 150,000.00</td> <td></td>	109	DIV 13	Pool Filtration and Pumps		La			\$ 150,000				\$ 150,000.00	
110       (V13)       Starting Block       1       1       4       2000       1       100000         110       (V13)       Starting Block       1       1       100000       1       100000         110       (V13)       Starting Block       1       1       100000       1       100000         110       (V13)       Starting Block       1       1       100000       1       100000         111       (V13)       Pool Tiss and Real in Marine       1       1       100000       1       100000         111       (V13)       Pool Tiss and Real in Marine       1       1       1       100000       1       100000         111       (V13)       Pool Tiss and Real Alexan in Marine       1       1       1       100000       1       100000         111       (V13)       Pool Tiss and Real Alexan in Marine       1       1       1       1       100000       1       100000       1       100000       1       100000       1       100000       1       100000       1       100000       1       100000       1       100000       1       100000       1       100000       1       100000       1       10				1									
116       CV 10       Life Gurd Ghars       Li       I <td>112</td> <td>DIV 13</td> <td>Starting Blocks</td> <td>1</td> <td>La</td> <td></td> <td></td> <td>\$ 30,000</td> <td></td> <td></td> <td></td> <td>\$ 30,000.00</td> <td></td>	112	DIV 13	Starting Blocks	1	La			\$ 30,000				\$ 30,000.00	
117       10/10       Feed Previous       1       1       0       1000000         118       0/10       Peed Previous       1       1       1000000       1       1000000         110       0/10       Peed Previous       1       1       1000000       1       1000000         110       0/10       Peed Previous       1       1       1       1000000       1       1000000         110       0/10       Peed Previous       1       1       1       1000000       1       1000000       1       10000000         111       111       1       1       1       1       1       1000000       1       1       1       10000000       1       1000000000000000000000000000000000000	115	OIV 13	Life Guard Chairs	1	1.6			6,780				\$ 6,780.00	1
If if (i) (1) Pool Pool Pool Pool Pool Pool Pool Poo				. 1				\$ 150,000		<u> </u>	<b>S</b> 31,600		1 1
170       DV: 10       Peod Scarebard (Alwaner)       1       Ls       3       7.000       1       7.0000       1       7.0000       <	118	DIV 13	Pool Flumbing and Piping	1						ļ		\$ 60,000.00	1 1
DUVE WELLS/IDIALCOSI	120	DIV 13	Pool Scoreboard (Allowance)		Ls		20,012.01	\$ 75,000		<b></b>	\$ 75,000	1 76,009.00	1
Instructional Pool         Pool Converte Fisce Binh         Pool Converte	12			1					l		\$ 4 <u>000</u>		
Part Dov/18       Part Converter Fried Hist       Part Converter Fried			A REAL PROPERTY AND A REAL	1.04	A. 89	3.55293	S 20 10 12	网络马马克	N. Secto	1780 B.	國家的民族政治		1,656,948.84
Part Dov/18       Part Converter Fried Hist       Part Converter Fried			instructional Pool							╂		· · · ·	
131       DV 13       Peol Cellector Burge Taxic       1, 50       1, 500       1, 500       1, 500       1, 500       1, 500, 500         132       DV 13       Peol Taxics and function       1, 51       3, 120,000       1, 640,000       1, 640,000       1, 640,000         132       DV 13       Martie Life       1, 51       3, 120,000       1, 640,000       1, 640,000       1, 640,000         132       DV 13       Acuette Life       1, 51       3, 100,000       1, 640,000,000       1, 640,000,000       1, 640,000,000       1, 640,000,000       1, 640,000,000       1, 640,000,0		DIV 13	Pool Concrete Floor Slab									\$ 30,580.00	1
128       DV/13       Pool Flatten and Funges       1 <t< td=""><td>124</td><td>DIV 13</td><td>Pool Collector Surge Tank</td><td></td><td></td><td></td><td></td><td>\$ 11,300</td><td></td><td><u>}</u></td><td>0.45</td><td>\$ \$1,300.00</td><td></td></t<>	124	DIV 13	Pool Collector Surge Tank					\$ 11,300		<u>}</u>	0.45	\$ \$1,300.00	
Span         Span <th< td=""><td>128</td><td>DIV 13</td><td>Pool Filtration and Pumps</td><td>1</td><td>Ls</td><td>  </td><td></td><td>\$ 165,500</td><td></td><td></td><td></td><td>\$ 165,500.00</td><td>l l</td></th<>	128	DIV 13	Pool Filtration and Pumps	1	Ls			\$ 165,500				\$ 165,500.00	l l
Stat         Stat <th< th=""><th>129</th><th>DIV 13</th><th>Aquatic Lift</th><th></th><th>1.5</th><th></th><th></th><th>\$ 8,400</th><th></th><th></th><th></th><th>\$ 6,400.00</th><th></th></th<>	129	DIV 13	Aquatic Lift		1.5			\$ 8,400				\$ 6,400.00	
132         DIV 13         Peol Flexibles and Flexa         1         4         9         43.52.00           133         DIV 13         Peol Ecolorest Includes on It is         \$         18.600.00         \$         14.500.00           134         DIV 13         Peol Ecolorest Includes on It is         \$         18.600.00         \$         14.500.00           134         DIV 13         Peol Ecolorest Includes on It is         \$         18.600.00         \$         14.500.00           Internet Includes on It is         \$         \$         18.600.00         \$         18.600.00           Internet Includes on It is         \$         \$         105.000.00         \$	131	DIV 13	Pool Tile and Race lane finishes	1	Ls	<b>.</b>		<ul> <li>105,500</li> </ul>		<u></u>	105,500.00 \$ 6,000	\$ 6,000.00	1 1
134         DV 10         Pool Electrical Bonding         Ls         S         2,000, S <th2,000, s<="" th=""> <th2,000, s<="" th="">         2,0</th2,000,></th2,000,>	132	DIV 13	Pool Plumbing and Fiping	1						{		\$ 48,352.00	1
Image: Second and sec	134	DIV 13	Peel Electrical Bonding							į	\$ 2,000		
Image: Second and sec				復建	38.15.1	456-453C-1	19. A.	2.849.8*A6	Same Cool	X (KAREA)		NAME AND	STREET & A MARKED SHOP
Profit         10%           135         DiV 13         Spa Complete         1         Ls         \$ 165,000 \$ 168,000.00           135         DiV 13         Spa Complete         1         Ls         \$ 165,000 \$ 168,000.00           136         DiV 13         Pool 3(6) Findsh         2000 \$ 100,000         \$ 60,007         \$ 60,007           138         DiV 13         Pool 3(6) Findsh         2000 \$ 57         \$ 60,007         \$ 60,007           138         DiV 13         Pool 3(6) Findsh         2000 \$ 57         \$ 60,007         \$ 60,007           138         DiV 13         Pool 2(6) Findsh         2000 \$ 57         \$ 60,007         \$ 60,007           140         DiV 13         Pool 2(6) Findsh         2000 \$ 57         \$ 155,000         \$ 156,000,000           141         DiV 13         Pool 2(6) Findsh         1 Ls         \$ 165,000         \$ 165,000,000           142         DiV 13         Pool 2(6) Findsh         1 Ls         \$ 165,000         \$ 165,000,000           143         DiV 13         Pool 2(6) Findsh         1 Ls         \$ 260,000         \$ 165,000,000           144         DiV 13         Pool 2(6) Findsh         1 Ls         \$ 260,000         \$ 165,000,000           144 </td <td>1</td> <td></td> <td>COST</td> <td></td> <td>國際</td> <td>1332 I</td> <td></td> <td>(保護)為</td> <td></td> <td>1000</td> <td>1268680</td> <td></td> <td></td>	1		COST		國際	1332 I		(保護)為		1000	1268680		
135         D/V 13         Spa Complete         1         Ls         \$         165,000,00           135         D/V 13         Spa Complete         1         Ls         \$         165,000,00           State of the				anatan -	TANGS.)	a.v	IN NO. WING AS	Profit	10%	1995 B. B. S.		ARTING A STATE	CHARGE AND
REFURBISH 60 Meter Pool         3           138         01/13         1         1         5         60/27         5         60/27         5         60/27         5         60/27         5         60/27         5         60/27         5         60/27         60/27         5         60/27         60/27         5         60/27         5         60/27         5         60/27         60/27         5         60/27         6         1         60/27         5         60/27         5         60/27         6         1         60/27         5         60/27         6         1         60/27         5         60/27         5         60/27         6         1         60/27         5         1         60/27         5         1         60/27         5         1         6         7         7         60/27         5         1         6         7         7         60/27         5         1         6         7         7         60/27         5         1         6         7         7         60/27         5         1         5         1         5         1         5         1         5         1         5         1         5	i		SPA		L			L	L	L			
REFURBISH 50 Meter Pool         3           136         0rV 13         Pool \$16\$ Findsh         2400         87           137         0rV 13         Pool \$16\$ Findsh         2400         87         0.40         1,000           138         0rV 13         Pool \$16\$ Findsh         2400         87         0.40         1,000           138         0rV 13         Pool Cellecter Surge Tank         1         1.5         1,000         1,000           138         0rV 13         Pool Cellecter Surge Tank         1         1.5         3         105,000         3         155,000,00           140         0rV 13         Movine Finitia         1         1.5         3         105,000,00         3         155,000,00           141         0rV 13         Aqueto 1.8         1.5         3         260,000,00         3         765,000,00           142         0rV 13         Aqueto 1.8         1.5         4         5,400,00         3         765,000,00           143         0rV 13         Pool Bianket         1         5.4         4,760,00         4         4,780,00           144         0rV 13         Pool Bianket         1         5         4,760,00         5,600,00	135	D/V 13	Spa Complete	1	Ls	ļ I		ļ		Į	\$ 165,000	\$ 165,000.00	
REFURBISH 50 Meter Pool         3           136         0rV 13         Pool \$16\$ Findsh         2400         87           137         0rV 13         Pool \$16\$ Findsh         2400         87         0.40         1,000           138         0rV 13         Pool \$16\$ Findsh         2400         87         0.40         1,000           138         0rV 13         Pool Cellecter Surge Tank         1         1.5         1,000         1,000           138         0rV 13         Pool Cellecter Surge Tank         1         1.5         3         105,000         3         155,000,00           140         0rV 13         Movine Finitia         1         1.5         3         105,000,00         3         155,000,00           141         0rV 13         Aqueto 1.8         1.5         3         260,000,00         3         765,000,00           142         0rV 13         Aqueto 1.8         1.5         4         5,400,00         3         765,000,00           143         0rV 13         Pool Bianket         1         5.4         4,760,00         4         4,780,00           144         0rV 13         Pool Bianket         1         5         4,760,00         5,600,00				141 A	3.7 8871	53.20 A 19.5	Same Provent	2017-22-22	医分离机结束	14.00 × 50	K-LANTIN MALEVILLO	N. COMPANY COMPANY	AND
136         Div 13         Peed Sids Finish         1         Ls         \$         60/07         \$         60/07 0           137         Div 13         Paol Collector Surge Tank         1         Ls         \$         1,600         0,45         \$         0,040         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         \$         1,020,000,000         <				1.27.55	100039986	1255 Y 41 90%	PA	52.876813 	12120223	1. P.O. 1982	an a	1	ACCOUNT OF SALES
138         DIV 43         Peol Cellecter Burge Tank         1         <											ļ		
138         DIV 43         Peol Cellecter Burge Tank         1         <	137	ON 13	Pool Slap Finish	2400	8F						s 60,027 0,45	\$ 60,027.00 \$ 1,080.00	4 1
140         DV/13         Moytha Panala         1         Ls         3         260,000           141         DV/13         Status 2000 and 100 and 10	. 138	OIV 13	Pool Collector Surge Tank Pool Elitration and Pumps									\$ 17,500.00	3
142         DV 13         Aquato Life         1         Ls         \$ 5,400         \$ 6,400,00           143         DV 13         Lite Guard Chains         1         Ls         \$ 4,760         \$ 4,760         \$ 4,760,00           144         DV 13         Pool Blanket         5         Ls         \$ 1,350,00         \$ 6,000,00           144         DV 13         Pool Blanket         5         Ls         \$ 1,350,00         \$ 6,000,00           144         DV 13         Geo Thermail Heatra         1         Ls         \$ 205,000         \$ 205,000,00           145         DV 13         Geo Thermail Heatra         1         Ls         \$ 205,000         \$ 205,000,00           147         DV 13         Pool Equipment Installation         1         Ls         \$ 88,073.65         \$ 88,073.65           149         DV 13         Pool Equipment Installation         1         Ls         \$ 26,000,00         \$ 28,600,00           149         DV 13         Relocit profil Sociebard         1         Ls         \$ 10,000         \$ 18,600,00	140	<b>ON 13</b>	Mrythe Panela		Ls			\$ 260,000			<b></b>	\$ 250,000,00	1 1
146         DV/13         Pool Tile and Race fam finishes         1. Ls         \$ 60,000         6,000.00           147         DV/13         Pool Flumbing and Pjoing         1. Ls         \$ 68,073.65         \$ 68,000.00           148         DV/13         Pool Equipment installation         1. Ls         \$ 26,800.00         \$ 26,000.00           148         DV/13         Pool Equipment installation         1. Ls         \$ 26,800.00         \$ 36,000	142	ON 13	Aquatio Life					\$ 765,00D \$ 5,400				\$ 786,000.00 \$ 5,400.00	f
146         DV/13         Pool Tile and Race fam finishes         1. Ls         \$ 60,000         6,000.00           147         DV/13         Pool Flumbing and Pjoing         1. Ls         \$ 68,073.65         \$ 68,000.00           148         DV/13         Pool Equipment installation         1. Ls         \$ 26,800.00         \$ 26,000.00           148         DV/13         Pool Equipment installation         1. Ls         \$ 26,800.00         \$ 36,000	143	ON 13	Life Guard Chairs Roal Blanket		L L			4 780	[			3 6 780 00	1 1
146         DV/13         Pool Tile and Race fam finishes         1. Ls         \$ 60,000         6,000.00           147         DV/13         Pool Flumbing and Pjoing         1. Ls         \$ 68,073.65         \$ 68,000.00           148         DV/13         Pool Equipment installation         1. Ls         \$ 26,800.00         \$ 26,000.00           148         DV/13         Pool Equipment installation         1. Ls         \$ 26,800.00         \$ 36,000	145_	DIV 13	Geo Thermai Heatern	<u> </u>	Ls			\$ 205,000				\$ 205,000.00	1
148         DV/13         Pool Equipment Instantion         1         L         \$ 26,800.00         \$ 36,000         \$ 16,600	147	DIV 15	Pool Plumbing and Piping		La	<u> </u>	\$ 69,973.55	<u> </u>			<u>∓ 8,000</u>	\$ 58,973.53	t i
180 DDV 13 Pool Electrical Banding 1 La	148	DN 13	Pool Equipment Installation		Lŧ		\$ 25,800.00				5 10.600	\$ 25,600,00	1
	150	DIV 13	Pool Electrical Banding	1						1		\$ 2,000.00	1

	_	REFURBISH 50 METER POOL TOTAL COST										1,608,860,66
		Pools Equipment Room										
162		Hasoory Enclosures Waterproofing floors Hollow Metel Doors, Framas and Hardware	10725	L6 51 L6						\$ 38,650 3,00 7,695.00	\$ 38,850,00 \$ 32,176.00 \$ 7,895.00	
164 184	DIV 9 DIV 9	Stucco wall finishes Well Painting		L6						16,600.00	\$ 18,600,00 \$ 9,800,00	
157	DJV 14	Holating Equipment	1	14	Sec. Ma	5.000 D.F	24.55	Careford and a state of the s	2012.HAN 13	3,600.00	\$ <u>3,500.00</u>	
	·	TOTAL COST		的论题	A. Contra			97.500A	8.96%			<b>110,520</b> ,00
159	DN 3	Dive Tower Structure Reinforced Conorete structure	1	L						s275,000	\$ 275,000.00	
	DIV 6 DIV 9 DIV 13	Atuminum Reling and Remps Wall Anlates Diving Platforms	1	61 1.6 1.6						24,000.00 15,600.00 66,510.00	\$ 15,500.00 \$ 68,540.00	
162	DIV 18 DIV 18	Accent and Bollards Lighting PA and Annnouncement	1	 						178,000.00	\$ <u>178,000.00</u> \$ <u>22,500.00</u>	
		DIVE TOWER STRUCTURE			A							578,580.00
		Grandstand & Seating					2 78.2			SPACE INE SHORE		
105	DIV 3 DIV 1	Reinforced Cohorete structure Interfor Finishes under structure Concession Faulament		 						\$ 323,500 \$ 285,222 \$ 50,000	\$ 323,500.00 \$ 285,232.00 \$ 50,000.00	
167 158	DIV 12 DIV 13 DIV 12	Concession Equipment Elevated Bigscherz and SeaUng Cheir Lift Step Seating at Refueltan 50 Pool		L\$ 						476,008.00 34,000.00 65,000.00	\$ 476,000.00 \$ 34,000.00	
170	DIV 16 DIV 16	PA Paging and Announcements / Audio Electrical and Lighting		Le Le						\$ 85,000 120,000.00	\$ 85,000.00 \$ 120,000.00	
		ORAND STANDA SEATING		186				887899				1,427,722,00
		Pool Decks	100692	196.3622	89990638K	120 132 12 X X X	1997-007 ON	127 C 305 60 52	69999939999	20000-002-0020060	Sources and the	1356 CT 1. 1994 CT 2094 / 11 44,00
173	0N3 0N3	Elevered Post Deck Pool Deck Drainage	45012	SF Lt						5.50 \$ 35,000	\$ 35,000.00	
175 178	DIV 3 DIV 16 DIV 5	Ground Floor Pool Deck High Mast Pool Deck Lighting Pool Deck Enclosure Upper	13000 1 1220				<b>5</b> 146			4.00 376,385.00	\$ 376,385.00 \$ 176,900.00	
177	6 M 6	Pool Deck Enclosures Lower plass	421	UF 1905-2-25	1220000	201703233	\$ <u>185</u>	682338	1218-2210-54	87 85783829450A285	\$ <b>69,485.00</b>	
		TOTA/ COBT			971 (S. 1831) S. 1874 (S. 1874)							956,316,00
178	DIV 13	Shade / Waterproof Covering Grand Stand Shade Covering	1	6						786,000.00	\$ 786,000.00	
	• • • •		33576	6550626	BESTORA NY		2605-255 W	51. 60 <i>1</i> . 56 6	8.72er#78	N. ACCELLANES	26.00.0735.020.000	
		SHADE I WATERPROOFING COVERING TOTAL COST								94 - 3 - 3 - 3 - 3 - 3 - 3 - 3 - 3 - 3 -	1999 - 1998 1999 - 1998	
	Ļ	IOTAL COST	3989 1	8888	建全体的	19	18-21-5-0	5788 A 697			886 698 898 898 898 898 898 898 898 898	788,000,00.
179	DIV 2	Bath House / Hall of Fame Bldg Auger Cast Pillings		6	<b></b>					\$ 150,000	\$ 160,000.00	
180 181 182	DIV 3 DIV 3 DIV 7	Building Concrete / Masonry Structure Precessi Joist and Beams Roofing and Flashings		6 6 16						526,827.00 298,800.00 168,594.00	\$ 526,827.00 \$ 295,800.00	
183	DN Y DN B DN 6	Weterproofing Open Breezwaay Glazing and Storefronts Batcony Relitings		  		-				45,000.00 246,400.00 35,000.00	\$ 45,000.00 \$ 240,400.00	
188 187	DIVE	Hollow Metal Doora, framea Handware Stucco		Li Li						21,540.00 98,500.00	\$ 21,540.00 \$ 98,500.00	
169	DIV 8 DIV 14 DIV 15	Painting Elevators Fire Protection		Lo						58,600.00 \$ 250,000 \$ 122,600	\$ 122,600.00	
192	OIV 15 DIV 16 DIV 16	Electrical , Lighting and Fine Alarm		Ls Ls Ls				-		\$ 189,500 \$ 289,650 \$ 450,000	\$ 289,850.00	
131	DIVIS	部の事件に、中日子子主体化中心。認知 歴史におけていったのでなるの。 認識が読み、予定などのになる。	8200 10200	Ls						\$ 38 \$ 85,000 \$ 45	\$ 311,600.00 \$ 85,000.00	
197		NOTE INOF Second and Third Floor	10,00		<b> </b>				<u> </u>			
198		Build out By Owner		<u>.</u>	<u> </u>				<u> </u>			
	÷	EATHHOUSE HORELOOT	105		žoš						0.825	3,656,411,00
		SY TUNICAN UMA UMA	  5:20%	15650428	10000000	(11))ANDAS	1996-1999-1999-1999-1999-1999-1999-1999	NAME AND A	12473.093756		19860201 R 1890	ENGLISH COMPONENT & COMPANY COMPANY
		TOTAL										80,337,548.09
			1		r	1		1				
243		Overhead & Fees Ceveloper's Overhead Profit Fees		67%	<u> </u>		<b> </b>		1	<b> </b>	\$ 3,633,272.13	
		BER LEVELOP REPERTIES AND A STREET	i i i i i i				974582AN	39233		A REAL FOR THE	3.6 <i>7.778</i>	3,653,772,13
					<del></del>							

		Other Costs											
	DIV 1	Parmits Fees Allowance	<u> </u>										]
238	OIV 1	Parmits Fees Allowance	1	Ls							\$	92,545.00	
	0/1	Performance and Payment Bonde	h	<u> </u>	ļ								4
240		Performance and Payment Donge Reserves such lists approximate device Science States and the Second Science Sc	<u> </u>	Li	J		ļ		ļ		<u>.</u>	305,000.00	4
241	01/1 DIV1	Brilder Risk Insurance	h		l						f	85,780.00	1
241		1995 Comparison and a statistic for the statistic of the	<u>├</u>				· · · · · · · · · · · · · · · · · · ·		<b> </b>	· · · · · · · · · · · · · · · · · · ·	ł*	09.100.00	1
242		Project Specific E&O Policy	<u> </u>	1.5	•		·		l		1	28,809.00	1
	<u> </u>			·····	<b>.</b>				· · · · ·		· · · · ·		1
	<b>1</b>	OTHER COST TOTAL		1925.1	15		455.833		14 16 16		$3\times 3$	5. 199.	Carlos and the second second
			1.1.1			1. A.		15.8 20 1	1.2963 3	S. N. Postada	2.50	11. 12263	812,096.00
	<b> </b>	Contingency	w. v.		<u> </u>		2191293 X 12	<u>. N. 116 (1816)</u>	1.4.4.19.65	102 G	÷. *		
244		Developer Contingency		4	<b>}</b> ────		,				1	237,897.78	1
244	1	Project Contingency		<u></u> []	1				<u> </u>	······	17	150,000.00	
244	Į'		<u> </u>	·····	μ		a		<b></b>	L			J
	<u> </u>	Project Contingency Total		26.00	100 A 1940	14 N A A	S	STATES?	16 2000	SPACE STOR	- 22	Ser Baras	387.897.78
		the state of the s	<u></u>	13.74		<u> </u>			and the second second	Contract Institute in the design		2 10-3 11-37-1	
	1												
			1005	1.4	2			0866	同時ない	1. S. C. S.	1. 12	18. 18. 18.	CONTRACTOR STATES
		A SAN A CONTRACTOR	i Za	10.000		197 3260	Sec. And	3.23 8.4	262.00 13	SN 87 18 28 63	- 83 e -	546,20834	なびしていろうみものものないかいか
			1.77		C		72733	3 3 43	12326	Sec. Sec.	1023		
		TOTAL CONCEPTUAL OMP	1	1.5		1 St. 6 1 6 1 8	A BERNSON	1.5 . 1.5 . 1.6	日常がら	S. N. W. W. N.	8.5	XX 72333	
			11.6		S	1.1.1	Sec.	18 1878 3	1.85 6	1. No. 18 Sec.	1. 9	728. SS	
· ·		AQUATIC PACILITY	18.15			12.1.43	1.1.1.1	State States	19 A A		27.		\$ 24,864,980.00

Redevelopment of the Fort Lauderdale Aquatic Complex 09/17/12

ŧ

J

2         DN 1           3         DN 1           4         DN 1           5         DN 1           5         DN 1           5         DN 1           7         DN 1           9         DN 1           10         DN 1           11         DN 1           12         DN 1           13         DN 1           14         DN 1           15         DN 1           16         DN 1           17         DN 1           18         DN 1           19         DN 1           20         DN 1           21         DN 1           22         DN 1           23         DN 1           24         DN 1           25         DN 1           26         DN 1           27         DN 1           28         DN 1<	Design Fees Mader Flanding Active Start Construction Active Start Construction Structure Design Active Start Construction Chill Engineer Herchanical Engineer Excitomental Engineer Engineer Engineering Traine Engineering Traine Engineering Active Engineering Assign Proceeding Fre Construction Discourse Pro Construction Discourse Assign Proceeding Assign Proceeding Proceeding and Reports Print and Copies Postage and Courier Design and Courier Design and Courier Design Engineering Courier Construction Design Feeder Construct Beneral Construction Between Construction Betwee		Lis Lis Lis Lis Lis Lis Lis Lis Lis Lis	5 1960.00 5 1960.00 5 63.76 5 63.76 5 63.76 5 63.76 5 65.00 5 65.00 5 65.00 5 1.600.00 5 1.6	6 25,500,00 5 25,000,00 5 25,000,00 5 25,500,00 5 33,000,00 5 33,000,00 5 33,000,00 5 77,400,00 5 55,400,00 5 55,600,00 5 77,400,00 5 75,600,00 5 75,600,000 5 75,600,000,0000,000 5 75,600,000,000,000,000,000,000,000,000,00		(F.1 X C)	\$ 35,000 60 5 195,000 60 45,000 60 5 20,000 60 5 20,000 60 5 20,000 60 5 20,000 60 5 20,000 60 5 30,000 60 5 10,000 60 5 10,0	1 050000 1 020500 2 050000 3 2050000 3 35500000 3 35500000 3 35500000 3 75400500 3 75400500 3 75400500 3 75400500 3 10,000000 3 0,000000 3 4,000000 3 4,0000000 3 4,0000000 3 4,0000000 3 4,0000000 3 4,0000000 3 4,000000 3 4,0000000 3 4,000000 3 4,000000 3 4,000000 3 4,000000 3 4,000000 3 4,0000000 3 4,000000000000000000000000000000000000	\$77,50
2         DN 1           3         DN 1           4         DN 1           5         DN 1           5         DN 1           5         DN 1           7         DN 1           9         DN 1           10         DN 1           11         DN 1           12         DN 1           13         DN 1           14         DN 1           15         DN 1           16         DN 1           17         DN 1           18         DN 1           19         DN 1           20         DN 1           21         DN 1           22         DN 1           23         DN 1           24         DN 1           25         DN 1           26         DN 1           27         DN 1           28         DN 1<	Architectural Design Aquisito Design Bruediural Design Chritepolnear Hechanical Engineer Excitonmental Engineer Landszage Infrastical Engineer Landszage Infrastica Engineer California Engineering Traffic Engineering Traffic Engineering Traffic Engineering Assistant Pre Con Project Manager Assistant Pre Con Project Manager Design and Courier Prostoge and Courier Prostoge and Courier Prostoge and Courier Contemport Assistant Status Berger Construction Design and Courier Construction and Control Berger Construction Estimating and Cast Control Setting Froject Manager Field Engineer Status Project Manager Field Engineer Compared and Structural Control Weiding and Structural Control Meding and Structural Project Manager Compared Rest Project Manager Field Engineer Status Project Manager Compared Structural Control Meding and Structural Project Manager	1 1 1 1 1 1 1 1 1 1 1 1 1 1	Ls L	\$ 6376 \$ 1,000.00 \$ 1,000.00 \$ 1,000.00 \$ 85.00 \$ 85.00 \$ 85.00 \$ 1,500.00 \$ 1,500	\$ 76,400,00 \$ 76,400,00 \$ 25,500,00 \$ 25,500,00 \$ 25,500,00 \$ 25,500,00 \$ 25,500,00 \$ 33,000,00 \$ 77,400,00 \$ 55,400,00 \$ 55,400,00 \$ 55,400,00 \$ 55,400,00 \$ 55,400,00 \$ 77,400,00 \$ 77,400,00 \$ 33,000,00 \$ 77,400,00 \$ 33,000,00 \$ 77,400,00 \$ 33,000,00 \$ 77,400,00 \$ 33,000,00 \$ 35,000,00 \$ 35,000,000 \$ 35,0000,000 \$ 35,000,0000,0000,0000 \$ 35,0000,0000,0000,00000,0000,		\$ 8,400.00	s 195,000,00 1 1 1 1 1 1 1 1 1 1 1 1 1	1 193,000,00 1 193,000,00 1 05,000,00 1 120,000,00 1 1	
2         DN 1           3         DN 1           4         DN 1           5         DN 1           6         DN 1           7         DN 1           8         DN 1           9         DN 1           10         DN 1           11         DN 1           12         DN 1           13         DN 1           14         DN 1           15         DN 1           16         DN 1           17         DN 1           18         DN 1           19         DN 1           10         DN 1           20         DN 1           21         DN 1           22         DN 1           23         DN 1           24         DN 1           25         DN 1           26         DN 1           27         DN 1           28         DN 1           29         DN	Architectural Design Aquisito Design Bruediural Design Chritepolnear Hechanical Engineer Excitonmental Engineer Landszage Infrastical Engineer Landszage Infrastica Engineer California Engineering Traffic Engineering Traffic Engineering Traffic Engineering Assistant Pre Con Project Manager Assistant Pre Con Project Manager Design and Courier Prostoge and Courier Prostoge and Courier Prostoge and Courier Contemport Assistant Status Berger Construction Design and Courier Construction and Control Berger Construction Estimating and Cast Control Setting Froject Manager Field Engineer Status Project Manager Field Engineer Compared and Structural Control Weiding and Structural Control Meding and Structural Project Manager Compared Rest Project Manager Field Engineer Status Project Manager Compared Structural Control Meding and Structural Project Manager	1 1 1 1 1 1 1 1 1 1 1 1 1 1	Ls L	\$ 6376 \$ 1,000.00 \$ 1,000.00 \$ 1,000.00 \$ 85.00 \$ 85.00 \$ 85.00 \$ 1,500.00 \$ 1,500	6 25,500,00 5 25,000,00 5 25,000,00 5 25,500,00 5 33,000,00 5 33,000,00 5 33,000,00 5 77,400,00 5 55,400,00 5 55,600,00 5 77,400,00 5 75,600,00 5 75,600,000 5 75,600,000,0000,000 5 75,600,000,000,000,000,000,000,000,000,00			s 195,000,00 1 1 1 1 1 1 1 1 1 1 1 1 1	1 193,000,00 1 193,000,00 1 05,000,00 1 120,000,00 1 1	877,60
5         OV 1           6         OV 1           7         DV 1           9         OV 1           10         OV 1           11         OV 1           12         DV 1           13         DV 1           14         DV 1           15         OV 1           16         OV 1           17         DV 1           18         DV 1           19         DV 1           10         OV 1           21         DV 1           22         DV 1           23         DV 1           24         DV 1           25         OV 1           26         OV 1           27         DV 1           28         DV 1           29         DV 1           20         DV 1           21         DV 1           22         DV 1           23         DV 1           35         DV 1           36 <t< th=""><th>Chill Engineer Hechnical Engineer Erstrical Engineer Erstrical Engineer Erstricher Hellen Trainis Engineerins Trainis Engineerins Trainis Engineerins Pro Coattrochon Director Probeit Manager Aastaan Pro Con rreys and Boundarias Resonstruct Geo Technical Testing Asteatos Testing and Reports Protage and Courier Postage and Courier DEBIGN PEER TOTAL Sustainable Building Reviews DEBIGN PEER TOTAL Sustainable Building Reviews Building North Statustics Admin Threehold Inspector Bissing and Courier Setty Project Manager Field Engineer Setty Project Manager Comparation Testing Comparation Project Manager Comparation Pro</th><th>1 1 1 1 1 1 1 1 1 1 1 1 1 1</th><th>Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls L</th><th>\$ 6376 \$ 1,000.00 \$ 1,000.00 \$ 1,000.00 \$ 85.00 \$ 85.00 \$ 85.00 \$ 1,500.00 \$ 1,500</th><th>6 25,500,00 5 25,000,00 5 25,000,00 5 25,500,00 5 33,000,00 5 33,000,00 5 33,000,00 5 77,400,00 5 55,400,00 5 55,600,00 5 77,400,00 5 75,600,00 5 75,600,000 5 75,600,000,0000,000 5 75,600,000,000,000,000,000,000,000,000,00</th><th></th><th></th><th>6 05,000 00 5 120,000 00 5 20,000 00 5 365,000 00 5 355,000 00 5 15,000 00 5</th><th>1 0500.00 1 020 800.00 1 020</th><th></th></t<>	Chill Engineer Hechnical Engineer Erstrical Engineer Erstrical Engineer Erstricher Hellen Trainis Engineerins Trainis Engineerins Trainis Engineerins Pro Coattrochon Director Probeit Manager Aastaan Pro Con rreys and Boundarias Resonstruct Geo Technical Testing Asteatos Testing and Reports Protage and Courier Postage and Courier DEBIGN PEER TOTAL Sustainable Building Reviews DEBIGN PEER TOTAL Sustainable Building Reviews Building North Statustics Admin Threehold Inspector Bissing and Courier Setty Project Manager Field Engineer Setty Project Manager Comparation Testing Comparation Project Manager Comparation Pro	1 1 1 1 1 1 1 1 1 1 1 1 1 1	Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls L	\$ 6376 \$ 1,000.00 \$ 1,000.00 \$ 1,000.00 \$ 85.00 \$ 85.00 \$ 85.00 \$ 1,500.00 \$ 1,500	6 25,500,00 5 25,000,00 5 25,000,00 5 25,500,00 5 33,000,00 5 33,000,00 5 33,000,00 5 77,400,00 5 55,400,00 5 55,600,00 5 77,400,00 5 75,600,00 5 75,600,000 5 75,600,000,0000,000 5 75,600,000,000,000,000,000,000,000,000,00			6 05,000 00 5 120,000 00 5 20,000 00 5 365,000 00 5 355,000 00 5 15,000 00 5	1 0500.00 1 020 800.00 1 020	
8         DV/1           9         DV/1           10         DV/1           11         D/1           12         D/1           13         D/1           14         D/1           15         D/1           16         D/1           17         D/1           18         D/1           19         D/1           10         D/1           10         D/1           10         D/1           10         D/1           10         D/1           11         D/1           12         D/1           13         D/1           14         D/1           15         D/1           16         D/1           17         D/1           18         D/1           19         D/1           20         D/1           21         D/1           22         D/1           23         D/1           24         D/1           25         D/1           26         D/1           27         D/1	Environmental Engineer Lindisepe Irrigetion Engineering Pro Construction Director Probate Rengineering Assignan Pro Construction Probate Manager Assignan Pro Construction Proper Manager Assignant Resonature Construction Reports Profile and Copies Profile and Copies Profile and Copies Profile Rengineering Statistics Construction Biotechnic Construction Construction Assignment Statistics Construction Statistics Statistics Statistics Statistics Construction Legonde and Supervision Compared Structure Profession Statistics Statistics Compared Structure Profession Statistics Statistics Statistics Statistics Compared and Structure Profession Statistics Statistics Compared Structure Profession Statistics Statistics Compared Structure Profession Statistics Statistics Compared Structure Profession Statistics Compared Structure Profession Statistics S	40 480 25 1 1 1 1 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	Lis Lis Lis Lis Lis Mikes mihis Mikes Lis Lis Lis Lis Lis Lis Lis Lis Lis Li	\$ 6376 \$ 1,000.00 \$ 1,000.00 \$ 1,000.00 \$ 85.00 \$ 85.00 \$ 85.00 \$ 1,500.00 \$ 1,500	6 25,500,00 5 25,000,00 5 25,000,00 5 25,500,00 5 33,000,00 5 33,000,00 5 33,000,00 5 77,400,00 5 55,400,00 5 55,600,00 5 77,400,00 5 75,600,00 5 75,600,000 5 75,600,000,0000,000 5 75,600,000,000,000,000,000,000,000,000,00			\$ 20,000 60 \$ 35,000 60 \$ 16,000 60 \$ 50,000 60 \$ 10,000 60 \$ 10,	1 20,000,00 3 30,000,00 1 10,000,00 1 20,000,00 1 22,000,00 1 22,000,00 1 22,000,00 1 22,000,00 1 0,500,00 1 0,500,00 1 3 4400,00 1 4 400,00 1 3 20,000,00 1 3 3,000,00 1 3 3,000,00	877,60
DDV 1           DDV 1 </th <th>Integrition Engineering Training Engineering Pre Construction Director Pre Construction Director Director Director Construction Director Construction Director Pre Construction Director Pre Construction Director Pression Director</th> <th>40 480 25 1 1 1 1 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7</th> <th>Le Le mhs Wks Le Le Le mhs mhs Le Le Le Le Le Le Le Le Le Le Ke Mks Wks Wks Wks HRS HRS HRS HRS</th> <th>\$ 6376 \$ 1,000.00 \$ 1,000.00 \$ 1,000.00 \$ 85.00 \$ 85.00 \$ 85.00 \$ 1,500.00 \$ 1,500</th> <th>6 25,500,00 5 25,000,00 5 25,000,00 5 25,500,00 5 33,000,00 5 33,000,00 5 33,000,00 5 77,400,00 5 55,400,00 5 55,600,00 5 77,400,00 5 75,600,00 5 75,600,000 5 75,600,000,0000,000 5 75,600,000,000,000,000,000,000,000,000,00</th> <th></th> <th></th> <th>\$ 16,000.00 \$ 50,000.00 \$ 19,000 \$ 20,000 \$ 23,000 \$ 23,000 \$ 23,000</th> <th>1 10,000,00 1 75,200,00 2 25,505,80 2 25,505,80 3 25,505,80 3 10,500,00 3 10,500,00 3 10,500,00 3 10,500,00 3 10,500,00 3 3,5705,00 3 25,500,00 3 3,500,000 3 3,500,0000 3 3,500,0000 3 3,500,0000 3 3,500,0000000 3 3,5000000</th> <th>877,60</th>	Integrition Engineering Training Engineering Pre Construction Director Pre Construction Director Director Director Construction Director Construction Director Pre Construction Director Pre Construction Director Pression Director	40 480 25 1 1 1 1 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	Le Le mhs Wks Le Le Le mhs mhs Le Le Le Le Le Le Le Le Le Le Ke Mks Wks Wks Wks HRS HRS HRS HRS	\$ 6376 \$ 1,000.00 \$ 1,000.00 \$ 1,000.00 \$ 85.00 \$ 85.00 \$ 85.00 \$ 1,500.00 \$ 1,500	6 25,500,00 5 25,000,00 5 25,000,00 5 25,500,00 5 33,000,00 5 33,000,00 5 33,000,00 5 77,400,00 5 55,400,00 5 55,600,00 5 77,400,00 5 75,600,00 5 75,600,000 5 75,600,000,0000,000 5 75,600,000,000,000,000,000,000,000,000,00			\$ 16,000.00 \$ 50,000.00 \$ 19,000 \$ 20,000 \$ 23,000 \$ 23,000 \$ 23,000	1 10,000,00 1 75,200,00 2 25,505,80 2 25,505,80 3 25,505,80 3 10,500,00 3 10,500,00 3 10,500,00 3 10,500,00 3 10,500,00 3 3,5705,00 3 25,500,00 3 3,500,000 3 3,500,0000 3 3,500,0000 3 3,500,0000 3 3,500,0000000 3 3,5000000	877,60
I2         DV 1           12         DV 1           13         DV 1           14         DV 1           15         DV 1           16         DV 1           17         DV 1           18         DV 1           19         DV 1           10         DV 1           10         DV 1           11         DV 1           12         DV 1           13         DV 1           14         DV 1           15         DV 1           16         DV 1           17         DV 1           18         DV 1           19         DV 1           20         DV 1           21         DV 1           22         DV 1           23         DV 1           24         DV 1           25         DV 1           26         DV 1           27         DV 1           28         DV 1           29         DV 1           20         DV 1           21         DV 1           22         DV 1           24	Pre Construction Director Project Manager Assistant Pre Con revers and Boundarias Resonstruct Geo Technical Testing Arbestor Testing and Reports Profit and Copies Profitige and Courier Profitige and Courier Profitige and Courier Profitige and Courier Profitige and Courier Profitige and Courier Profitige and Courier Construction Administry Boundaries Construction Administry Boundaries Construction Administry Boundaries Construction Administry Boundaries Construction Boundaries Construction Boundar	40 480 25 1 1 1 1 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	mhs Wite Ls Ls Ls Ls Ls mhs Mhs Ls mhs Ls ts Hi Ks HAS Wite HAS Wite HAS	\$ 6376 \$ 1,000.00 \$ 1,000.00 \$ 1,000.00 \$ 85.00 \$ 85.00 \$ 85.00 \$ 1,500.00 \$ 1,500	6 25,500,00 5 25,000,00 5 25,000,00 5 25,500,00 5 33,000,00 5 33,000,00 5 33,000,00 5 77,400,00 5 55,400,00 5 55,600,00 5 77,400,00 5 75,600,00 5 75,600,000 5 75,600,000,0000,000 5 75,600,000,000,000,000,000,000,000,000,00			\$ 10.500 \$	1 23,000,00 3 25,000,00 3 10,000,00 3 10,000,00 3 0,000,00 3 0,000,00 3 0,000,00 3 0,000,00 3 0,000,00 3 25,000,00 3 25,000,00 3 25,000,00 3 25,000,00 3 25,000,00 3 33,700,00 3 33,700,00 4 77,600,00 4 77,600,00	877,60
16         D(V 1)         SU           16         D(V 1)         SU           17         D(V 1)         SU           18         D(V 1)         SU           18         D(V 1)         SU           18         D(V 1)         SU           19         D(V 1)         SU           10         D(V 1)         SU           11         D(V 1)         SU           12         D(V 1)         SU           13         D(V 1)         SU           14         D(V 1)         SU           15         D(V 1)         SU           16         D(V 1)         SU           17         D(V 1)         SU           18         D(V 1)         SU           19         D(V 1)         SU           10         D(V 1)         SU           13         D(V 1)         SU           14         D(V 1)         SU           15         D(V 1)         SU           16         D(V 1)         SU           17         D(V 1)         SU           18         D(V 1)         SU           19         D(V 1)	Incere and Boundaries Reconstruct Ges Tephical Testing Arbeator Testing and Reports Print and Copies Prostage and Courier Comparison of Courier Comparison of Courier Construction States of Courier Courier Construction Admin Threshold Inspector Behading Estimating and Cost Control Courier Construction Admin Threshold Inspector Behading Estimating and Cost Control Courier Construction Admin Threshold Inspector Scheduling Estimating and Cost Control Construction Language Instruction Scheduling Estimating and Cost Control Construction Language Information Scheduling Construction Language Information Bookteeping & Accounting Compared Structural Testing Welding and Structural Testing Compared Structural Testing	1 1 1 1 7 7 7 7 7 7 7 7 7 7 7 7 7	Ls Ls Ls mbs mbs mbs Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls Ls	3 85.00 3 85.00 3 85.00 3 85.00 3 85.00 3 85.00 3 1.500.00 3 1.500.00	\$ 26,500,00 \$ 22,500,00 \$ 33,000,00 \$ 77,400,00 \$ 77,400,00 \$ 55,400,00 \$ 55,400,00 \$ 55,400,00 \$ 55,400,00 \$ 55,400,00 \$ 72,400,00 \$ 50,000,00 \$ 72,400,00 \$ 72,400,000 \$ 72,400,0000 \$ 72,400,0000 \$ 72,400,0000 \$ 72,400,0000 \$ 72,400,0000 \$ 72,400,0000 \$ 72,4000000000000000000000000000000000000			\$ 10.500 \$	1 10,000,00 10,000,00 3 4,000,00 4,000,00 1 4,000,00 1 20,000,00 3 35,700,00 3 25,000,00 3 25,000,00 3 25,000,00 3 25,000,00 3 33,000,00 3 33,000,00 4 77,600,00	
B         DIV 1           10         DIV 1           11         DIV 1           12         DIV 1           13         DIV 1           14         DIV 1           15         DIV 1           16         DIV 1           17         DIV 1           20         DIV 1           21         DIV 1           22         DIV 1           23         DIV 1           24         DIV 1           25         DIV 1           26         DIV 1           27         DIV 1           28         DIV 1           29         DIV 1           20         DIV 1           21         DIV 1           22         DIV 1           23         DIV 1           24         DIV 1           25         DIV 1           26         DIV 1           27         DIV 1           28         DIV 1           29         DIV 1           20         DIV 1           21         DIV 1           22         DIV 1           24         DIV 1	Print and Copies Postage and Courier Postage and Courier DEDISIGN FEBRITOTAL General Conditions General Conditions Sustainable Building Reviews mittal Reviews Consultations Admin Threshold Inspector Cuelly Assurance and Control Beheduling Estimating and Cost Control Beheduling Estimating and Cost Control Status (Froject Managet Field Engineer Satus (Froject Managet Field Engineer Satus (Froject Managet Consultations and Common Bookkeeping & Accounting Bookkeeping & Accounting Bookkeeping & Accounting Compution Testing Compution Testing Weiding and Bruckural Forling Weiding and Bruckural Forling	1 420 1 300 300 300 300 300 300 300 300 45 300 45 45 45 45 45 45 45 45 45 45 45 45 45	mkhs mkhs mkhs ke ke ke ke ke Wks Wks Wks Wks Wks Wks Wks Wks Wks Ke Ke Ke Ke Ke	\$ 85 00 \$ 55 00 \$ 2,160,00 \$ 1,550,00 \$ 1,550,00 \$ 1,600,00 \$ 1,600,00 \$ 1,600,00	\$ 5 33,000.00 \$ 77,400.00 \$ 59,400.00 \$ 64,000.00 \$ 50,050.00 \$ 72,000.00			\$ 20,000 \$ 35,700 \$ 25,000	1 4,000.00 3 20,000.00 3 25,000.00 3 3 3 5,000.00 3 5,0000.00 3 5,00000	\$77,60
Image: state	General Conditions Statunbb Building Reviews mittel Reviews Construction Admin. Threshold Inspector Guilty Assurance and Control Scheduling Estimating and Cost Control Welling and Cost Control Market Construction Control Scheduling Senior Project Manager Field Engineer Safety / Fire Wetch Officer Construction Layous and Surays Construction Layous and Surays Construction Layous and Surays Construction Layous and Surays Construction Structural Faring Compared Testing Compared Testing Compared Structural Faring Welding and Structural Faring	1 420 1 300 300 300 300 300 300 300 300 45 300 45 45 45 45 45 45 45 45 45 45 45 45 45	mha Le hn es mha Wka Wka Wks HRS Wks HRS HRS HRS HRS HRS HRS HRS	\$ 85 00 \$ 55 00 \$ 2,160,00 \$ 1,550,00 \$ 1,550,00 \$ 1,600,00 \$ 1,600,00 \$ 1,600,00	\$ 5 33,000.00 \$ 77,400.00 \$ 59,400.00 \$ 64,000.00 \$ 50,050.00 \$ 72,000.00			\$ 20,000 \$ 35,700 \$ 25,000	\$ 35,700,00 \$ 28,000,00 \$ 28,600,00 \$ 23,000,00 \$ 33,000,00 \$ 77,400,00 \$ 77,400,00	977,60
Image: state	General Conditions Statunbb Building Reviews mittel Reviews Construction Admin. Threshold Inspector Guilty Assurance and Control Scheduling Estimating and Cost Control Welling and Cost Control Market Construction Control Scheduling Senior Project Manager Field Engineer Safety / Fire Wetch Officer Construction Layous and Surays Construction Layous and Surays Construction Layous and Surays Construction Layous and Surays Construction Structural Faring Compared Testing Compared Testing Compared Structural Faring Welding and Structural Faring	1 420 1 300 300 300 300 300 300 300 300 45 300 45 45 45 45 45 45 45 45 45 45 45 45 45	mha Le hn es mha Wka Wka Wks HRS Wks HRS HRS HRS HRS HRS HRS HRS	\$ 85 00 \$ 55 00 \$ 2,160,00 \$ 1,550,00 \$ 1,550,00 \$ 1,600,00 \$ 1,600,00 \$ 1,600,00	\$ 5 33,000.00 \$ 77,400.00 \$ 59,400.00 \$ 64,000.00 \$ 50,050.00 \$ 72,000.00			35,700 23.000	\$ 35,700,00 \$ 28,000,00 \$ 28,600,00 \$ 23,000,00 \$ 33,000,00 \$ 77,400,00 \$ 77,400,00	
B         DV7 1           10         DV7 1           20         DV1 1           21         DV1 1           22         DV1 1           23         DV1 1           24         DV1 1           25         DV1 1           26         DV1 1           27         DV1 1           28         DV1 1           29         DV1 1           20         DV1 1           21         DV1 1           22         DV1 1           23         DV1 1           24         DV1 1           25         DV1 1           26         DV1 1           27         DV1 1           28         DV1 1           29         DV1 1           20         DV1 1           21         DV1 1           22         DV1 1           23         DV1 1           24         DV1 1           25         DV1 1           26         DV1 1           27         DV1 1           28         DV1 1           29         DV1 1           20         DV1 1	Buildinable Building Reviews mittel Reviews Construction Admin. Threshold Inspector Quality Astrustonce and Centrol Bellmiking and Centrol Bellmiking and Centrol Assistant Project Manager Field Engineer Safety / Fra Watch Officer General Buper Intendant Bootkeeping & Accounting Deorper Jona Structural Terring Weiding and Structural Terring Weiding and Structural Terring Weiding and Structural Terring Weiding and Structural Terring States / Frances States / Structural Structural Compared Terring Deorper Jona Structural Terring Weiding and Structural Terring States / Frances States / Structural Structural States / Structural Terring States / States / S	1 420 1 300 300 300 300 300 300 300 300 45 300 45 45 45 45 45 45 45 45 45 45 45 45 45	mha Le hn es mha Wka Wka Wks HRS Wks HRS HRS HRS HRS HRS HRS HRS	\$ 85 00 \$ 55 00 \$ 2,160,00 \$ 1,550,00 \$ 1,550,00 \$ 1,600,00 \$ 1,600,00 \$ 1,600,00	\$ 5 33,000.00 \$ 77,400.00 \$ 59,400.00 \$ 64,000.00 \$ 50,050.00 \$ 72,000.00			35,700 23.000	\$ 35,700,00 \$ 28,000,00 \$ 28,600,00 \$ 23,000,00 \$ 33,000,00 \$ 77,400,00 \$ 77,400,00	
DV 1           2         DV 1           33         DV 1           44         DV 1           45         DV 1           46         DV 1           47         DV 1           48         DV 1           49         DV 1           40         DV 1           41         DV 1           42         DV 1           43         DV 1           44         DV 1           45         DV 1           46         DV 1           47         DV 1           48         DV 1           49         DV 1           40         DV 1           41         DV 1           42         DV 1           43         DV 1           44         DV 1           45         DV 1           46         DV 1           47         DV 1           48         DV 1           49         DV 1           40         DV 1           41         DV 1           42         DV 1           43         DV 1           44         DV 1	Threshold (napsetor Quality Arturance and Control Scheduling Estimating and Cast Control Assistant Project Managur Assistant Project Managur Field Engineer Safety / Fra Watch Officer General Buger Intendant Bookkeeping & Accounting Dombridge and Super Intendant Construction Leycols and Supervised Construction Resource Testing Compared Testing Compared Testing Compared Testing Compared Testing Compared Testing Compared Testing Compared Testing Compared Testing	1 300 100 38 38 38 38 38 38 38 770 45 1600 45 1600 45 1600 11 8 45 1600 11 8 10 10 10 10 10 10 10 10 10 10 10 10 10	Ls hrs mhs Wks Wks Wks HRS Hrs Ls Ls	\$ 85 00 \$ 55 00 \$ 2,160,00 \$ 1,550,00 \$ 1,550,00 \$ 1,600,00 \$ 1,600,00 \$ 1,600,00	\$ 5 33,000.00 \$ 77,400.00 \$ 59,400.00 \$ 64,000.00 \$ 50,050.00 \$ 72,000.00			\$ 25,000	\$ 25,000.00 \$ 25,500.00 \$ 23,000.00 \$ 33,000.00 \$ 77,600.00	
44         DV/1           5         DV/1           6         DV/1           7         DV/1           9         DV/1           10         DV/1           14         DV/1           15         DV/1           16         DV/1           17         DV/1           18         DV/1           19         DV/1           10         DV/1           11         DV/1           12         DV/1           13         DV/1           14         DV/1           15         DV/1           16         DV/1           17         DV/1           18         DV/1           19         DV/1 </td <td>Estimating and Cast Centrel Weild and Project Managet Assistant Project Managet Setty / Fried Match Officer Central Super Infendent Bookkeeping &amp; Accounting Densution Layout and Surveys Compared Testing Dompution Testing Weiding and Structural Testing Weiding and Structural Testing</td> <td>800 333 386 330 770 455 1500 8 1500 20 1 150 20 1</td> <td>mhs Wke Wks HRS Wks His Ls</td> <td>\$ 2,150.00 \$ 1,550.00 \$ 1,500.00 \$ 05.00 \$ 05.00 \$ 1,600.00</td> <td>\$ 77,400.00 \$ 59,400.00 \$ 64,000.00 \$ 50,050.00 \$ 72,000.00</td> <td></td> <td></td> <td></td> <td>\$ 33,000.00 \$ 77,400.00</td> <td></td>	Estimating and Cast Centrel Weild and Project Managet Assistant Project Managet Setty / Fried Match Officer Central Super Infendent Bookkeeping & Accounting Densution Layout and Surveys Compared Testing Dompution Testing Weiding and Structural Testing Weiding and Structural Testing	800 333 386 330 770 455 1500 8 1500 20 1 150 20 1	mhs Wke Wks HRS Wks His Ls	\$ 2,150.00 \$ 1,550.00 \$ 1,500.00 \$ 05.00 \$ 05.00 \$ 1,600.00	\$ 77,400.00 \$ 59,400.00 \$ 64,000.00 \$ 50,050.00 \$ 72,000.00				\$ 33,000.00 \$ 77,400.00	
7         DV(1)           8         DV(1)           9         DV(1)           10         DV(1)           11         DV(1)           12         DV(1)           13         DV(1)           14         DV(1)           15         DV(1)           16         DV(1)           17         DV(1)           16         DV(1)           17         DV(1)	Assistant Project Manager Field Engineer Safety / Fire Watch Officer Genaral Super Infendant Bookteering & Accounting Construction Legode and Surveys Construction Legode and Surveys Construction Legode and Surveys Construction Testing Dompastion Testing Walding and Structural Farling Construction Photographe	38 30) 770 45 1500 8 1 1 1 1 50 20 	Wks Wks HRS Wks Hn Ls	\$ 1,650.00 \$ 1,600.00 \$ 65.00 \$ 1,600.00	\$ 59,400.00 \$ 54,000.00 \$ 50,050.00 \$ 72,000.00			······································	\$ 17,400.00 \$ 59,400.00	
We         Div 1           33         Div 1           34         Div 1           35         Div 1           36         Div 1           37         Div 1           38         Div 1           39         Div 1           36         Div 1           37         Div 1           38         Div 1           39         Div 1           40         Div 1           41         Div 1           43         Div 1           44         Div 1           45         Div 1           46         Div 1           47         Div 1           48         Div 1           49         Div 1           40         Div 1           41         Div 1           42         Div 1           43         Div 1           44         Div 1           45         Div 1           46         Div 1           47         Div 1           48         Div 1           49         Div 1           40         Div 1           41         Div 1 <td>General Bugar Intendiant Bookteeping &amp; Accounting Construction Levante and Surveys Construction Levante and Surveys Construction Testing Weiding and Structural Terring Testing Construction Photography related Construction Photography</td> <td>45 1600 1 1 1 1 1 50 20 20 1</td> <td>Wites Hits Ls 48</td> <td>\$ 1,600.00</td> <td>\$ 72,000.00</td> <td></td> <td></td> <td></td> <td>\$ 54,000.00</td> <td></td>	General Bugar Intendiant Bookteeping & Accounting Construction Levante and Surveys Construction Levante and Surveys Construction Testing Weiding and Structural Terring Testing Construction Photography related Construction Photography	45 1600 1 1 1 1 1 50 20 20 1	Wites Hits Ls 48	\$ 1,600.00	\$ 72,000.00				\$ 54,000.00	
A         B // 1           0         D// 1           0	Construction Levolds and Surveys Construction Tealing Comparison Tealing Weiding and Structural Farling Hand Construction Photography related Construction Photography	1 150 20	49					 	\$ 50,050.00 \$ 72,000.00 \$ 82,500.00	
4 DiV 1 5 DiV 1 5 DiV 1 7 DiV 1 9 DiV 1 9 DiV 1 9 DiV 1 1 DiV 1 1 DiV 1 9 DiV 1 1 DiV 1 2 DiV 1 1 D	Consrete Testing Compartion Testing Welding and Structural Testing Testing and Structural Testing Testing and Structural Testing (Structure Testing and Structure Testing)	150 20 1		-				\$ 25,000	\$ 25,000.00	
Image: Constraint of the second sec	rial and Construction Photographs							\$ 50 \$ 105 \$ 5,000	\$ 2,100.00	
8         Div 1           9         Div 1           0         Div 1           1         Div 1           2         Div 1           3         Qiv 1           4         Div 1           5         Qiv 1           6         Div 1           7         Div 1           8         Qiv 1           9         Div 1           10         Div 1           11         Div 1           12         Div 1           13         Div 1           14         Div 1           15         Div 1           16         Div 1           17         Div 1           17         Div 1           17         Div 1	Mobilitation and Salus					1 225		<u>\$ 3,000</u>	\$ 2,250.00	
Image: 1 minipage         34582123           1         DiV 1         34582123           2         DiV 1         34582123           3         DiV 1         34582123           4         DiV 1         34582123           5         DiV 1         34582123           6         DiV 1         34582123           7         DiV 1         345823           9         DiV 1         345823           1         DiV 1         345823           2         DiV 1         345823           3         DiV 1         345823           3         DiV 1         345823           4         DiV 1         345823           5         DiV 1         345823           7         DiV 1         345823	Steging Aree			\$ 26.00	<u>\$</u>	ş			3 <u>.</u> 5 <i>.</i>	
2 0// 1 3 0// 1 3 0// 1 5 0	Office Trails Bet up	<u> </u>	LL			\$		 	<u> </u>	
0.00.1         0.00.1           0.00.4         0.00.4           7.00.7         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4           0.00.4         0.00.4	Construction Signage	0 50 10	Ls.			\$ \$ 300			\$ 3,000.00	
8         DIV 1           9         DIV 1           1         DIV 1           2         DIV 1           2         DIV 1           3         DIV 1           4         DIV 1           5         DIV 1           6         DIV 1           7         DIV 1           9         DIV 1	Blorage Containsis	10	L4 MUh3			\$ 3,800 \$ 460		 	\$ 3,800.00 \$ 4,500.00	
0         DIV 1           1         DIV 1           2         DIV 1           3         DIV 1           4         Civ 1           5         DIV 1           6         DIV 1           7         DIV 1           4         Civ 1           5         DIV 1           6         DIV 1           7         DIV 1           4         Giv 4	Tollet Facilities Temporary Fending Barrisades	- 10 - 1 - 10	Ls			\$ 200 \$ 12,500 \$ 300	· · · ·		\$ 2,600.00 \$ 12,600.00 \$ 3,000.00	
2 DIV 1 DIV 1 4 DIV 1 5 DIV 1 5 DIV 1 5 DIV 1 5 DIV 1 7 DIV 1 6 DIV 1	Temporary Water Connection Temporary Electric	1	L6 L6			1			3 ·	
5 DIV1 6 DIV1 7 DIV1	Fire Department Connection Telephone Connection Water Monthly	10	<u>15</u> L#			\$ \$ \$ 800		· · · · · · · · · · · · · · · · · · ·	\$ 5 5 5,000.00	
	Electrical Monthly Phone Monthly	10	Mîhu		·	\$ 400 \$ 300			3 4,000.00 3 3,000.60	
10 C 20 C	Dumpatiere Trash Chifee	65				\$ 400 \$ 400			8 26,000.00 8 4,000.00	
0 [DIV1]	Temporary Labor Delty Clean Up	2050 720	)] mhs	\$ <u>15.00</u> \$ <u>15.00</u>	\$ 31,200.00 \$ 10,800.00				\$ 31,200.00 \$ 10,800.00	
1 DN 1 2 DN 1	Find Clean Up Seatfoldings		L.s.	<u>\$ 6,500.00</u>		\$ 840			\$ 5,500.00 \$ 7,560,00	
3 DIV1 4 DIV1	Safety Neta Water , Ice ( Workmen Dally) Safety Natariala		Mth Mth			\$ 400 \$ 800			\$ 3,000.00 \$ 7,200.00 \$ 4,050.00	
5 DIV 1 6 DIV 1	Equipment Rental Holeting		) M(h			\$ 450 \$ 1,200		· · · · · · · · · · · · · · · · · · ·	\$ 10,800.00	
7 DIV 1 8 DIV 1	Security Wetchman	40	) Wks	\$ 1,200.00 \$ 400.00					\$ 48,000,00 \$ 15,000.00	
9 DIV 1 0 DIV 1	Gale and Delivery Security					\$ 1,200 \$ 600 \$ 670			\$ 12,000.00 \$ 6,000.00 \$ 6,700.00	
2 01/1	Milesge and Travel RDC Trucks Equipment Field	20	)/ <u>Mth</u>	\$ 1,750.00	\$ 3,500.00				\$ 3,500.00	
73 DIV 1 74 DIV 1 75 DIV 1 Pr	Milesge and Transf ROC Trucks Equipment Fuel Punch List	20	Mth Wite	<u>4 1,700.00</u>				 § <u>2.000</u>		
16 DN 1	Mileage and Travel RDC Trucks Equipment Fuel	20	Mth			8 1,500 5 500			\$ 3,500.00 \$ 1,500.00 \$ 500.00	

<u> </u>												
	1	Parking Garage	<u> </u>	ľ				<u> </u>				
		Parking Garage				ş .				<u>s</u> .	i	
19B	DIA 2	20 " Diameter Auger Cest Piles Extra contractor in the international action of the internation of th	1	L8				\$ <u>.</u>		\$ 225,000.00	3 225.000.00	
	DIV 3	Pile Gaps	1	18						\$ 60,000.00	\$ 80,000,00	
		Ciumum Starters Shear Wall concrete	5	<u>LS</u>						\$ 15,850.00 \$ 34,580.00	\$ 15,850.00 \$ 34,560.00	
203		Grade Beams Stair Walls	1	LS						\$ 90,500.00 \$ 14,500.00	\$ 90,500,00 \$ 14,500,00	
Т	<u>₽/¥ 3</u>	网络拉斯地址的自己自己的自己的和地址的原始										
	DN 3 DN 3	Ground floor Sizh Visgueen	1	LS LS						\$ 470,200.00 \$ 10,300.00	\$ 470,200,00 \$ 10,300,00	
	0143	Ploor backfilling	Í	1.6						\$ 27,500.00	\$ 27,500,00	
208	DIV 3	Concrete columnia	- 1	1.5				<b>\$</b> -		\$ B7,500.00	\$ B7,500.00	
209	DIV 3	Shear Wells Beams concrete	1	<u>L6</u> L8						\$ 48,500.00 \$ 167,600.00	\$ 45,500.00 \$ 167,500.00	
210	DIV 3	Precast Joists	;	L8				·		\$ 225,300.00	\$ 225,300.00	
212	DN 3	Concrete deck elab Stairs concrete	1	L9 L8						\$ 285,300.00 \$ 14,850.00	8 285,300.00 \$ 14,850.00	
	DIVS	MARIANANANANANANANANANANANANANANANANANANA						s -				
		Concrete columns Shear Wells	- 1	L8 L8						\$ 87,500.00 \$ 46,500.00 \$ 176,000.00	\$ 07,503.00 \$ 48,603.00	
216	DIV 3	Beams congrete	1	18		· · · · · · · · · · · · · · · · · · ·				\$ 175,000.00	\$ 175,000,00	
	DIV 3 DIV 3	Preçast Jolats Concrete deck eleb		16 16						\$ 225,000.00 \$ 285,765.00	\$ 225,000.00 \$ 285,765.00	
	014.3	Stalro concrete	1	16						\$ 14,500.00		
220	DIV 3	Concrete golumns	1	LS			<b>_</b>	\$		\$ 28,769.00	3 28,760.00	
221	DIN 3	Shear Walta	1	LS						\$ 42,500.00	\$ 42,500.00	
	DIV 3	Béame concrete Precest Jolais	<u>├</u> ¦	18 18			<u> </u>			\$ 158,400.00 \$ 220,450.00	\$ 158,400.00 \$ 220,450.00	
224	DIV 3	Conciste deck alab Stalva concrete	î	16 18						\$ 269,300.00 \$ 14,650.00	\$ 269,300.00	
	0 N S	MARSNELSSING AND THE COUNTRIES	Ľ									
	DIV 3 DIV 3	Barrier walls Second Floor Barrier walls Third Floor Floor	1	LS LS						\$ 45,300,00 \$ 56,000,00	\$ 45,309.00 \$ 65,009.00	
228	DIV 3	Tenelon cable first fibor	1	LS						\$ 55,000.00	\$ 85,000.00	
	017 3 D17 3	Tension cable second floor floor Tension cable third floor		LS					$\vdash$	\$ 35,000,00 \$ 42,000,00	1 35,000.00 \$ 42,000.00	
		REPORTED AND A CONTRACT STREET										
	DIV 2 DIV 2	Skipping Arrows	1	<u>18</u> 18						\$ 11,250.00 \$ 2,660.00	\$ 11,750.00 \$ 2,660.00	
233	DIV 2	Stop bars	ļ	L6						\$ 900.00 \$ 665.00		
	DN 2	Handicap markings Handicap algos on polos		LS L8						\$ 2,052.00	\$ 2,062.00	
	D/V 18	Handlesp signs on poles Manual State Stat		16						\$ 260,675.00	\$ 280,675.00	
	DIV 15	Fire Sprinkler	Í	ĬŠ						\$ \$2,100.00	8 92,100.00	
238	DIV 14	Elevators	1	1.8						\$ 226,000.00	\$ 225,000.00	
		GARAGE STRUGTURETOTAL	16 A	1995 (S) 1997 (S)	8 S K	10.33	35 N 26		SS 233	3 6 8 1 2 0 E Z		
┝╍═╋			1464.61	$S_{2}^{*} S_{2}^{*} S_{2$	0.45.47	839 A 14	N 23 87 97	S. 28 S. A	2448834	23253-388893	\$55550 (CENS)	4,195,697.00
			Sec.	10-2-5	$0 \le \infty$	ala kesi:	12. 208	C. Marshard	.08.685		ZAGARAK	ALC
( I		7074	紧紧	A	$\mathcal{A}$	1 60. 199 - 1 1 6 / 199 - 1						6,027,307.00
├──╂		ALL CONTRACTOR OF A DATA AND A DATA AND A DATA	CO. Aline		11.22303933		N YO NW	<u>-36, 5 5 666 (</u>	128.026340	1990, N 92 (97, 33) 442	A 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 100 / 1	
		Overhead & Fees										
243		Developer's Overhead / Profit Fee	17	.87%							\$ 1,077,305.78	
243		· · · · · · · · · · · · · · · · · · ·						····	<b>.</b>			
		DEVELOPERS FLESTOTAL	\$38.5	158434		A. 5. 8		1. A. W.	S. 8		201612-05-050	1,077,303.78
┝──┦		Other Costs										
++	DAL	ENDANGAL THE AND A SARAGES					r		L			1
238	DN 1 DN 1	Perants Fees Allowance	1	L8			ļ		Į		\$ \$0,000,00	
240	DN 1	Performance and Payment Bonds	1	Ls.							\$ 65,000.00	
941	01/1	Builder Bick Insurance	<u> </u>	$\square$					Į		\$ 35,000.00	
	<u>ōw i</u>	<b>建设,建筑和市场和市场的</b> 。					1		<b>.</b>		• •••••	1
242	DN 1	Project Specific E&O Policy	1	L8.	·		1		ł	<b> </b>	\$ 10,000.00	
		CONTRACTOR AND ADDRESS OF A DECISION ADDRESS	1208	335/	a and a second	2.6.2	1. AN 5-	1. 57. 84.4	1. 1. 1. 1. 1.	2000 822		SA SA SA SA AKAKA
			132	院際	公元的	國的為	湯や湯		123.6347			150,000.00
			1									
		Contingency	1		<b></b>	· · · ·	T	r			r	1
244		Developer Contingency	ĩ			ļ					\$ 217,871.22	1
244		Project Centingency	1		<b></b>	· · · · · · ·	l		Į	{	\$ 100,000.00	4 .
		1							h	i <u></u>	<u> </u>	<u> </u>
		Provide Contineed as a second	19424			1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	$\phi$	$y \leftrightarrow x \times x$	680 E	12 6 6 6 6	1995 1996 1996	<b>经资料税税税</b> 税利益利益利益
1 T	_											
												CONTRACTOR ACCOUNT AND A D
		TOTAL GONGARTUAL ONE GANGING GARAGE	]									
1												\$ 7,572,484.90
1 1		and the second of the second state of the seco	3							I.,,,		ELANDER AND A STATE AND A STATE AND A
		· · · · · · · · · · · · · · · · · · ·										

## FORT LAUDERDALE AQUATIC CENTER

Developer : Recreational Design & Construction Owner : City of Fort Lauderdale

۰Г	BASIS OF GMP	3.10	.3.4	AQUA	CTIC FA	CILITY	Y GENE	RAL C	ONDITIONS	j	
ITEN NO.	B. ITEM DESCRIPTION		UNIT	1. Unit Coal	ABOR 2. Total	F. MA 1. Únil Cost	TERIAL	EQUIPMENT	SUB CONSULTANTS	TOTAL LINE ITEM PRICE (E.2 + F.2)	GRAND TOTAL
	General Conditions										
		· · · · · · · · · · · · · · · · · · ·									
19 20	Sustainable Building Reviews Submittal Reviews/ Construction Admin	1800	L6 mhs	\$ 65,00					\$ 45,000 \$ 138,000	45,000.00 136,000.00	
21	Threshold Inspector	1	Ls						\$ 55,000	\$ 65,000.00	
22 23	Quality Assurance and Control Scheduling	520	65	\$ 85.00	\$ 44,200.00				\$25,000	\$ <u>44,200.00</u> \$ 25,000.00	
24 8	Estimating and Cost Control	1700		\$ 55.00	\$ 93,500.00					\$ 93,600.00	
26	Senior Project Manager	11		\$ 2,150.00	\$ 165,550.00					\$ 165,560,00	
27 20	Assistant Project Manager Field Engineer	77		<u>\$ 1,650,00</u> \$ 1,500,00	\$ 115,500.00					\$ 127,050.00 \$ 118,600.00	
28 30	Salety / Fire Watch Officer General Super Intendent	770		\$ 65.00 \$ 1,600.00	\$ 50,050,00					\$ \$0,050.00 \$ 123,200.00	
31	Pool Foreman	48	Wks	\$ 1,600.00	\$ 72,000.00					\$ 72,000.00	
32	Building Foreman Boolkeeping & Accounting	1500			\$ 115,500.00 \$ 82,500.00					\$ 115,600.00 \$ 82,600.00	
ji	Boolkeeping & Accounting			·							
33	Construction Layouts and Buryays	'	L\$						\$ 45,000	\$ 45,000.00	
34 35	Concrete Testing Compaction Testing	150	69 69	h		<u> </u>			\$ <u>50</u> \$105	\$ 7,500.00 \$ 2,100.00	
38	Welding and Structural Testing	ĭ							\$ 5,000	\$ 6,000.0b	
17 T	Arial and Construction Photographs	18	mths			\$ 225	··			\$ 4,050.00	
38	Mobilization and Safup	120	mhs	\$ 25.00	\$ 3,000.00					\$ 3,000.00	
39	Staging Area	1	L:	20,00		\$ 2,000				\$ 2,000.00	
40	Office Trails Set up		E	l		\$ 1,700			··	\$1700.00	
41	Project Signage	2	· en			\$ 500 \$ 2,000				\$ 1,000.00 \$ 2,000.00	
<u>*</u> 5	Construction Signate Construction Signate Conce Job Site Trailer						·				
43	Office Job Site Trailer Sub Contractor Filed Office	<u>18</u> 17				\$ 300 \$ 225		<u> </u>		\$ 5,400.00 \$ 3,828.00	
48	Field Office Furnishings	1	8			\$ 3,800				\$ 3,400.00	
46	Storage Containers Toilet Facilities	15	Miths Miths			450 450 450				\$ 8,760.00 \$ 4,680.00	
49	Temporary Feacing Barricades	1	Ls. milhs			\$ 12,500 \$ 300				\$ 12,500.00 \$ \$,400.00	
1	Temporary Water Connection		1								
50 51	Temporary Water Connection Temporary Electric		<u>لة</u> لة	···		\$ 2,00D \$ 5,000		{		4 2,000.00 \$ 6,000.00	
5 <u>2</u> 53	Fire Department Connection Telephone Connection	1	<u>ե</u> ։ Այ			\$ 7,500 \$ 50D				\$ 7,800.00 \$ 500.00	
64	Water Monthly	16	Miths			\$ 500				\$ 9,000.00	
66 58	Electrical Monthly Phone Monthly	18			<u> </u>	\$ 400 \$ 300	}	<b> </b>		\$ 7,200.00 \$ 6,400.00	
67	Phone Monthly Phone Monthly Dumpsters	65				\$ 400				\$ 28,000.00	
Řπ.	Trash Chudes	10				\$ 400				\$ 4,000.00	
59 [4	Temporary Labor	_ 2080	mhs	\$ 15.00	\$ 31,200.00			ļ		\$ 31,200.00	
60	Daily Clean Up	720	mhs	5 15.00 3 8,500.00	\$ 10,800.00					\$ \$0,800.00	
61 	Final Clean Up	<u> </u>	<u>_L</u> 6	\$ 0,000.00	<u> </u>	<u> </u>				\$ 6,500.00	
<u>62</u> 63	Scaffoldings Safety Nets	16				<u>\$                                    </u>		<u>↓</u>		\$ 13,440.00 \$ 6,400.00	
64	Water, Ice (Workmen Dally)	18	Mth	I		\$ 800		<u> </u>		\$ 14,400.00	
65 68	Gafety Materials Equipment Rental Holsting	17				\$ 450 \$ 1,200				\$ 7,650.00 \$ 20,400.00	
67	Becurity Watchman	70	Wks	\$ 1,200.00	<u> </u>					\$ \$4,000.00	
68	Gate and Delivery Security			\$ 400.00						\$ 28,000.00	
69 1	Mileage and Travel	14	Mth	<u> </u>	<u> </u>	\$ 1,200			<b>.</b>	\$21,600.00	
70 71	Equipment Fuel	18				\$ 000 \$ 670				\$ 10,600.00 \$ 12,060.00	
į,	1999년 20일 - 1999년 1999년 1997년 199 1997년 1997년 1997			<b></b>							
72	Punch List As Builts	4		<u>\$ 1,750.00</u>	\$ 7,000.00	<b></b>	<u>t</u>		\$ 4,000	6 7,000.0p 5 4,000.00	
73 74 75	Demobilization Printing and Binding, Turn Over Docs		Lș Ls		[	\$ 1,200		<u> </u>	·····	\$ 5,000.00 \$ 1,200.00	
76	Parangend Balang, turn over Docs Postage and Delivery		Ls Ls			<b>1</b> 500				\$ 800.00	
	CONDITIONS TO YAL CONT										¥ 747,508.
<u>דר</u> ה					<b>Basesses</b>		833-63 <b>5</b> 64	T			
1000 C	CONDITIONS CONDITIONS AQUATICY/AGISTY/										\$ 1,717,396.

### FORT LAUDERDALE AQUATIC CENTER

Developer : Recreational Design & Construction Owner : City of Fort Lauderdale

#### **BASIS OF GMP 3.10.3.4: PARKIGN GARAGE GENERAL CONDITIONS** F. MATERIAL EQUIPMENTI SUB CONSULTANTS GRAND TOTAL B ITEN DESCRIPTION ÖTY LINIT LABOR TOTAL LINE . ITÊM ITEM PRICE (E.2 + F.2) NO. 1. Unit Cost 2. Total 1. Unit Cosl 2. Total **General Conditions** Sustainable Building Reviews 20,000.00 35,700.00 25,000.00 25,500.00 25,000.00 19 Ls 20,000 85.00 Submittel Reviews/ Construction Admir 420 mha 21 Threshold inspector Quality Assurance and Control Ls 25.00 300 65.00 25,500.00 22 23 24 Lipit resource and Biology and Cost Control Extracting and Cost Control Extracting and Cost Control Senior Project Manager Field Engineer Batey / Fire Watch Officer General Super Intendent Bookkeeping & Accounting Construction Layouts and Sureys Construction Layouts and Sureys 25,000 ī 88 55,00 33.000.00 600 mhs 33.00 \$ 2,150,00 \$ 77,400,00 \$ 1,550,00 \$ 59,400,00 \$ 1,500,00 \$ 54,400,00 \$ 1,500,00 \$ 54,400,00 \$ 35,00 \$ 50,950,00 \$ 1,800,00 \$ 72,000,00 \$ 1,800,00 \$ 72,000,00 \$ 3,55,00 \$ 92,500,00 25 77,400.00 59,400.00 54,000.00 50,050.00 72,000.00 \$2,500.00 Wks Wks Wks HRS Wks 36 38 28 27 28 29 38 770 30 4 1500 Hns LS 25,000 25,000.00 33 Concrete Testing Compaction Testing Wetding and Structural Testing 7,600.00 34 36 38 15( eà 10 20 8ð 1.5 2,100.00 Arial and Construction Pholographs Arial and Construction Pholographs Medication and Barus approximation Mobilization and Barus 1609/251 milha 2,250.00 37 225 mhs 38 39 40 25.00 obilization and Staging Area Office Trails Bet up 1.6 1.8 Office Traine of the second se 41 42 ee Le . 3,000.00 3,600.00 4,800.00 2,600.00 12,800.00 3,000.00 Mths 300 3,800 .450 43 Field Office Furnishings Storage Containere Toilet Facilities La 45 46 47 Milas Toilet Facilities Temporary Facilities Barricades Statute Connection Temporary Valer Connection Temporary Electric Fire Departmet Connection Telephone Consection Weier Monthly Electrical Morthly Phone Monthly 260 48 49 Ls mihs 300 60 L8 . • L6 L8 L8 Miths Miths 61 62 63 5,000.00 4,000.00 3,000.00 64 Ĥ 500 <u>55</u> 400 Phone Montally Tesh Chura II Tesh Chura II Tesh Chura II Temporary Labor Daily Clean Up Elas Clean Up EA Milhs 67 58 400 26,000.00 85 10 15,00 \$ 15,00 \$ 208 mhu 31,200.00 10,800.00 8,600.00 31,200.00 59 60 mhs Ls \$ 8,509.00 61 Final Clean Up 7,660,00 3,600,00 7,200,00 4,050,00 Mlh 640 62 Bratfoldings Salety Nata Water , Ice ( Workmen Dally) Salety Materials 6caffold Mih Mih Mih Mih 400 600 450 63 84 65 66 Equipment Rental Holsting 1.200 10 800 0 5355355 Security Watchman Gate and Delivery Security 40 Wks \$ 1,200.00 \$ 400.00 48,000.00 67 68 Wks 40 - A. Mileage and Travel RDC Trucks Mih 69 1,200 12,000.00 70 Mth 800 6,000.00 Punch List Mih 3556 72 73 74 75 78 W/X B \$ 1,750.00 \$ 3.500.00 3.600.0 As Builts Demobilization Printing and Binding, Turn Over Docs Postage and Delivery 100 L8 L8 L9 L9 2,000.00 3,600.00 1,600.00 500.00 2,000 1,500 500 GENERAL CONDITIONS TOTAL COST (200)资格 199 854,210.00 GONDITIONS FARRING CARAO ŝ 0.00

# REDEVELOPMENT OF THE FORT LAUDERDALE AQUATIC COMPLEX

### BASIS OF GMP STAFFING PLAN

STAFFING DESCRIPTION	PERCENTAGES ARE BASED ON ATTACHED CONSTRUCTION SCHEDULE DURATION
1 - Plan, Permit & Inspection Processor	
Project Start Up	0%
Design Stage:	25%
Construction Documents / Permitting Stage:	100%
Construction Stage:	50%
Closeout Stage:	100%
Cibabout Stage.	10078
1 - Project Manager	
Project Start Up	15%
Design Stage:	25%
Construction Documents / Permitting Stage:	100%
Construction Stage:	100%
Closeout Stage:	100%
1 - Project Engineer	
Project Start Up	0%
Design Stage:	0%
Construction Documents / Permitting Stage:	25%
Construction Stage:	100%
Closeout Stage:	100%
1 - Clerical Personnel	
Project Start Up	0%
Design Stage:	0%
Construction Documents / Permitting Stage:	25%
Construction Stage:	100%
Closeout Stage:	100%
1 - General Superintendent	
Project Start Up	0%
Design Stage:	0%
Construction Documents / Permitting Stage:	15%
Construction Stage:	100%
Closeout Stage:	100%
4 - Deal Superintendent	
1 - Pool Superintendent Breiset Start Lie	
Project Start Up	0%
Design Stage: Construction Documents ( Permitting Stage)	0%
Construction Documents / Permitting Stage: Construction Stage:	0%
Closeout Stage:	70% 25%
Cioseout Stage.	2070

Page 24 of 27 Exhibit 1 CAM 15-1467 Page 154 of 158

# REDEVELOPMENT OF THE FORT LAUDERDALE AQUATIC COMPLEX

### BASIS OF GMP STAFFING PLAN

STAFFING DESCRIPTION	PERCENTAGES ARE BASE ON ATTACHED CONSTRUCTION SCHEDULE DURATION
1 - Building Superintendent	
Project Start Up	0%
Design Stage:	0%
Construction Documents / Permitting Stage:	0%
Construction Stage:	100%
Closeout Stage:	50%
1 - Layout, Safety, As-Bulit Superintendent	
Project Start Up	0%
Design Stage:	0%
Construction Documents / Permitting Stage:	25%
Construction Stage:	100%
Closeout Stage:	50%
1 Layout Safety Crew	
Project Start Up	001
	0% 0%
Design Stage: Construction Documents / Permitting Stage:	0% 10%
	100%
Construction Stage: Closeout Stage:	15%
Cioseour Stage.	1570
2nd Layout Safety Crew	
Project Start Up	0%
Design Stage:	0%
Construction Documents / Permitting Stage:	0%
Construction Stage:	50%
Closeout Stage:	0%
1 Labor Crews - Perform clean up, maintain safety barriers, and other misc items that will be needed	
Project Start Up	0%
Design Stage:	0%
Construction Documents / Permitting Stage:	0%
Construction Stage:	100%
Closeout Stage:	25%
2nd Labor Crews - Perform clean up, maintain safety parriers, and other misc items that will be needed	
Project Start Up	0%
Design Stage:	0%
Construction Documents / Permitting Stage:	0%
Construction Stage:	50%
Closeout Stage:	0%

Page 25 of 27 Exhibit 1 CAM 15-1467 Page 155 of 158

### 3.10.3.1 Basis for the GMP, Drawing List

This GMP is based on the drawings and renderings listed below, dated April 13, 2012 and attached hereto:

- 1. Basis of GMP Cover Sheet
- 2. Basis of GMP First Floor Plan
- 3. Basis of GMP Second Floor Plan
- 4. Basis of GMP Parking Level 3 Plan
- 5. Basis of GMP Third Floor Plan
- 6. Basis of GMP Fourth Floor Plan
- 7. Basis of GMP Northwest Rendering
- 8. Basis of GMP Southeast Rendering
- 9. Basis of GMP Southwest Rendering
- 10. Basis of GMP East Rendering
- 11. Basis of GMP Northeast Rendering
- 12. Basis of GMP North Rendering
- 13. Basis of GMP Renovated 50m Pool Rendering
- 14. Basis of GMP West Rendering
- 15. Basis of GMP Private Development Parcel Rendering
- 16. Basis of GMP DC Alexander Parcel Rendering 1
- 17. Basis of GMP DC Alexander Parcel Rendering 2



.

• • •

•

Exhibit 1 CAM 15-1467 Page 158 of 158